Factors influencing Trust, Reliance, Performance and Cognitive Workload in Human-Agent Collaboration



Sylvain Daronnat Computer and Information Sciences University of Strathclyde

A thesis submitted for the degree of Doctor of Philosophy

Glasgow 2021

Declaration of Authenticity and Author's Rights

This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of a degree.

The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

Signed: **Datom** Date: 22/09/2022

Acknowledgements

I would like to thank my supervisors Leif Azzopardi and Martin Halvey for supporting me during this PhD.

I would also like to thank Mateusz, Olivia, Zuzana, Vasilis and Amine for their continuous support during this journey.

And of course, thanks to my parents and family for having supported me in my choices, no matter how far away I would go.

Pour Theo.

Contents

Ι	Int	troduction, Background and Methodology	1
1	Intr	roduction	2
	1.1	Motivation	2
	1.2	Context	3
	1.3	Research Questions	3
	1.4	Contribution	3
	1.5	Thesis Summary	4
	1.6	Publications	5
2	Bac	ckground	7
		2.0.1 Human-Computer Interaction	7
		2.0.2 Human-Agent Collaboration	8
		2.0.3 Challenges of Human-Agent Collaboration	9
	2.1	Trust	10
		2.1.1 Interpersonal Trust	11
		2.1.2 Trust in Automation	12
	2.2	Reliance	14
	2.3	Task Performance and Agent Reliability	17
	2.4	Uncertainty	19
		2.4.1 Uncertainty in Agent Reasoning	20
		2.4.2 Uncertainty in the Environment	21
	2.5	Measuring behavioural information	22
		2.5.1 Task Performance	22
		2.5.2 Reliance	23
	2.6	Measurement of reported information	23
		2.6.1 Trust	24
		2.6.1.1 Multi-item Instruments	24
		2.6.1.2 Single-item and Analogue Instruments	24
		2.6.2 Cognitive Workload	24
		2.6.3 Situational Awareness	25

	2.7	Resear	rch Goals	25
3	Met	thodol	ogy	27
	3.1	Motiva	ation	27
	3.2	Missil	e-Command Framework	29
		3.2.1	Godot Engine	31
		3.2.2	Interaction & game controls	32
		3.2.3	Interface	33
		3.2.4	Task Difficulty	34
	3.3	Collab	porative Agents	35
		3.3.1	Purpose & Interaction	35
		3.3.2	Aiming & accuracy	36
		3.3.3	Target Awareness & Prioritisation	37
	3.4	Evalua	ation	39
		3.4.1	Experimental Procedure	39
		3.4.2	Survey-based assessment	41
			3.4.2.1 Trust	41
			3.4.2.2 Cognitive Workload	41
			3.4.2.3 Situational Awareness	42
		3.4.3	Behavioural metrics	43
			3.4.3.1 Performance	43
			3.4.3.2 Reliance	44
		3.4.4	Data Visualisation and Statistical Testing	44
		3.4.5	Coding	45
	3.5	Study	Modalities	46
		3.5.1	Lab-based	46
		3.5.2	Online	46
Π	Ir	ivesti	gating Collaborative Human-Agent relationships	48
4	Age	ent Rel	liability and Predictability	49
	4.1	Motiv	ation	49
	4.2	Relate	ed Work	50
	4.3	Metho	od	51
		4.3.1	Agent Reliability	52
		4.3.2	Agent Predictability	52
		4.3.3	Agent configurations	53
		4.3.4	Task Difficulty	54

	4.3.5	Piloting
	4.3.6	Independent and dependent variables
	4.3.7	Experimental Procedure
	4.3.8	Demographics
4.4	Result	s
	4.4.1	Performance
	4.4.2	Reliance
	4.4.3	Trust
	4.4.4	Cognitive Load
	4.4.5	Predicting Trust
4.5	Discus	sion
	4.5.1	Predictability in the agent's actions
	4.5.2	Factors influencing trust
4.6	Conclu	1sion
Age	nt Err	ors and Behaviours 68
5.1	Motiva	ation
5.2	Relate	d Work
	5.2.1	Agent Reliability and Errors in HAI studies
	5.2.2	Defining errors
5.3	Metho	d 71
	5.3.1	Agent Errors 71
		5.3.1.1 Mistakes - Agent Delta
		5.3.1.2 Lapses - Agent Epsilon
		5.3.1.3 Slips - Agent Zeta
	5.3.2	Task Difficulty 72
	5.3.3	Independent and dependent variables
	5.3.4	Experimental Procedure
	5.3.5	Demographics
5.4	Result	s
	5.4.1	Performance
	5.4.2	Reliance
	5.4.3	Trust
	5.4.4	Cognitive Load
	5.4.5	Participants' Feedback
5.5	Discus	sion
	5.5.1	Lapses, Errors of omission
	5.5.2	Slips and Mistakes, Error of commission and intention
	 4.4 4.5 4.6 Age 5.1 5.2 5.3 5.4 	$\begin{array}{c} 4.3.5 \\ 4.3.6 \\ 4.3.6 \\ 4.3.6 \\ 4.3.7 \\ 4.3.8 \\ 4.4 \\ Result \\ 4.4.1 \\ 4.4.2 \\ 4.4.3 \\ 4.4.4 \\ 4.4.5 \\ 4.5 \\ 1 \\ 4.5.2 \\ 4.6 \\ 1 \\ 4.5.2 \\ 4.6 \\ 1 \\ 5.2 \\ 4.6 \\ 5.2.1 \\ 5.2.1 \\ 5.2.1 \\ 5.2.2 \\ 5.3 \\ 1 \\ 5.2.2 \\ 5.3 \\ 1 \\ 5.2.1 \\ 5.2.1 \\ 5.2.1 \\ 5.2.1 \\ 5.2.2 \\ 5.3 \\ 1 \\ 5.2.1 \\ 5.2.1 \\ 5.2.1 \\ 5.3.1 \\ 1 \\ 5.4.1 \\ 5.3.5 \\ 5.4 \\ 1 \\ 5.4.2 \\ 5.4.3 \\ 5.4.3 \\ 5.4.4 \\ 5.4.5 \\ 5.5 \\ 1 \\ 5.5.1 \\ 5.5.1 \\ 5.5.1 \\ 5.5.1 \\ 5.5.2 \\ 1 \\ 5.5.1 \\ 5.5.1 \\ 5.5.2 \\ 1 \\ 5.5.1 \\ 5.5.2 \\ 1 \\ 5.5.1 \\ 5.5.2 \\ 1 \\ 5.5.2 \\ 1 \\ 5.5.2 \\ 1 \\ 5.5.1 \\ 1 \\ 5.5.2 \\ 1 \\ 5.5.2 \\ 1 \\ 5.5.1 \\ 1 \\ 5.5.2 \\ 1 \\ 5.5.1 \\ 1 \\ 5.5.2 \\ 1 \\ 5.5.1 \\ 1 \\ 5.5.2 \\ 1 \\ 5.5.1 \\ 1 \\ 5.5.2 \\ 1 \\ 1 \\ 5.5.2 \\ 1 \\ 1 \\ 5.5.2 \\ 1 \\ 1 \\ 5.5.2 \\ 1 \\ 1 \\ 5.5.2 \\ 1 \\ 1 \\ 5.5.2 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ $

		5.5.3 Perception of the agents	•••	 83
	5.6	Conclusion	•••	 83
6	Vis	ual Uncertainty		85
	6.1	Motivation		 85
	6.2	Related Work		 86
	6.3	Method		 87
		6.3.1 Visual Uncertainty		 88
		6.3.2 Task Difficulty and Complexity		 90
		6.3.3 Agent Performance		 91
		6.3.4 Independent and dependent variables		 91
		6.3.5 Experimental Procedure		 91
		6.3.6 Demographics		 92
	6.4	Results		 93
		6.4.1 Performance		 93
		6.4.2 Reliance		 95
		6.4.3 Trust		 96
		6.4.4 Cognitive Workload		 97
		6.4.5 Situational Awareness		 98
		6.4.6 Participants' Perceptions		 99
	6.5	Discussion		 100
		6.5.1 Trust under Visual Uncertainty	• • •	 101
		6.5.2 Situational Awareness under uncertainty	• • •	 101
		6.5.3 Adapting under uncertainty	• • •	 102
	6.6	Conclusion		 102
7	Vis	ual Explanation and Agent Transparency		104
•	71	Motivation		104
	7.2	Related Work		 105
		7.2.1 Visualisation modalities		 105
		7.2.2 Visualisation & Situational Awareness		 107
	7.3	Method		 108
		7.3.1 Visual Agents		 108
		7.3.1.1 SA Level 1		 108
		7.3.1.2 SA Level 2		 109
		7.3.1.3 SA level 3		 109
		7.3.2 Task Difficulty		 111
		7.3.3 Agent Reliability		 113

	7.3.4	Independent and dependent variables 113
	7.3.5	Experimental Procedure
	7.3.6	Demographics
7.4	Result	s
	7.4.1	Relative Performance
		7.4.1.1 Visual Agent without Aiming Agent
		7.4.1.2 Visual Agent and Aiming Agent
	7.4.2	Reliance
		7.4.2.1 Visual Agent and Aiming Agent
	7.4.3	Relative Reported Trust
		7.4.3.1 Visual Agent and Aiming Agent
	7.4.4	Relative Cognitive Workload 121
		7.4.4.1 Visual Agent without Aiming Agent
		7.4.4.2 Visual Agent and Aiming Agent
	7.4.5	Relative Situational Awareness
		7.4.5.1 Visual Agent without Aiming Agent
		7.4.5.2 Visual Agent and Aiming Agent
	7.4.6	Participants' Feedback
7.5	Discus	ssion
	7.5.1	Task Performance and Visual Agent
	7.5.2	Reliance, Trust and Visual Agent
	7.5.3	Overloading Users
7.6	Conclu	usion

III Final Discussion and Conclusion

8	Disc	Discussion 1		
	8.1 Trust in agents		134	
		8.1.1 With regards to Agent Reliability and Error Types	135	
		8.1.2 With regards to visual uncertainty and agent transparency	135	
	8.2	Reliance on agents	137	
	8.3	Task Performance	138	
	8.4	Cognitive Workload $\ldots \ldots \ldots$	140	
	8.5	Situational Awareness	141	
	8.6	Lab-based and Asynchronous Online Studies	142	
	8.7	Limitations	143	
		8.7.1 Experimental Constraints	143	
		8.7.2 Domain-specific Constraints	144	

	8.8	Recon	nmendations	. 146
		8.8.1	Agent transparency and consistency can be prioritised over reliability $% \mathcal{A}_{\mathrm{eq}}$.	. 146
		8.8.2	Agent behaviours should be clearly categorised and defined $\ldots \ldots \ldots$. 147
		8.8.3	Taking into account the environment of interaction to better anticipate	,
			changes in human-agent collaboration	. 148
		8.8.4	Creating a framework to categorise transparent interface elements and	l
			their impact on situational awareness $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$. 149
9	Con	clusio	n	151
		-		
Bi	bliog	graphy		152
тт	T 4			
11	/	Apper	laices	171
\mathbf{A}	Age	ent Rel	liability and Predictability Study	172
	A.A	Pre-H	oc Survey	. 173
	A.B	Post-H	loc Survey	. 181
в	Age	ent Err	cors and Behaviours Study	184
	B.A	Pre-H	oc Survey	. 185
	B.B	Post-H	loc Survey	. 190
С	Visı	ual Un	certainty Study	193
	C.A	Pre-H	oc Survey	. 194
	C.B	Post-H	Hoc Survey	. 198
D	Visu	ual Ex	planation and Agent Transparency Study	201
	D.A	Pre-H	oc Survey	. 202
	D.B	Post-I	foc Survey	. 208
\mathbf{E}	Sur	vey In	struments	210
	E.A	Comp	lacency Potential Questionnaire	. 211
	E.B	Check	list for Trust between People and Automation	. 212
	E.C	NASA	Task Load Index (TLX) \ldots	. 213
	E.D	Situat	ional Awareness Rating Technique (SART)	. 214

List of Figures

3.1	Screen-capture of the latest version of the missile command game framework	
	(as used in Chapter 7), where elements of interest are annotated and described	
	below, in Section 3.2.	30
3.2	Controls used in our studies. For lab-based experiments, a controller was used,	
	whereas online studies relied on keyboard inputs	32
3.3	Experimental setup used in our lab-based studies presented in Chapters 4 and 5 $$	33
3.4	Early interface located at the bottom of the screen and used for our first study	
	focusing on agent reliability and predictability	34
3.5	Most up-to-date interface located at the bottom of the screen and used in studies	
	2, 3 and 4	34
3.6	Main variables controlling the task difficulty. Each variable can be modified to	
	induce different levels of task complexity	35
3.7	Simplified representation of how different sigma values (noted $\pmb{\sigma})$ skew the agent's	
	error margin, resulting in different levels of accuracy. The greater the value, the	
	lower the agent's performance	37
3.8	Distribution of agent's errors per missile when manually inputting a target num-	
	ber of False Positive and False Negative errors. Here, the examples shows a	
	balanced distribution of 50% False Negative and 50% False Positive errors	38
3.9	Experimental procedure using this framework. Both survey instruments (in blue)	
	and the logging of behavioural metrics (orange) were used to assess the human-	
	agent relationship at different points in the study	40
3.10	Example of the in-game prompt used for self-reported measures (in the example:	
	trust). When prompted, questions are displayed (one at a time) and participants	
	have to rate statements according to verbal anchors. Numbers and extra labels	
	can be added when needed. The pointer has to have been clicked at least once	
	before the user can click on Continue	41
3.11	$\label{eq:implementation} Implementation of the SAGAT survey instruments to evaluate Situational Aware-$	
	ness at the first level (perception of data). When prompted, the task is immedi-	
	ately frozen and hidden in the background	43
3.12	Example of a boxplot with a description of each of its components. \ldots .	44

4.1	Visualisation of the different biases applied to the agents in the study (not to	
	scale). The greater the bias, the lower the accuracy of the agent. For the system-	
	atic bias, a quadrant is randomly chosen for each participant at the beginning of	
	a session. Low systematic bias and low random variance or high systematic bias	
	and high random variance result in the same performance output	53
4.2	Simplified representation of all agents used in this study and their respective	
	aiming patterns. Both agents displaying either HIGH or LOW performance are	
	programmed with the same accuracy, regardless of their aiming patterns. $\ \ldots$.	54
4.3	Recall scores for each session. A higher score indicates a better performance.	
	Recall scores give a measure of how many missiles participants and agents hit.	
	We can see that agent Alpha (high predictability and reliability) led participants	
	to achieve higher median Recall scores than agent Gamma (low predictability	
	and high reliability)	58
4.4	Precision scores for each session. A higher score indicates a better performance.	
	Precision scores give a measure of how effective, in terms of shots fired per missile	
	hit, participants were at the task. Overall, precision scores are higher for more	
	predictable agents.	59
4.5	F1 scores for each session. A higher score indicates a better performance. F1 is	
	a harmonised mean of Recall and Precision scores. Overall, F1 scores are higher	
	for high predictability agents.	59
4.6	Amount of time each participant spent correcting the agents. A longer duration	
	indicates less reliance on the agents. At at high level of agent reliability, partici-	
	pants corrected the more predictable agent (Alpha) less than the less predictable	
	agent (Gamma).	60
4.7	Average reported trust in the agents. Higher scores indicate greater trust in the	
	agents. At a high level of agent reliability, participants reported a higher trust	
	in the more predictable agent (Alpha) than the less predictable one (Gamma).	61
4.8	Raw NASA TLX ratings for each session with agents. Higher scores indicate	
	greater cognitive loads. Raw TLX scores are lower for participants that interacted	
	with the high reliability and high predictability agent (Alpha) compared to the	
	high reliability and low predictability agent (Gamma)	62
5.1	Abstract representation of the different errors each agent made. Mistakes are	
-	errors of priority. Lapses result in no response from the agent and Slips are	
	experienced as an extreme loss of accuracy.	71
5.2	Recall scores for each session with or without agents. A higher score indicates	
-	better performance. Overall, participants benefited from the addition of an agent.	
	regardless of its error type.	76
	references of the effort of berner and a second s	10

Precision scores for each session with or without agents. A higher score indicates 5.3better performance. Overall, the presence or absence of an agent, regardless of 76its error type, did not affect participants' Precision scores in any important way. 5.4F1 scores for each session with or without agents. A higher score indicates a better performance. Participants interacting with agent Epsilon (lapses) managed to achieve higher performance than in any other condition. 77User Control Time for each session with or without agents. A higher score 5.5indicates lower reliance on the agent. As expected, the agent without any error type (Gamma) induced more reliance on its inputs than any other agent. . . . 785.6Participant ratings for the statement "I can trust the agent" for each session with agents. A higher score indicates greater trust in the agent. Participants trusted the agent without any error type (Gamma) the most, followed by the agent with the "slips" error type (Zeta)..... 79Raw TLX score for each session. A higher score indicates a greater reported 5.7cognitive load. Among the error-prone agents, the agent with the "mistakes" 80 Abstract representation of the different types of visual obstruction used in this 6.1 study. Each obstruction results in occlusions that are either near-complete or partial and static or fixed. 89 All of the different types of visual uncertainties used in this study. (a) "Top 6.2Disruption" and (b) "Bottom Disruption" occlude, respectively, the upper and bottom-most parts of the screen while (c) "Clouds" occludes the top and bottom alternatively and (d) "Darkness" obscures the totality of the screen except for the immediate proximity of the cross-hair, producing tunnel-vision. 89 6.3Recall scores for each session with or without agents. A higher score indicates better performance and a higher number of missiles hit. While a higher complexity level led to worst overall performance, participants in the "Top Disruption" condition performed better than in any other condition with visual uncertainty. 946.4Precision scores for each session with or without agents. A higher score indicates a better performance and that fewer attempts were required to hit missiles. Similarly to Recall scores, participants in the "Top Disruption" condition performed better than in any other condition with visual uncertainty. 94 6.5F1 scores for each session with or without agents. A higher score indicates a better performance. Overall, the presence of an agent actually led to overall worse performance in terms of F1 scores compared to sessions without any agent. 95

6.6	User Control Time for each session. A higher score indicates lower reliance on	
	the agent. Among all of the sessions with an agent, participants relied on the	
	agent the least in the "Top Disruption" one	96
6.7	Reported Trust scores per session, where a higher score indicates greater reported	
	trust in the agent. Participants reported higher trust levels in the "Clouds"	
	condition compared to any other type of visual uncertainty	97
6.8	Reported Cognitive Load per session. A greater Raw TLX score indicates a	
	more cognitively taxing experience. Overall, the level of task complexity affected	
	reported Raw TLX scores the most	98
6.9	Situational Awareness scores. The closer a score is to 0, the better the partic-	
	ipants' situational awareness is. The top-most plot presents overall situational	
	awareness scores while the bottom-most plot presents SA scores related to ele-	
	ments located in the bottom half of the screen. The "Top Disruption" led to more	
	under-estimations while the "Bottom Disruption" led to more over-estimations	99
7.1	Example of an Head-Up Display (HUD) interface from a study by Charissis et	
	al. [23]. A good HUD interface must manage to present task-specific information	
	without obstructing the user's field of view	106
7.2	Abstract representation of the visual agents developed for this study. Each vi-	
	sualisation supports a different level of Situational Awareness.	110
7.3	All visual agents intended to support different SA levels: (a) represents SA1	
	Threat Shape, (b) Priority Number, (c) Threat Prioritisation, (d) Agent Plan	
	display, (e) Missiles' paths and (f) Performance Graph.	111
7.4	Relative Threat Recall scores for each visualisation, organised by Situational	
	Awareness (SA) levels. A positive Relative Threat Recall score indicate a better	
	performance in terms of Threat missiles hit (True Positive) compared to the	
	baseline session without the help of any agent or visualisations. Overall, the	
	addition of an aiming agent was the main reason for increases in Relative Threat	
	Recall scores.	117
7.5	Relative Threat Precision scores for each visualisation, organised by Situational	
	Awareness (SA) levels. A positive Relative Threat Precision score indicates a	
	more efficient ratio of missiles hit (True Positive) to shots fired compared to the	
	baseline session without the help of any agent or visualisations. As opposed to	
	Relative Threat Recall scores, the addition of an aiming agent led to compara-	
	tively poorer Relative Threat Precision scores	118

7.6	Amount of time spent controlling the crosshair during each session, organised by	
	Situational Awareness (SA) levels. Positive scores indicate more manual control	
	and lower reliance on the agent. Unsurprisingly, the addition of an aiming agent	
	led to overall lower user control times	119
7.7	Relative ratings reported by participants based on the following statement: "I can	
	trust the agent". Scores are organised by Situational Awareness levels. Higher	
	scores indicate a higher reported trust in the agent. Overall, reported trust levels	
	were the most inconsistent for the "Threat Shape" visual agent	121
7.8	Relative scores calculated from the participants' ratings of the the NASA TLX	
	survey to evaluate cognitive load. Relative overall Raw TLX scores are reported	
	here, organised by Situational Awareness (SA) levels. Higher scores indicate a	
	more cognitively taxing experience. Overall, reported Raw TLX scores were lower	
	when an aiming agent was present. The "Threat Shape" (SA1) and "Performance	
	Graph" (SA3) conditions saw the most variance in reported cognitive workload	
	scores	123
7.9	Relative ratings collected using the 3D SART survey to evaluate situational	
	awareness. Relative ratings are organised by visualisations intended to support	
	specific Situational Awareness (SA) levels. Higher scores indicate a better sense	
	of situational awareness, as reported by participants. Overall, the addition of an	
	aiming agent and/or a visual agent did not change reported SA scores in any	
	significant way.	124
0.1		
8.1	venn diagram representing interactions between a user, agent, environment of	
	interaction and type of task. Blue areas indicate which combination this work	140
	studied while orange areas are avenues left for future work	140
E.1	Complacency Potential Questionnaire by Singh et al. [167]	211
E.2	Checklist for Trust Between People and Automation by Jian et al. [90]	212
E.3	NASA TLX Cognitive Workload survey designed by Hart et al. [76]	213
E.4	Situational Awareness Rating Technique designed by Taylor et al. [173]	214

List of Tables

2.1	Different types of errors presented in the work of Marinaccio et al. [118] and	
	inspired by the work of Reasons [145]	19
3.1	Summary of all agents used in our different studies with their associated levels	
	of reliability and differences in the way they behaved and made errors	36
3.2	Comparison of the lab-based and online modalities of the framework	47
4.1	Metrics related to performance (Recall, Precision and F1, higher scores = better	
	performance) and reliance (User control time (in seconds) higher = less reliance	
	on the agent). Superscript letters next to the results indicate which agents yielded $% \left(\frac{1}{2} \right) = 0$	
	significantly worse scores $(p < 0.05).$ Highest values are highlighted in bold	57
4.2	Metrics related to reported trust and cognitive load. Higher scores indicate	
	greater agreement with the statements. Highest values are highlighted in bold	61
4.3	Linear regression results when predicting participants' trust ratings by using con-	
	textual (difficulty) and behavioural measures (performance and reliance). Only	
	the most important results are presented. A higher R^2 value indicates more	
	accurate predictions	63
4.4	Spearman's correlation tests between behavioural or reported metrics and trust	
	ratings. A higher ρ scores indicates greater correlation	64
5.1	Metrics related to task performance and reliance on the agents. Higher Recall,	
	Precision and F1 scores indicate a better performance while higher User Control	
	Time indicate less reliance on the agent. Highest values are highlighted in bold	
	for each element	75
5.2	Ratings given to statement related to cognitive load and reported trust in agents.	
	Higher values indicate greater agreement with the statement or higher reported	
	rating for questions. Highest values are highlighted in bold	78
5.3	Spearman's correlation tests between behavioural or reported metrics and trust	
	ratings. A higher ρ scores indicates a greater correlation	80

6.1	Metrics related to performance and reliance recorded during the task. Higher recall, precision and F1 scores indicate a better performance while a higher user control time indicates lower reliance on the agent. Highest values are highlighted	
6.2	in bold	93
	are highlighted in bold	96
6.3	Spearman's correlation tests between behavioural or reported metrics and trust	
	ratings. A higher ρ scores indicates greater correlation	99
6.4	Results from qualitative coding analysis on post-hoc surveys data asking par-	
	ticipants to report how each condition changed their reliance on the agent and ability to play. The highest Kappa scores for each visual uncertainty type are in	
	bold	100
7.1	Average scores for Performance and Reliance metrics yielded by participants in each group related to Situational Awareness level 1. Scores are presented for	
	each session with or without an aiming agent and/or with or without a visual	
	or lower reliance on the agent	110
72	Average scores for Performance and Beliance metrics yielded by participants in	115
	each group related to Situational Awareness level 2. Scores are presented for	
	each session with or without an aiming agent and/or with or without a visual	
	agent (noted as "VA" in the table). Higher scores indicate a better performance	
	or lower reliance on the agent	120
7.3	Average Scores for Performance and Reliance metrics yielded by participants in	
	each group related to Situational Awareness level 3. Scores are presented for	
	each session with or without an aiming agent and/or with or without a visual	
	agent (noted as "VA" in the table). Higher scores indicate a better performance	
	or lower reliance on the agent	120
7.4	Average ratings or scores reported by participants for every session in the Situ-	
	ational Awareness level 1 groups. Scores are presented for each session with or	
	without an aiming agent and/or with or without a visual agent (noted as "VA"	
	in the table). Depending on the dimensions, a lower or higher score indicates a	
	better or worse experience	125
7.5	Average ratings or scores reported by participants for every session in the Situ-	
	ational Awareness level 2 groups. Scores are presented for each session with or	
	without an aiming agent and/or with or without a visual agent (noted as "VA"	
	in the table). Depending on the dimensions, a lower or higher score indicates a	1.05
	better or worse experience	125

7.6	Average ratings reported by participants on questions related to Trust, Cognitive	
	Load (NASA TLX) and Situational Awareness (SART) in Situational Awareness	
	level 3 groups. Higher trust ratings indicate greater trust in the agent, higher	
	Raw TLX scores a more cognitively taxing experience and higher SART scores	
	a better overall situational awareness. \ldots	126
7.7	Spearman's correlation tests between behavioural or reported metrics and trust	
	ratings. A higher ρ scores indicates a greater correlation	126
7.8	Results from qualitative coding analysis on post-hoc survey data asking partici-	
	pants to report one positive and one negative thing about each visual agent they	
	encountered. 3 independent researchers then assigned relevant comments to indi-	
	vidual SA levels (codes) as described in the framework by Chen et al. ("What",	
	"Why", and "Projection")	127

Abbreviations

- ${\bf FPS}\,$ Frame Per Second
- ${\bf GUI}$ Graphical User Interface
- ${\bf HAC}\,$ Human-Agent Collaboration
- ${\bf HAI}$ Human-Agent Interaction
- HAT Human-Agent Teaming
- ${\bf HMI}$ Human-Machine Interface
- ${\bf NASA\text{-}TLX}\,$ NASA Task Load Index
- ${\bf SA}\,$ Situational Awareness
- ${\bf SAGAT}$ Situation Global Assessment Technique
- ${\bf SART}$ Situation Rating Technique
- **SPAM** Situation Present Assessment Method
- ${\bf UAV}$ Unmanned Aerial Vehicle

Glossary

Agent Automated system capable of interacting with a user.

- Aiming Agent Automated agent whose sole purpose is to provide assistance with aiming.
- Cognitive Workload The reported mental effort associated with accomplishing a task.
- Collaborative Agent Automated agent designed to assist user(s) in conducting a task.
- **Game Theory** A branch of mathematics and statistics dealing with the maximisation of gains and minimisation of losses in competitive or collaborative situations involving constraints and uncertainty.
- Goal Oriented Task Tasks in which performance is measured according to pre-defined goals.
- **On-boarding** The process of introducing a system, its functionalities and modalities of interaction to users..
- Robot Physical entity capable of interacting with other human or systems.
- **Situational Awareness** Users' perception of their environment of interaction with respect to time, space, understanding of the situation, and ability to predict future outcomes.
- **Spawning** In this thesis, "spawning" describes the apparition of new missiles from the top of the screen.
- User Human operators interacting with a system.
- Viewport The area rendered by the computer that can be visualised by the user..
- Virtual Agent Synthetic non-physical entity processing information and capable of interacting with other human or artificial systems.
- **Visual Agent** Automated agent whose sole purpose is to provide assistance with the display of visualisations.

Abstract

Increasingly, automated systems are being incorporated in collaborative environments where they are used to alleviate the cognitive load of human operators while increasing task performance. Automated agents are present in a variety of domains, from safety critical environments to leisure-oriented activities, and more and more, they are being considered as a virtual teammate rather than simple decision-aid tools.

Trust is a key factor that will determine how much a human operator is willing to take into account or rely on the help provided by an automated agent. Past research on trust in automation highlights key elements that will influence its development, such as how the automated agent is perceived, how reliable the agent appears to be and how transparent its actions are. However, most related work make use of turn-based tasks where trust is measured post-hoc, which does not entirely capture the evolving aspect of trust.

This thesis presents the development and use of a real-time collaborative game where human operators can choose the extent to which they rely on the help of automated agents displaying different behaviours and various levels of performance. We used different levels of task difficulty as well as survey instruments and the logging of task-specific behavioural information to elicit and measure variables that are important to understand the human-agent relationship such as trust, reliance, task performance, cognitive load or situational awareness.

We ran four user-studies using this apparatus. The first study tested the effects of different levels of agent reliability and predictability on the human-agent relationship while the second study experimented with different types of agent errors. The third study tested the impact of different types of environmental uncertainty on the human-agent relationship while the fourth and final study measured the benefits of different kinds of visualisation-based decision-aid systems.

Overall, this work sheds lights on under-investigated issues in Human-Agent Collaboration scenarios by providing insights on factors that are most likely to harm the human-agent relationship and underline how the behaviour of agents as well as the context of interaction can drastically alter a person's attitude toward an automated agent.

Part I

Introduction, Background and Methodology

Chapter 1

Introduction

1.1 Motivation

The idea of human operators and automated agents working together to solve problems has been theorised and studied since the very early days of computer science [111]. Thanks to technological advances in computer sciences and a better understanding of user behaviours, we are starting to shift from perceiving automation as simple decision-aid tools to virtual teammates actively collaborating with users. Human-agent collaborative scenarios now involve tasks where duties and responsibilities are shared between agents and human operators [16,97]. To fulfil the potential of human-agent interaction, not only are automated agents required to be reliable and trustworthy, human operators also have to be willing to rely on their decisions and trust the agents they are interacting with. A significant amount of work has examined how trust in agents is affected over the course of both human(s)-human(s) [197] and human(s)agent(s) [103] interactions. Less work, however, has gone into the study of trust during real-time collaboration, and how specific types of agent error and visual attributes influence human-agent teaming.

This thesis seeks to better our understanding of the properties in either (a) the agents or (b) the context of interaction that are influencing users' propensity to trust and rely on automated agents. In particular, we are focusing on how the way agents make errors and uncertainty or added information in the environment of interaction affects the human-agent relationship. To answer our research questions, we designed a collaborative aiming-game in which human operators have to cooperate with agents in tasks of various difficulty and visual uncertainty. We chose an aiming task to craft a real-time human-agent collaboration scenario, as opposed to past HAI work which mostly made use of turn-based tasks. We used both validated survey instruments and behavioural information to measure, infer and study changes in the human-agent relationship. The ecological validity and real-world relevance of this task are discussed in Section 8.7.

1.2 Context

In this thesis, we focus on human-agent interaction in a real-time goal-oriented task where users and agents have explicit goals and responsibilities. The framework we developed in our work, which is described in Chapter 3, allows for the study of the human-agent relationship via validated pre- and post-hoc survey instruments, as well as task-specific behavioural metrics logged during the study.

While the focus of our work is on trust, we also study other important variables related to the human-agent relationship such as: **reliance**, **task performance** (studied via behavioural metrics), **cognitive load** (as reported by users) and **situational awareness** (when relevant). Every study was conducted using the same framework. The first two (about agent predictability in Chapter 4 and errors in Chapter 5) took place in a controlled lab-environment while the last two (about visual uncertainty in Chapter 6 and visual aid in Chapter 7) were conducted remotely.

1.3 Research Questions

This thesis focuses on how (a) **the behaviour of automated agents** and (b) **the environment of interaction** impact the human-agent relationship in a real-time, goal-oriented scenario. More specifically, we seek to answer the following sub research questions:

- RQ1 How do changes in agent predictability (how easy it is to guess its next actions) and reliability (how good the agent is at the task) impact the human-agent relationship?
- RQ2 How do different types of agent errors defined from previous related work such as slips, mistakes and lapses affect the human-agent relationship?
- RQ3 How do different types of environmental conditions (static or moving), which impair vision and induce uncertainty, affect the human-agent relationship?
- RQ4 How do different types of visual help (designed to elicit different levels of situational awareness) influence the human-agent relationship?

1.4 Contribution

The main contributions of this thesis are based on empirical findings gathered throughout four studies using an interactive human-agent collaborative framework. We show how agents behaviours, visual changes in environment and added transparency about agents' actions impact the human-agent relationship. More specifically, this work:

- Details changes in user behaviour during a collaborative human-agent interactive task using quantitative data captured during the interaction.
- Provides insights on the different ways users perceive collaborative agents from a qualitative point of view in terms of trust in the system, cognitive load and situational awareness.
- Investigates the impact of different types of error-prone agents, as well as the implications for the design of future, more trustworthy systems (Chapters 4 and 5).
- Investigates how adverse visual conditions in the environment of interaction can impact upon the human-agent relationship (Chapter 6).
- Details the process that unfolds when decision-making in human-agent collaboration is supported through different levels of visual help (Chapter 7).
- Provides an empirically tested framework that allows for the manipulation of variables important for most HCI work (task difficulty, agent performance, behaviour and transparency) and that can be used to get insights on a wide range of issues relevant to real-time human-agent collaboration.

1.5 Thesis Summary

The work presented here is organised as follows: **Chapter 1 - Introduction**: Presents the motivation, challenges and main contributions of the thesis.

Chapter 2 - Background: This section discusses the key elements in understanding the motivation for this research, and relevant related work pertaining to the study of trust in automated systems. More precisely, this section:

- 1. Presents a brief overview of the field of HCI and the context in which collaborative systems were created as well as the challenges they face.
- 2. Discusses related work on trust in automation and how most studies measure it.
- 3. Goes into details about which elements have the most impact on the development of trust, including the systems' performance and uncertainty in the context of interaction.
- 4. Summarises the research goals of this thesis in relation to previous work and gaps in literature.

Chapter 3 - Methodology: Presents the interactive game framework used in all studies presented in this thesis. This section details the motivation for using such a framework, including related studies employing similar means. This section also presents technical details about the inner working of the game, including how agents are designed, what information

the framework records and how relevant metrics such as task performance and reliance are computed.

Chapter 4: Agent Reliability & Predictability: Presents the initial study focused on how different levels of agent predictability (how easy it is to guess the agent's next actions) and reliability (how good the agent is) affect human-agent collaboration.

Chapter 5: Agents' Errors: Presents the second study focused on the way agents make errors (according to previously defined frameworks of error types) and how agent behaviours affect users differently, at the same level of agent reliability.

Chapter 6: Visual Environmental Uncertainty: This third study investigates how different types of uncertainty that impair vision in various ways (static or moving) affect the human-user relationship, including situational awareness.

Chapter 7: Visual Help: This fourth and final study investigates how different types of visualisations of the agent's reasoning process affect the human-agent relationship.

Chapter 8 - Discussion: In this chapter, a summary of our findings is presented for each independent variable investigated in our studies. In addition, recommendations are made for the design of future interactive agents.

Chapter 9 - Conclusion: This chapter provides answers to our main research questions and conclude the thesis.

1.6 Publications

Most of the work presented in this thesis was previously published at the following peer-reviewed conferences or journals:

- 1. DARONNAT, S., AZZOPARDI, L., HALVEY, M., AND DUBIEL, M. Human-agent collaborations: trust in negotiating control. *CHI 2019* (2019) [39]
- DARONNAT, S., AZZOPARDI, L., HALVEY, M., AND DUBIEL, M. Impact of agent reliability and predictability on trust in real time human-agent collaboration. In *Proceedings* of the 8th International Conference on Human-Agent Interaction (2020), pp. 131–139 [40]
- DARONNAT, S., AZZOPARDI, L., AND HALVEY, M. Impact of agents' errors on performance, reliance and trust in human-agent collaboration. In *Human Factors and Ergonomics Society Annual Meeting* (2020), pp. 1–5 [37] (Recipient of the 2020 CSTG Mark Resnick Best Paper Award)
- DARONNAT, S. Human-agent trust relationships in a real-time collaborative game. In Extended Abstracts of the 2020 Annual Symposium on Computer-Human Interaction in Play (2020), pp. 18–20 [36]

- DARONNAT, S., AZZOPARDI, L., HALVEY, M., AND DUBIEL, M. Inferring trust from users behaviours; agents' predictability positively affects trust, task performance and cognitive load in human-agent real-time collaboration. *Frontiers in Robotics and AI 8* (2021), 194 [41]
- 6. DARONNAT, S., AZZOPARDI, L., AND HALVEY, M. Investigating the impact of visual environmental uncertainty on human-agent teaming. In *Proceedings of the Human Factors* and Ergonomics Society Annual Meeting (2021), vol. 65, SAGE Publications Sage CA: Los Angeles, CA, pp. 1185–1189 [38]

Chapter 2 Background

Computer systems are implemented to make work easier for users. Automated systems are implemented in various domains and environments, from optimising the management of a nuclear power-plant with multi-agent systems [80] to booking holiday trips using a virtual assistant [176]. Studies focusing on the interactions between users and computer systems feature different implementations of computerised systems, from embodied robotic systems to disembodied software-based agents that act for the benefit of users or on behalf of them. Depending on the context of interaction, users can either interact with one or multiple systems in tasks where decisions have to be made continuously, over time. In the Human-Computer Interaction literature, different terms have been employed to describe the type of automated systems users are interacting with, going from "Machine" [13], to "Computer" [177, 196] and later, "Agents" [109]. The use of these terms is always influenced by their context of interaction and reflects the evolution in the way automation is perceived by both researchers and end-users. In this Chapter, we look at different domains of interaction and paradigms in which humanagent interaction takes place. We then present relevant work that explores these concepts and highlight gaps in knowledge. At the end of this Chapter, we introduce the research goals that motivated our research questions (presented in Section 1.3) leading, subsequently, to our user studies presented in Chapters 4, 5, 6 and 7.

2.0.1 Human-Computer Interaction

Human-Computer Interaction (HCI) encompasses all scenarios where automated systems provide support to users via software (virtual) and/or hardware capabilities (robots). Human-Computer Interaction and its potential benefits for task performance and increased safety have been theorised since before the inception of personal computing, and is described in the work of Licklider [111] in 1959 as a being somewhere between a "mechanically extended man" and "Artificial Intelligence", where computer systems, "if introduced effectively [...] would improved or facilitate human thinking and problem solving in an important way" [111]. This notion of symbiotic interaction between humans and machines was further developed in the 1970s, with companies like Xerox pioneering innovations such as the "Desktop metaphor" or "Direct Manipulation Interfaces" which allowed non-expert users to interact more seamlessly with computerised systems. Extensive reviews of studies in the the fields of mobile HCI [96] or HCI game research [19] offer insights regarding the most commonly used research methods to study users' interaction with systems. In general, HCI studies focused on user-system interactions are conducted either in a "natural" (field studies) or artificial (simulation) environment where users interact with one or more systems. The purpose of these studies, as described in the work of Kjeldskov et al. [96], can be described as the following:

- understand phenomena through data analysis,
- engineer new solutions or improve existing systems,
- evaluate the benefit or impact of a theory on users,
- or **describe** desirable properties that a system should have.

These research paradigms are helpful if we are to understand the motivation of a specific type of research, and will inform our choice of experimental design. Within HCI research, automated systems can take many forms depending on the context of interaction and type of studies undertaken. There is a key distinction between studies involving physical, embodied robots (Human Robot Interaction - HRI) and virtual, disembodied virtual agents (Human Agent Interaction - HAI). A study by Kramer et al [100] compared theories and empirical results derived from interactions with robots, agents or other humans. In their work, Kramer et al. explain that while HRI and HAI studies are both concerned with understanding users' feelings, thoughts and motivations when interacting with systems, HRI studies tend to be better at creating studies focused on affect and emotions due to the wider range of anthropomorphic acts possible for a physical robot.

As this thesis does not take place in settings with physical, embodied agents, our focus is mostly on HAI research literature, while also incorporating ideas and findings emanating from broader HCI or HRI studies. Furthermore, this thesis focuses on *evaluating* methods to study human-agent collaboration and *describing* negative and positive properties that facilitate interactions with collaborative systems. Human-Agent Collaboration (HAC) or Human-Agent Teaming (HAT) is a subset of HCI and HAI research that focuses on scenarios and environments where decision-making is shared between systems and users. In these scenarios, humans and systems constitute a team where parties collaborate to solve issues together.

2.0.2 Human-Agent Collaboration

The benefit of Human-Agent collaboration resides in combining the inherent strengths of both humans and automated agents while making up for their individual shortcomings. Nowadays, the availability of computer processing power and sensors often leads to situations where it is impossible for users alone to make sense of available data and complete their tasks efficiently and effectively. A solution to this information-overload problem is often found in the use of collaborative or fully-automated agents [5,35] which human operators can rely on in order to complete tasks successfully. Automated agents are often designed as a way to reduce cognitive load on human operators, help maintain a certain level of performance and allow for clearly defined roles and responsibilities between users and agents. This has led to automated agents being implemented in a number of environments, from disaster response scenarios where agents help with time-constrained decision making [143] to scenarios involving scheduling [187] and monitoring problems [142].

2.0.3 Challenges of Human-Agent Collaboration

A number of challenges have to be overcome before decision-aid systems can evolve from automated tools to intelligent collaborative agents [98]. The introduction of collaborative agents led to important changes in the way human operators complete tasks in scenarios supported by automation. For instance, early work on the impact of industrial robots on users elicited benefits in human productivity and decision-making while also outlining concerns about complacent behaviours due to the increase in general downtime [8]. Complacency, defined by Moray et al. [126] as "self-satisfaction which may result in non-vigilance based on an unjustified assumption of satisfactory system state", has become an ever-present problem in scenarios involving automated aid. These problems became apparent in safety critical environments where catastrophic complacency-induced accidents occurred, a fact which motivated numerous studies on the topic of complacency and reliability [132, 137].

In addition to complacency, the implementation of more decision-support systems made apparent issues related to disuses (under-utilisation), misuses (over-reliance) and abuses (no regard for consequences). Parasuraman's seminal work [134] detailed how these 3 types of wrongful use of automation could be mitigated via, for instance, appropriate training, but also required system-makers to focus on system transparency and clearly delimiting the responsibilities and capabilities of humans and automated systems in collaborative environments [134].

While the aforementioned issues are linked to how users perceive automated systems, other key problems have to do with the actual capabilities of systems. Users' mental models of automated systems and the way users interact with them change over time [103]. As early as the 1980s [127], research on the usage of automated systems found reported trust to be an important variable regarding how effective human-agent interaction could be. The work of Muir et al. stressed that "a decision aid (system), no matter how sophisticated or 'intelligent' it may be, may be rejected by a decision maker who does not trust it" [127]. Based on past work in HCI and particularly HAI, we elicited specific elements that are of particular importance regarding the quality of an interaction between a system and a user, namely:

- Trust in automation. The explicitly reported trust in a system by a user.
- **Reliance on the agent.** As operationalised given a task and captured via a context-specific proxy.
- **Performance and Reliability.** As assessed via explicit task goals clearly defined in the context of interaction. Performance is often used to describe users' success at a task while reliability is used to describe the system's capabilities or perception thereof by users.
- Uncertainty. This concept can be used to describe either the lack of transparency regarding a system's actions or a lack of information regarding the context of interaction.

In the next Sections, we present the construct listed above with relevant studies that assessed their importance in HCI, and particularly HAC research.

2.1 Trust

Trust is defined as an important factor for understanding *how* and *why* humans are willing to interact with other people [168] or computerised systems [85]. The development and evolution of trust is influenced by numerous factors such as experience, self-confidence and a general propensity to rely on another party [103,133,167]. While defining trust depends on the domain of interaction, application and task, it is generally accepted that trust represents a willingness to act based on the decisions, words or actions of "another" [106]. This other can either be a system, automated agent or human operator. The similarity between trust in automation and people is reflected in the work of Lee and See, and in particular their proposed definition of trust:

"the attitude that an agent will help achieve an individual's goals in a situation characterised by uncertainty and vulnerability [...] an agent can either be an automated system or another person that actively interacts with the environment on behalf of the person" Lee and See. [103, p. 2].

This above definition and the rest of the work by Lee and See are of particular interest as they highlight that trust:

- 1. Neither differs between team members nor differentiates between humans and automated systems.
- 2. Involves collaboration and cooperation between team members.
- 3. is task dependent.
- 4. Evolves over time and through interactions.

Past research has found that trust is a key component for effective human-human and human-agent collaboration in a wide range of domains and scenarios. In the following sections, we present studies about *interpersonal trust* and *trust in automation* that explored the development, evolution, loss and repair of trust in a variety of scenarios.

2.1.1 Interpersonal Trust

Most studies on interpersonal trust have focused on how trust is developed, maintained and lost between individuals or groups of individuals in the fields of Psychology and Organisational Studies. For instance, the work of Six et al. [168] examined organisations that explicitly favoured the development of interpersonal trust via internal policies compared to similar organisations that did not. To conduct their work, the authors relied on Eisenhardt's method to build theory based on case-studies [47], which involved looking at both theories and empirical findings. In their findings, they present four effective policies that help develop interpersonal trust:

- 1. Promoting a culture in which building relationships and showing care are valued.
- 2. Facilitating relational signalling (communication) between colleagues, no matter their rank in the company.
- 3. Socialising newcomers to teach them which values the company favours.
- 4. Actively managing and developing employees' skills and competencies.

From their recommendations, we can notice how important *transparency* and *communication* are in the development and calibration of trust over time, and how people have to understand their roles and responsibilities to maintain a positive trust relationship towards a company and its employees.

The development of trust in organisational studies has often been investigated alongside methods to best maintain trust, or repair trust in the likely event of a trust-damaging violation. While studies on trust repair are relatively recent in human-agent interaction, they were more thoroughly investigated in organisational studies and help us to understand what can negatively affect trust and how it can be regained. The work of Lewicki and Bunker [105], for instance, presents a series of recommendations to repair trust, consisting of

- 1. Acknowledging that a violation has occurred.
- 2. Determining the cause of the violation.
- 3. Admitting that the action was destructive.
- 4. Accepting responsibility.

Again, it is clear that transparent communication and a clear delineation of roles and responsibilities play an important part in regaining trust. In a related study, Gillespie et al. [66] investigated how trust repair is attempted after an organisational failure. They designed a framework of how internal factors (within the company) and external factors (outside of the company) influence employees' perception of the company's trustworthiness. This framework include lists of actions that can *contain the development of distrust*, such as imposing sanctions for individuals that breached trust, and other actions aim at *favouring the development of trust*, such as trust-enhancing communications of new regulations, or the diagnosis of previously found problems.

As we have seen, interpersonal trust is a complex construct that is impacted by many factors; some positive such as transparency and clear communication, and some negative such as violation acts that instil distrust and loss of confidence. Trust relationships between people or within organisations will fluctuate over time, and must therefore be maintained, repaired when needed, and explicitly valued in order to create good environments of interaction for all parties involved.

2.1.2 Trust in Automation

While interpersonal trust and trust in automation share similarities, for instance in the way transparency and clear communication positively affect their development, they also differ in key areas. To understand how interpersonal trust and trust in automation differ from one another, one can consult the work of Hoff et al. [85], in which the authors present a framework based on numerous prior studies on trust in automation. The framework describes some of the most important components that influence the evolution of trust prior to, and during interactions with automated agents. The model highlights how a system's features as well as users' cultural backgrounds can determine whether a user will trust the decisions of an automated system. In this work, trust is divided into three main categories: "dispositional trust, situational trust, and learned trust" [85]. All components of trust are subject to change, not only during the interaction but also outside of it. As a whole, the work of Hoff et al. emphasises that despite the complexity of the interplay of components influencing the evolution of trust, it is nevertheless meaningful to focus on single elements, and evaluate how they affect the trust relationship over time.

Some studies have attempted to isolate and study how specific users' attitudes affect their trust in automation. In the work of Singh et al. [167], an aviation monitoring task was undertaken by non-expert users to investigate the impact of complacent behaviours on task performance. Their results show that it is hard to predict complacency behaviours based on individual characteristics alone, and that "other individual and social factors may play a role, particularly in work environments" [167, p. 17]. Trust in automation is difficult to assess, monitor and measure. This difficulty is elicited in an article by Hoffman et al. [86] where challenges related to trust in automation in modern systems are presented [86]. The work by Hoffman et al. underlines that there is more than one way of measuring trust, and that each method has to take into account its context of interaction to allow for the meaningful capture of trust-related components. Most methods of studying trust and the evolution of its components consist in survey instruments administered before or after a task, such as the "Checklist for Trust between People and Automation" by Jian et al. [90] which was designed to assess both *trusting* and *distrusting* behaviours in order to get a single score reflecting participants' attitude toward automation. These methods of assessing trust, while non-context dependent, fail to take into account the ever-evolving nature of trust, as they require participants to divert their attention from a task and reflect on their past interactions. This modality of trust assessment can prove problematic, for instance, in fast-paced scenarios that require the constant monitoring of multiple systems, or when operators must remain alert in a safety critical environment.

Understanding, measuring and monitoring trust is, however, important to appreciate how human operators are willing to cooperate with automated agents, as it sheds light on the evolution of the human-agent relationship itself. The work of Merritt et al. [125] focused on the issue of "Trust Calibration", with a task requiring participants to look at luggage X-rays and decide whether they are suspicious and require further investigation or not. In their experiment, participants were helped by automated agents that provided recommendations regarding which decision to take, as well as information regarding the accuracy of the agent's recommendations. Task performance was evaluated as the number of correct decisions taken at each turn while trust was operationalised by looking at participants' perceptual accuracy, sensitivity and trust sensitivity. In their findings, the authors found that giving information regarding an agent's performance was highly correlated with positive reported trust in the agent and helped reduce the development of complacent or distrustful behaviours. The combined operationalisations of trust calibration, however, was found to be poorly correlated with task performance, indicating that there is more work to be done on the role of trust in human-agent relationships. In addition, while factors likely to predict trust may be inferred from individual user capabilities and informed by prior related work, they can ultimately prove to be bad predictors of actual reported trust in the system.

As we have seen in the work of Merritt et al. [125], the study of reported trust is paramount to help reduce the likelihood of *complacent* and *distrusting* user behaviours. In the context of human-agent interaction, many factors can lead to incidents with potentially extremely negative implications in safety-critical environments. For instance, the catastrophic accident related to the Boeing 737 MCAS system [161] was caused by software giving pilots wrong instructions after having suffered sensor malfunctions. In this situation, pilots followed the system's recommendations based on their past experience and training, but were given an erroneous mental model of the situation. Subsequent attempts to override the system ultimately failed, which led to the incident.

In past work, most studies looked at the effects of an agent's reliability on users. For instance, the work of Fan et al [57] investigated how different levels of agent reliability affects users' reliance and trust in an agent during a simulated battlefield command-and-control scenario. In their findings, the authors indicate that while higher agent reliability leads to higher reliance on the agent, users' individual skills heavily influence their attitude toward the system, with expert users remaining more cautious and less prone to complacent behaviours that novice users. In a related study about agent reliability, Hussein et al. [87] investigated how participants relied on agents in a dispatching/foraging scenario. Their findings highlight that while increasing agents' accuracy leads to lower completion times and better overall performance, it also leads to a significant decrease in users' ability to reject agents' decisions correctly, further emphasising the need to understand how to manage nascent complacent behaviours.

While the relationship between agent reliability, trust and reliance is not straightforwarded but dynamic and heavily context-dependent as highlighted by the work of Fan et al. [57] or Hussein et al. [87], recent work has focused on other factors influencing the human-agent relationship. A study by Jensen et al. [89], for instance, investigated how users' emotional experience relate to trust in agents, whereas the study of Correia et al. [34] focused on the impact that added transparency in the agent's actions can have on the user after experiencing automation failure.

As we have seen in this Section, reported levels of trust in an agent are never static, and evolve over time and through interactions [125]. Most past work on trust in automation has focused on turn-based tasks where the users and agent interact asynchronously and where reported trust in agents is measured with pre and/or post-hoc instruments. As trust is dynamic and context-dependent, more work is needed that investigates both its measurements (which methods to employ) and purpose (how does trust relate to other elements in human-agent interaction). This thesis seeks to further our understanding of trust in agents by studying how different kinds of agent behaviours (see Chapters 4 and 5), environments of interaction (see Chapters 6) and information regarding the agent (see Chapter 7) can positively or negatively affect users' reported trust in agents. The relevant literature motivating our research focus is presented in each chapter detailing our empirical studies (see Chapters 4, 5, 6 and 7).

2.2 Reliance

In HAI research, reliance is a purely behavioural construct defined by Ross et al. as a "tendency to employ automation to replace manual control" [148]. In Human-Agent studies, the concept of reliance usually describes the voluntary act of following the recommendations of an agent or not. In the influential work on trust in automation by Lee and See [103], the authors describe reliance as "a discrete process of engaging or disengaging [on automation]", and acknowledge that this definition is a simplification to explain certain relationships with trust more clearly. In their work, Lee and See further noted that "trust guides but does not completely determine reliance," [103]. And indeed, the mixed findings from many related studies in human-agent scenarios such as the ones presented in Section 2.1 by Jensen et al. [87] or Fan et al. [57] would seem to bear this out. For instance, while both the Fan and Jensen studies highlight that high agent reliability induces better task performance and higher reliance on the agent, Jensen's study [89] reports a greater failure to correct false positive errors as agent reliability increases. Due to the relationship between trust and reliance, most work has investigated both in parallel, seeking to study how reliance develops under various levels of system transparency or system reliability.

Reliance on automation and the overall propensity for users to rely on an agent are affected by many factors coming from either the users themselves or the environment of interaction. A study by Sanchez et al. [151] investigated elements likely to have an important impact on reliance on automation such as experience, age (users) or error type and error distribution (agents). In their studies, the authors employed a framework requiring participants to confirm or reject an agent's decision in an agricultural setting. The automated aid suffered from different types of errors and participants were grouped according to their age and experience with agricultural engines. In their findings, Sanchez et al. found that older participants took longer than younger participants to adapt and properly respond to the system when needed. Older participants, however, continued to check alarms more consistently than younger participants, and were therefore better able to respond to sudden automation failure. As we can see, individual factors pertaining to either users (age, experience) or the system (error types) will influences attitudes toward reliance. While these characteristics are important, they are not sufficient to explain how reliance evolves over time. In their conclusion, the authors noted that all participants, no matter their age or experience, tried to adjust their behaviours to collaborate more effectively with the automated aid, as they grew more used to interacting with the system. The findings of Sanchez et al. are reminiscent of the concept of "appropriate reliance", presented in the work of Lee and See [103], which describes the link between a system's capabilities and users' perception of them.

In a related study by Dzindolet et al. [46] on the role of trust in automation reliance, the authors focused on how explaining an agent's behaviour affects reliance in a human-agent setting. The authors found that giving reason as to why an agent might fail tends to lead to higher reliance on the agent, even when compared to scenarios including less reliable agents, and that the type of error experienced by participants (false negative or false positive) did not significantly change their attitudes toward the system. These findings are different from the
ones presented by Sanchez et al, and indicate that the context of interaction is of paramount importance in understanding reliance, as its measurement is purely behavioural.

As we have seen so far, reliance on automated systems can be studied in a number of ways, but no matter the modality, understanding users' roles, experience and the context of interaction is important to operationalise reliance and measure it in a meaningful way. For instance, when the agent provides help in the form of explicit feedback, reliance can be studied in terms of the likelihood of users adopting the agent's help in their decision-making process. When the user and agent have similar decision-making capabilities, reliance can be studied as the amount of corrections the user issues, with fewer corrections indicating more reliance on the agent. Most studies relied on the manipulation of automation failures (error types and distribution [46,151]) to understand how users interact and adapt to automation in various domains of interaction.

In an important work about reliance on automation, Parasuraman and Riley [134] detailed the different types of attitudes toward reliance, and categorised them as follows:

- Misuses: over-reliance on automation.
- Disuses: under-utilisation of automation.
- Abuses: "inappropriate" application of automation, from a system designer perspective.

This categorisation is useful if we are to understand the intent behind different types of inappropriate reliance on a system. Like trust, reliance is calibrated over time and through interaction, and for an appropriate reliance to develop, systems must be designed to help users while not leading to complacent attitudes.

The work of Garnick et al [65] sought to study reliance by inciting users to take into account an agent's recommendations. Their experiment took place in a controlled game-like environment where reliance was measured as a behavioural factor (how many times participants followed the agent's advice), and where agent and user performance could be compared or aggregated into one measure of team performance. In their results, the authors underlined the importance of designing frameworks and controlled environments to study a construct as context-specific as reliance. Changes in agent reliability, error type or mission goal can have a major impact on the development of reliance over time. The use of controlled game-like environments also allows us to quantify and model reliance via task-specific metrics, as demonstrated by the work of Boubin et al. [14] which uses the framework of Garnick et al. [65] to model users' interaction and understand appropriate reliance and trust calibrations.

As we have seen throughout this section, reliance is a behavioural factor that is linked to trust but does not necessarily share a straightforward, linear relationship with it. The study of reliance is context-specific, and most studies investigated relationships between reliance and other constructs by varying agents' errors [151] or the degree of information given to users about the system [46]. Interactive frameworks such as the one found in the study of Garnick et al. [65] and Boubin et al. [14] constitute a good, albeit abstract, avenue to contextualise, measure and study reliance on agents. In a similar fashion, this thesis measures reliance within a game-based environment, where corrections issued by the participants are recorded and operationalised as our main measure of reliance.

2.3 Task Performance and Agent Reliability

According to the work of Wiebe et al. [190], performance is generally understood as the measure of an outcome in cognitive tasks. Similarly to reliance, performance is a context-specific construct. In HAI studies, performance is often understood as how well the human-agent team can succeed at a task presenting clearly defined goals, given different levels of human expertise and agent reliability. In an article by Lewis [109] on "Designing for Human-Agent Interaction", the author shows that most studies vary the level of information and feedback about the system's actions and reliability to see resulting changes in users' task performance, reliance and trust.

In HAI studies, while task *performance* represents how good the users and/or agent are at meeting specific goals, agent *reliability* represents the accuracy of the help given by the agent. In general, it is assumed that agent reliability is key to understanding how human operators perform and calibrate their trust in agents. In a work by Hoc et al. [83], the authors present a framework where both agents and users are considered as separate agents pursuing their own goals in a driving task. In their framework, the impact of both the user's performance and the agent's level of reliability can be evaluated individually or collectively, as tasks require users to act on their own or rely on inputs given by the agent. For maintaining high task performance in a safety-critical scenario, the authors recommend that agents should be designed to support users, not replace them, and that the responsibilities of both users and agents should be clearly defined.

The relationship between users' performance and agents' reliability is, however, not always straightforward. In human-agent collaborative scenarios where user(s) and agent(s) form a team, performance is often evaluated as a single construct, which makes it harder to study the impact that either the user or agent has on overall team performance. The work of Fan et al. [57], outlined in Section 2.1, studies team performance in an experiment focusing on agent reliability and user expertise. The authors found that an increased knowledge of the agent's reliability level help mitigate the chances of false positive errors. Similarly, Chavaillaz et al., [24] tested the impact of different levels of agent reliability on trust, reliance and task performance in a turn-based X-ray scanning scenario. Their results showed that a decrease in agent reliability resulted in a decreases in users' reported trust in the agents. Furthermore, Chavaillaz et al. found that users' perception of the reliability of agents was more accurate when interacting with low performing agents, which led to complacent behaviours. As we can see from both Fan and Chavaillas' studies, added transparency regarding the agent's level of reliability significantly

influences the way users trust and rely on the agent's input, which, in turn, positively affect overall team performance. In addition to studies focusing on different degrees of agent reliability in assessment tasks, the work of Shirado et al. [166] explored turn-based coordination problems, in a colour-selection game. In their work, the authors found that error-prone agents (up to 30% loss in accuracy) can be beneficial to collaborative task performance as they reduce the chance of users being complacent while interacting with the agent.

Given the evidence of past research, it is clear that agent reliability is one of the major factors that will influences both team performance and a user's propensity to trust and rely on the agent. Most studies manipulated the way agents made errors by having different levels of agent reliability, but other methods have been employed to vary the way an agent performs and makes errors. While some studies (as described in Section 2.1) introduced false-alarms (Merritt et al. [125]) or systematic biases (Fan et al. [58]), other studies focused on the *way* agents make errors, rather than on testing different levels of agent reliability. For instance, some studies experimented with agents that suddenly stop working, such as presented in the work of Correia et al. [34] that involves a robot and a user playing a collaborative card game. In their study, the robot would explain (or not) its faults to the user. With their findings, Correia et al. showed that minor faulty behaviours that are harmful to trust and task performance can be mitigated by simply having the robot acknowledge its own shortcomings. More serious loss in the robot's reliability, however, still resulted in inferior team performance, no matter the justification used by the robot.

In most cases, prior work delving into agent reliability has showed that participants react differently to various types of automation failures, and report higher trust in systems that can justify their behaviours. This represents a new avenue to not only understand how team performance can be degraded, but also mended back to suitable level.

Any type of agent failure can be loosely described as an "error". "Errors", however, is a term that fails to explain the detailed nuances of how an agent can stray towards undesirable outcomes. The work of Marinaccio et al. [118] presents 4 unique types of errors: mistakes, lapses, slips and violations. All of which are derived from human-human interactions studies [145] and are presented in table 2.1. These definitions of errors all come from studies focusing on *human errors*, where they were also conceptualised in the context of human-human interaction in healthcare in a study by Kim et al. [94] that investigated these types of errors alongside different repair mechanisms on individual and groups of people. In their findings, Marinaccio et al. explain how more work should be done to analyse how the particular features of a system and the way in which it errs influence human-agent relationships.

Overall, a number of studies empirically tested the effect of trust-damaging acts in both human-human and human-agent settings by manipulating the nature of an agent's violation. The work of Baker et al [11,125] presents a comprehensive review of past research in the field of human-robot interaction with ideas for future research on maintaining and repairing trust in robotic agents. Among the 6 avenues for future research mentioned by the authors, one is of particular relevance for human-agent collaboration: "Adapt existing trust research to investigate how robot features affect trust" [11, p. 20]. As we have seen in previous work on agent reliability [118], one of the most important features of future research should be the way in which agent make errors: how to categorise them and how users perceive them.

In this section, we have looked at past work on human-agent interaction to understand how agent reliability is defined and varied to study its resulting impact on users' trust and performance. We have seen that task performance in HAI scenarios is often measured with taskdependent metrics related to the success of one particular activity, whereas agent performance (how good at the task the agent is), is controlled in terms of reliability [14, 58, 65]. However, as we have seen when looking at trust in Section 2.1, system reliability is only one factor that influences trust in agents and the overall human-agent relationship. In this thesis, in addition to reliability, we study the impact of different agents' behaviours in terms of predictability and intent, and their resulting impact on participants.

Table 2.1: Different types of errors presented in the work of Marinaccio et al. [118] and inspired by the work of Reasons [145].

Error Type (Reason, 1990)	Examples	Violation Type (Kim et al. 2013)	Effective Repair (Kim et al. 2013)
Slips - Errors of commission - when an intended action is wrongly executed	Flipping the wrong switch on an IV pump	Integrity-based	Denial
Lapses - Errors of omission - resulting in failure to carry out the action	Forgetting to administer medication	Competence-based if due to memory failure, integrity-based if attention failure	Context-dependent
Mistakes - Errors of planning or judgment	Prescribing an incorrect dosage	Competence-based	Apology
$\label{eq:Violations-Intentional commission} Violations \ - \ Intentional \ commission \ of \ an \ error$	Prescribing an inappropriate medication because of sponsor loyalty	Integrity-based	Denial

2.4 Uncertainty

Uncertainty is a subjective construct that involves risk and opportunities, as defined by Rachev et al. in their work on uncertainty in the financial domain [140]. Uncertainty is most studied in the field of economics for risk management purposes, in order to predict potential negative outcomes or lessen their impact. In HAI research, uncertainty is inherent to most environments of interaction, and it is often the role of an automated agent to lessen uncertainty by processing data and anticipating potential outcomes. Uncertainty is harmful to any collaborative settings, whether they include only people [188] or people with agents [101], the latter being the focus of a number of important human-agent teaming studies assessing uncertainty in high-risk domains such as driving [101] or aviation [167]. As uncertainty comes from aspects either within or outside of the human-agent relationship, the following subsections will present prior work that studied uncertainty related both to agents and the environment of interaction.

2.4.1 Uncertainty in Agent Reasoning

In the work of Wu et al. [198], the authors proposed frameworks to manage uncertainty between entities (humans or automated agents). The constituents of the agents' reasoning are represented using the BDI (Belief, Desire and Intention) attributes where each agent is programmed with "beliefs" (what the agent knows), "goals" (desired outcome) and "plans" (how to achieve the desired outcome, step by step). This framework allows for reducing uncertainty by making evident what the agent knows or doesn't know, and by openly sharing its intentions with other agents or users. From the work of Wu et al., we see that both assigning clear roles and enforcing transparent intent in the decision-making process of all entities are paramount to reduce risk and uncertainty. In HAI studies, the concepts of "trust calibration" or "appropriate reliance" (described respectively in Sections 2.1 and 2.2) are commonly found and employed in an effort to increase transparency and reduce uncertainty. During calibration, by increasing the transparency of the agent's actions (in terms of how its inner reasoning is presented), users can judge the capability levels of the agent in order to adapt their trust and reliance accordingly.

The work of Kunze et al [101] investigated how to best communicate uncertainty in an HAI context. Their findings show that while displaying some degree of information about the agent's reasoning helps improve performance, prolonged attention and monitoring induce a higher cognitive load, which in turns leads to users not being willing to interact with the system. As demonstrated by Kunze's study, transparency may be helpful to reduce uncertainty but it comes at a cost, especially given the primary function of agents is to reduce rather than increase cognitive load. Related work from Sacha et al. [149] thoroughly reviewed prior studies that researched how to best communicate uncertainty and improve trust with visual analytic tools. In their findings, they highlight what they call "traps" concerning uncertainty, such as being clear about exactly what is meant by agent uncertainty, and having a system that overloads users with information that is not of immediate relevance. Such an information overload could even lead users to create erroneous mental models of the agent's inner workings, which in turns provokes more uncertainty. For instance, the study by Wang et al [186] created a framework for automated agents to automatically come up with an explanation for their reasoning for particular actions. As transparency is a key element in reducing uncertainty, the authors found that explanations did indeed lead to higher trust in the agent and better team performance. A side-effect, however, was that the added confidence in the system led participants to believe that they fully understood the robot's decision-making process, whereas the explanations provided by the system were *not* sufficient to make such claims [186].

No matter the context, we have seen throughout this section that uncertainty in the agent's decision-making process can be reduced by increasing transparency. This added clarity, however, comes at a cost: users increase their attention levels, which in turn affects cognitive load. To mitigate such problems, researchers have experimented with various visualisation techniques and issued some recommendations, such as:

- Reducing the attention required or the amount of information provided by the system [101].
- Taking into account the context of interaction and being clear about the kind of uncertainty that the system should manage [149].
- Not over-simplifying the information communicated, so as to prevent users from having erroneous mental models of the system's capabilities [186].

2.4.2 Uncertainty in the Environment

A major factor in the development of uncertainty is the context of interaction itself. In this Section, we will use the terms "environment" and "context of interaction" interchangeably. In HAI settings, uncertainty can manifest itself differently depending on the task and context of interaction. In these situations, Game Theory offers relevant frameworks and techniques to study and assess uncertainty, as it often involve optimisation problems and minimising risk. For instance, in a game scenario where information is incomplete, a common strategy for a player who seeks security would be to work towards a "secure equilibrium", which is a state of action that would lessen the impact of the worst predicted outcome [179]. In an HAI task, however, this secure equilibrium can be hard to define, as dynamic changes in the environment can alter the decision-making process of both users and agents in real-time, which makes planning difficult. A number of studies have focused on the visual (un)availability of information in the environment of interaction and its impact on users' decision-making in HAI settings.

Uncertainty coming from the environment of interaction is especially an issue in situations where task-sensitive information used to inform decisions is limited or unavailable. To help understand and mitigate this type of uncertainty, a study by Sarter et al [152] focused on the impact of different display aids for pilots in uncertain and time-sensitive scenarios, especially in the aviation domain. In their findings, the authors show that the type of display and the accuracy of the information were the most important components related to task performance. In addition, environmental uncertainty has been found to be mitigated by agents capable of giving information that lessens the impact of uncertainty. Such findings are presented in the work of Kunze et al. [101] where different types of visualisations and behavioural metrics were used to adapt and present information to users while not overloading them with data. While agents can help mitigate uncertainty, paying prolonged attention to explanations, however, could lead to lowered situational awareness and even missing real-time changes in uncertainty displays [101]. Beyond the unavailability of data, uncertainty comes into play in scenarios where predictions are being made based on the assessment of existing information. In these situations, prior information and the way it is constructed, shared and understood is crucial to how future decisions are made. A study by Herdener et al. [81] demonstrates that providing information about potential outcomes, with a sense of variability, does indeed have an impact on decision-making and cognitive load.

Most work in the field of Human-Agent interactions has focused on the uncertainty inherent to the decision-making capabilities and transparency of automated agents. As we have seen in Section 2.4.2, uncertainty can also come from the environment of interaction, and there is currently a knowledge gap regarding how users' behaviour evolves during a human-agent task in an uncertain environment. As a means of responding to this gap in knowledge, the current thesis will present work related to human-agent collaboration under visual uncertainty in Chapters 6 and 7.

2.5 Measuring behavioural information

Behavioural data are information captured during human-agent interactions that act as a proxy for some of the concepts detailed in this Chapter, such as reliance (see Section 2.2) or performance (see Section 2.3). These measures are all context-specific and are often collected by systems sensors and operationalised according to the explicit goal of the task, such as *task performance* and *reliance*. Every study conducted during this thesis and presented from Chapters 4 to 7 records and analyses behavioural data to understand and model user behaviour.

2.5.1 Task Performance

In human-agent collaborative scenarios, the outcome of a task is often measured in order to assess the quality of interaction, as detailed in Section 2.3. As performance is an important variable in any kind of goal-oriented task, it is important to measure it via adequate methods that are in accordance with the goal of the task. The work of Lebas [102] in the field of organisational sciences defines performance as "contextual both in terms of users and in terms of purpose." [102, p. 2]

Some studies, focusing on the acceptance of automated aid, report performance as the number of times participants relied on the agent when the agent was providing reliable inputs. Examples of these studies include the work of Fan et al. [57] or Hussein et al [87] presented in Section 2.3. Other researchers measured team performance as a whole (human and agent combined), which is only possible during collaborative scenarios where users and agents have similar capabilities and levels of agency, such as is shown in the work of Shirado et al. [166]

One of the most common metrics used to measure performance (notably in the field of Information Retrieval and Natural Language Processing) are Recall, Precision and F1 [136], which are based on the notion of false or true positives and negatives, and can be easily adapted to most goal-oriented tasks where agents provide help that can vary in terms of reliability. For the studies presented in this thesis, we will mostly rely on Recall, Precision and F1 for the assessment of individual and team performance. These metrics are further detailed in Section 3.4.3.1. It is also worth noting that we do not measure the rapidity at which participants complete the task as every session is timed and speed is not considered as a proxy for performance in our studies.

2.5.2 Reliance

Reliance and trust are often studied together in HAI and Human Factors studies [46, 120], as they share an evident, albeit complex, relationship as described in Sections 2.1 and 2.2.

In terms of behavioural measurement, reliance is often studied in conjunction with compliance, as elicited by the work of Chancey et al: "The operator responding when a signal is issued is referred to as compliance. The operator refraining from a response when the system is silent, or indicating normal operation, is referred to as reliance" [21, p. 1]. In HAI studies where the agent not only guides the user but also interacts on its own or on the behalf of the user, reliance is often studied as the amount of corrections the user issues, with fewer corrections indicating a greater reliance on the agent. An example of such a behavioural measure can be found in the work of Hussein et al. [87] where users' acceptance of the agent's recommendations can be recorded by the number of time they followed said recommendations.

While there is no consensus on which behavioural or physiological metric best represents a robust proxy for the measure of trust in automation, a few other physiological variables have been explored, such as Heart rate [101], post-neuronal activity (EEG [3]), gaze and electrodermal activity [185]. While the usefulness of some of these variables might be highly affected by the context of interaction, they still represent potential avenues for monitoring reliance in a context-free manner.

In this thesis, reliance will be measured using a context-specific variable: "user control time", or the amount of time for which users took over control from the agent. This variable is presented in Section 3.4.

2.6 Measurement of reported information

In user studies, reported information is often recorded as ratings given to either statements or questions on likert-scale instruments administered before, during and/or after a task. Survey instruments are commonly used to assess subjective concepts such as trust, cognitive workload or situational awareness that are otherwise hard or impossible to record using behavioural measurements. Every study presented in this thesis (from Chapters 4 to 7) relies on validated survey instruments to understand how users perceived automated agents, and how their perceptions relate to their behaviours while interacting with agents. Most of the original survey instruments used in this thesis are presented in Appendix E.

2.6.1 Trust

Trust is often measured using rating scales where users have to indicate how they agree with one or more statements, eliciting either an overall attitude toward the system or evaluating sub components linked to the development of trust.

2.6.1.1 Multi-item Instruments

One of the most widely used self-reporting scales is the "Checklist for Trust between People and Automation" by Jian et al. [90] which consists of 12 statements that participants have to rate on a 7-point Likert scale (with higher scores indicating greater agreements). The purpose of each statement is to elicit different attitudes toward the system linked to trust (reliability, honesty...) or distrust (suspicion, deception...). This instrument can be used before or after an experiment in order to measure how trust and distrust evolve. This 12-item instrument has been widely used in a wide range of studies owing to its context-free nature, from Satterfield's work involving UAVs [153], to Erebak's study on robots in the field of elderly care [55].

2.6.1.2 Single-item and Analogue Instruments

Multi-item survey instruments are best used to assess sub-components of multi-dimensional concepts, such as distrust in the work of Jian et al [90]. In many scenarios, however, multi-item survey instruments can prove too cognitively taxing to use, reducing the quality of the answers provided, since many studies not only assess trust but other relevant metrics such as cognitive load [2] or situational awareness [70]. As a result, many studies have resorted to the use of short and quick-to-deploy instruments for the measurement of trust in a system. Some are made and tailored to the need of particular studies, such as the one used by Wiczorek et al. [189] consisting of an analogue trust scale with verbal anchors ranging from "my trust is very strong" to "I barely trust the system" [189, p.7], where participants' answers are then transformed using a scale from 0 to 100.

2.6.2 Cognitive Workload

While Cognitive Load theory is a concept that attempts to understand how people make decisions based on evolutionary theory [171], Cognitive Workload is a concept that has been most studied in human-machine interface research, and refers to the study of the "cost of accomplishing mission requirements for the human operator." [75] By far, the most common way of assessing Cognitive Load is via a pre- or post-hoc 6-item survey instrument named NASA Task Load Index (TLX) [76] that has been used and tested by numerous studies since its inception in 1988, including in an important meta-article by its original authors that analyses 20 years worth of studies that relied on the instrument. [75] In this thesis, cognitive load is measured in all studies using NASA TLX, and its score is usually reported using the Raw TLX technique [17].

2.6.3 Situational Awareness

As alluded to in Section 2.4.2, an important concept in understanding how users make decisions is Situational Awareness (SA). SA can be described as the capacity to locate and remember important information in order to make predictions about future outcomes [48]. There are three commonly used methods to measure situational awareness, namely "SAGAT", "SART" and "SPAM":

- **SAGAT** [49] (Situation Awareness Global Assessment Technique) consists in designing a set of SA related queries related to a specific task. During the experiment, the task is frozen, giving time for participants to answer the queries. A numerical value is then derived from each question and aggregated into an overall score evaluating participants' overall SA.
- **SPAM** [45] (Situation Present Assessment Method) is similar to SAGAT but puts an emphasis on quick questions focused on locating information in the environment. The response time is taken into account as an indication of SA.
- **SART** [173] (Situation Awareness Rating Technique) consists of a pre-determined set of general queries related to Situational Awareness.

All methods have pros and cons. While some require freezing of the current task (SAGAT), others require multi-tasking (SPAM), or the completion of a post-hoc survey (SART). A recent comparison of all these methods [52] found the SAGAT method was the least intrusive and least harmful in terms of users' task completion performance. The SART method, however, offers a post-hoc context-free survey instrument that could potentially be adapted to any scenario. In this thesis, Chapters 6 and 7 investigate the impact of different visual conditions on situational awareness and make use of the SAGAT and SART methods.

2.7 Research Goals

The work of Baker et al. [11] discusses important challenges for studies on trust in automated systems and robots, and makes recommendations on which issues are important to tackle in future work. Of the 6 recommendations elicited, 2 are of particular significance for humanagent collaboration: "Adapt existing trust research to investigate how robot features affect trust" and "Develop & validate measures of human-robot trust". Following on from these recommendations, and based on previous work in the field of human-agent interaction, this thesis seeks to contribute to our understanding of trust in collaborative automated agents during real-time scenarios, both through self-reported survey instruments and behavioural information directly captured from human-agent interactions. More specifically, this work seeks to address the following gaps in the HAI literature by:

- 1. Experimenting with new ways to elicit and record human-agent interactions.
- 2. Assessing both reported subjective metrics and recorded behavioural information related to the human-agent interaction.
- 3. Studying how trust in automated agents develops and which elements influence it the most.
- 4. Experimenting with different types of agent behaviours and environments of interaction.

While our focus is on the development of trust in automated agents, we sought to study the development of related elements based on previous relevant work. In the study we conducted, we are measuring the following dependent variables:

- 1. Trust. With self-reported metrics.
- 2. Reliance. By recording and analysing task-specific variables.
- 3. Performance. According to the tasks' goals and performance metrics.
- 4. Cognitive Load. Via validated self-reported questionnaires, such as Cognitive Load.
- 5. Situational Awareness. Via different validated methods including validated surveys and task-specific prompts.

By taking into account the gap in knowledge identified in this section, and the research goal elicited above, we present the following research questions, also presented in Section 1.3:

- RQ1 How do changes in agent predictability (how easy it is to guess its next actions) and reliability (how good the agent is at the task) impact the human-agent relationship?
- RQ2 How do different types of agent errors defined from previous related work such as slips, mistakes and lapses affect the human-agent relationship?
- RQ3 How do different types of environmental conditions (static or moving), which impair vision and induce uncertainty, affect the human-agent relationship?
- RQ4 How do different types of visual help (designed to elicit different levels of situational awareness) influence the human-agent relationship?

Chapter 3

Methodology

3.1 Motivation

Past human-agent interaction studies assessing trust, reliance and cognitive load often consist of either competitive (user VS agent(s)) [128] or collaborative (user AND agent(s)) [34, 91] scenarios where users interact with one [169] or multiple agent(s) [147]. In these studies, agent(s) either provide information aimed at helping users make a decision [27] or directly interact with the task as another autonomous entity [135]. In most HAI studies, the humanagent relationship is assessed via validated, general-purpose survey instruments which focus on different reported variables, such as trust [90], cognitive load [76], situational awareness [49] or system usability [108].

To answer our Research Questions presented in Section 1.3, we needed a human-agent collaborative framework that meets the following requirements:

- 1. Provides a collaborative scenario where human and agents have to work together in a **real-time task**.
- 2. Can record information related to the human-agent interaction task itself, such as performance or reliance on the agent.
- 3. Allows the usage of **standardised survey instruments** to measure reported measures such as trust, situational awareness or cognitive load.
- Can allow the modification of both the agent's attributes (behaviours, reliability) and the task's features (difficulty, information provided or excluded).

Finding a framework that allows for the elicitation and collection of meaningful information while keeping participants engaged is an on-going problem in many research domains. For more than two decades [20,200], HCI studies have reliably used commercially available games (such as "Minecraft" [32]) as well as purpose-made games (Atomic Orchid by Fischer et al. [61]) in order to study participants' behaviours in controlled environments. Based on our needs and after carefully reviewing relevant work, we decided to use a well-known and simple arcade game named "Missile Command" [10] and adapt its gameplay to a human-agent teaming scenario. We chose this particular game as its gameplay requires users to make quick decisions in timelimited scenarios where the difficulty of the tasks and reliability of a collaborative agent can be easily controlled. In comparison to other popular and accessible arcade games such as Space Invader or Pac-Man, Missile Command offers more flexibility regarding the addition of an agent, as our tasks strive to elicit fast decision-making, rather than focus on decision-planning. Other games were considered, such as DOTA 2 or Counter-Strike due to their anteriority and presence in related research [12, 141]. However, their inherent complexity, learning-curve and limited flexibility for custom-made agents meant that they failed to meet our study requirements.

Carter and collaborators [20] argue that games in HCI studies can be understood within 4 distinct research paradigms: "Operative", "Epistemological", "Practice" and "Ontological". In the context of our work, we are interested in the **Epistemological paradigm**, as our research seeks to use a game-like activity to generate insights on a broader topic, namely *how humans trust and collaborate with an automated system*. We also focus on the **Ontological paradigm**, as our work strives to inform the design of future interactive systems by focusing on the elements that make up the game's interface or mechanics.

In order to satisfy all our requirements, we created an interactive human-agent collaborative task in the form of a 2D collaborative game. In this game, the user and agent have to cooperate to protect assets by destroying a series of incoming missiles. The agent can aim automatically, however only the user has the ability to fire projectiles. This framework offers the following advantages:

- It offers a real-time task where important behavioural constructs can be operationalised and recorded, such as task performance (number of missiles hit or shot fired) and reliance on the agent (amount of times the user corrected the agent).
- The task can be easily divided into blocks of various difficulty levels where survey instruments can be deployed to measure pre- and post-hoc concepts such as trust, cognitive load or situational awareness.
- It can be deployed either in a lab environment (with gamepad or keyboard controls) or exported online as a web-app where participants go automatically through all stages of the study.
- Flexibility, as elements such as the agent's accuracy, the availability of information or the complexity of the task can be controlled and monitored to design immersive scenarios.
- The task is user-friendly and requires no prior knowledge. Controls are easy to learn, and the goal of the task is straightforward.

Using commercially-available games can be a quick and effective way to have participants engage in immersive tasks, especially when involving a multitude of users. For instance, the work of Schaekermann et al. [156] used the game "Destiny" to understand what motivates player and how this relates to their own personality. Results from such studies, while insightful, are heavily context-specific, and commercially available games are often not flexible enough to incorporate custom-made AI agents or the logging of specific behavioural information. In addition to commercial games, frameworks used in previous work often occurred in either domain-specific [57, 60, 61] or turn-based scenarios [157] where the human-agent relationship evolves asynchronously. Trust, as described in Section 2.1, evolves over-time and designing frameworks that allow for the study of its interactions with other elements would benefit our understanding of trust in automation. More recently, an increasing number of studies have used arcade-like games to study human-agent relationships, as noted in the work of Rapp et al. [144] who describe arcade games as being "inspiring" thanks to their "simplicity and immediateness" [144, p. 4].

3.2 Missile-Command Framework

In this section, we present the overview of our framework, mostly inspired by the 1980 "Missile Command" arcade game [10]. The goal of our real-time interactive task consists in aiming at and destroying missiles appearing from the top of the screen in order to protect 4 cities situated at the bottom of the screen. To achieve this goal, participants can move a crosshair across the screen and fire projectiles in the direction of their choosing. In our studies, agents help participants by moving the crosshair automatically. At any moment, however, participants can choose to override the agents' input and manually move the crosshair. In all scenarios, only participants can fire projectiles to destroy incoming missiles. As in related studies where reliance is defined as "the tendency to employ automation to replace manual control" [148], we decided to ensure that there was a way for participants to *correct* the agent's decision by taking over the controls and *confirm* theirs or the agent's actions by assigning firing capabilities to participants. This design decision also ensures that participants have to keep on monitoring the situation, in order to keep a maintain the necessary level of engagement with the task.



Figure 3.1: Screen-capture of the latest version of the missile command game framework (as used in Chapter 7), where elements of interest are annotated and described below, in Section 3.2.

Figure 3.1 shows our interactive game in action, where the main elements are numbered and described as follows:

- 1. **Gun-turret**: controlled by either the participant or the agent in order to aim and target incoming missiles. All projectiles are fired from the turret.
- 2. **Projectile**: fired by the participant, it travels at a fixed speed (1250 pixels/second) until it explodes in a small circular area. If a missile lies within this area, it is destroyed. The speed of the Projectile was constant and hard-coded for each study after initial informal pilots were conducted to tailor the different elements in the game.
- 3. Crosshair: provides a visual indication of where the participant or agent is aiming. The crosshair changes colour depending on who is controlling it: yellow for the participant, white for the agent, and dark-grey for neither. Crosshair colours were selected to be easily distinguishable depending on whether the user or the agent was controlling the crosshair. In addition, we ensured that each colour (white and yellow) represented a significant contrast to the dark blue background of the game for increased accessibility. The speed at which the agents or the users can move the crosshair was the same throughout each study (800 pixels / second).
- 4. **Red Indicator Area**: appears when a projectile is fired to show participants the area where the projectile will explode.
- 5. **Projectile's explosion/halo**: In order to be destroyed, missiles have to enter the radius of such an explosion.

- 6. City: Assets that the participants are tasked with protecting.
- 7. **Missile Impact**: when a missile reaches a city, it produces an orange/red explosion with smoke emanating from the city.
- 8. User and Agent panels: The participant's panel (on the bottom left of the screen) and the agent's panel (on the bottom right) light up in green when one of them is moving the crosshair.
- 9. Enemy missile: progresses at a fixed speed and angle depending on the task difficulty. At the end of a session (with or without an agent), participants are shown how many missiles they hit and/or missed. All missiles missed eventually hit a city. Depending on the study, the number of missiles appearing (spawning) and their individual speed was varied in order to test different levels of task complexity.

3.2.1 Godot Engine

After conducting a review of the different tools dedicated to the creation of interactive software, we decided to use a game engine [110] as game engine specialise in the prototyping and deployment of highly interactive software while taking into account hardware constraints and the need for speed of execution, all of which are paramount to the success of seamless, real-time and immersive human-agent interactions. Multiple engines were considered, such as Unity [74], Unreal [54] and Godot [69]. All of these engines have been used in HCI studies, for instance to simulate visual impairments via Virtual Reality [107] or to promote e-learning [31]. After carefully examining the pros and cons of each game engine, we decided to choose the newer Godot Engine to create our interactive game as it is:

- A highly mature 2D engine with a very simple Python-like programming language called Godot Script, which makes prototyping much easier that with, say, Unity which uses C++. This drastically cut-down the learning curve associated with designing prototypes.
- Much lighter than Unity or Unreal, which reduces hardware requirements for both developers and end-users.
- Completely open-source, which could facilitates the reproduction of results and dissemination of the game.
- Easily exportable to WebGL and HTML, which was important to conduct online studies running on different types of hardware via web browsers. Such exports are possible under the Unity or Unreal Engine, but have an impact on the performance of 2D games (as of 2021).

All studies presented in this thesis were designed using the aforementioned Godot Engine in its versions 3.0 (for the study presented in Chapter 4) and 3.2 (for the studies presented in Chapters 5, 6 and 7).

3.2.2 Interaction & game controls

For participants to play the game, they have to assume control of the gun tower located at the bottom of the screen to fire at incoming missiles (see element 1 in Figure 3.1). In our lab-based studies, presented in Chapters 4 and 5, participants were interacting with the game via an Xbox360 controller using the analog stick to move and the (A) button to fire projectiles. Subsequent studies presented in Chapters 6 and 7 were conducted entirely online, where participants interacted with the game using their keyboard's directional arrows to move the crosshair and space bar to fire projectiles. The controls used in all studies are displayed in Figures 3.2 and 3.3.



Figure 3.2: Controls used in our studies. For lab-based experiments, a controller was used, whereas online studies relied on keyboard inputs.

In all studies, participants alone were able to fire projectiles, which fact we used as a way of measuring the extent to which participants were explicitly confirming agents' actions. When a projectile is fired, the following events happen:

- 1. A red target/cross is placed at the location where the crosshair is located. This red target indicates where the projectile is heading and will remain there until a projectile has reached it.
- 2. A projectile is travelling from the gun-tower to the red target at a fixed speed hard-coded into the game. The speed of the projectile was calibrated after initial pilots to adjust game elements.
- 3. Upon reaching the previously spawned red target, the projectile will disappear and spawn in the same location an "explosion" in the form of a blinking circle with a fixed radius. The blinking circle will remain in place for a set duration.

During sessions with the agents, participants could decide to let the agent guide the crosshair, or take over the controls and correct the agent manually. We recorded the number of times participants took control and the duration of each takeover as a way of measuring explicit reliance on the agents.



Figure 3.3: Experimental setup used in our lab-based studies presented in Chapters 4 and 5

3.2.3 Interface

While the majority of the elements in our game remained unchanged throughout our studies, some visual changes and improvements were incorporated by making use of participants' feedback. The most noticeable changes were brought to the game interface (see Figure 3.4 used in the first study, and 3.5 used in all subsequent studies) where we decided to simplify three items:

- The "lines" indicating the agent or the user was controlling the crosshair. We decided to make them bigger and use colours (yellow for users and white for agents) depending on which one was in charge of aiming.
- 2. Names and textbox messages. In the first study (see Figure 3.4), we used simple messages when certain actions were conducted (for instance "get ready..." when the task was about to start, "fire now!" when the agent was in position). We decided to remove them for sub-sequent studies as the message could have distracted participants and anthropomorphised agents in ways that would be hard to assess and control.

3. Labels AIMING HELP (agent aiming automatically) and VISUAL HELP (agent providing visual information). These only appeared in the study presented in Chapter 7.



Figure 3.4: Early interface located at the bottom of the screen and used for our first study focusing on agent reliability and predictability.



Figure 3.5: Most up-to-date interface located at the bottom of the screen and used in studies 2, 3 and 4.

3.2.4 Task Difficulty

In our game, we controlled the task difficulty by modifying the following variables, also presented in Figure 3.6:

- Spawning rate of missiles (how often missiles appear from the top of the screen).
- Speed of each missile (in pixel/seconds, with faster speed requiring quicker decisionmaking to destroy).
- Type of missile (threat or non-threat), which was only added for our last study presented in Chapter 7. When describing missiles' types, we use the term "Threat" to indicate when a missile is heading toward a city and "Non-Threat" when it is not.

As our game is inspired by the 1980 arcade game "Missile Command" [10], we decided to use equivalent features to control the pace and difficulty of the task. Our decision to incorporate different difficulty levels was motivated by previous HCI studies that demonstrated how different levels of task difficulty can affect dependant variables such as cognitive load or task performance [1]. In our game, the difficulty of each level is defined by the delay between each missile spawning (spawn-rate) as well as their individual speed. A lower spawn-rate and faster missiles lead to a higher degree of difficulty. For consistency reasons, the delay mentioned above was not altered within a single round, but rather between studies in order to craft immersive and challenging scenarios. While these changes don't allow for between-studies comparisons in terms of difficulty levels, we always included a "no agent" (as in, *user-only*, without any agent) session in order to judge users' skills on their own. This assessment allowed us to monitor the individual level of performance of each user without the help of a collaborative agent, which in turn provided information about the perceived difficulty of the task. In the studies presented from Chapter 4 to Chapter 7, we often qualify each level of difficulty as "easy", "medium" or "hard" in order to compare results more easily.



Figure 3.6: Main variables controlling the task difficulty. Each variable can be modified to induce different levels of task complexity.

3.3 Collaborative Agents

3.3.1 Purpose & Interaction

In our experiments, we tested the impact of collaborative agents that varied in terms of their reliability and behaviour in an explicit human-agent collaborative task. Agents helped with the decision-making process by either guiding the crosshair towards targets (see Chapters 4, 5 and 6 and 7) or by providing visual information regarding the agent's reasoning and/or the environment of interaction (see Chapter 7).

We developed a series of agents each displaying different behaviours in order to study the resulting impact on the human-agent relationship and performance. As our studies focus exclusively on agents' behaviours through their in-game decisions, we decided not to include anthropomorphic elements such as human-like avatars or voice-enabled agents. Furthermore, findings from studies with anthropomorphic systems are usually "highly sensitive to the human individual in the loop" [p. 7] [59], which makes generalisation based on the results harder in studies with a limited amount of participants. Table 3.1 presents a summary of all different agents used in our studies. Depending on the individual study and its respective focus, variables were modified to affect one or more of the following elements:

- 1. The Agent's accuracy (how reliable it is, see Section 3.3.2).
- 2. The Agent's error pattern (how easy it is to anticipate its next errors, see Section 3.3.2).
- 3. The Agent's behaviour (the kind of overall behaviour that affects its capacity to detect and target missiles, see Section 3.3.3).

4. The Agent's error types (the amount of False Positive and False Negative errors the agent makes) (only used in the study presented in Chapter 7, see Section 3.3.3).

Study	Agent Codenames	Baseline Reliability	Error Pattern	Temporary Error Behaviour	Balanced Type 1&2 Errors
1	Agent 1.a	70%	Systematic Biases	None	False
	Agent 1.b	30%	Systematic Biases	None	False
	Agent 1.c	70%	Random Biases	None	False
	Agent 1.d	30%	Random Biases	None	False
	Agent 1.e	100%	None	None	False
2	Agent 2.a	70%	Random Biases	Lapses	False
	Agent 2.b	70%	Random Biases	Slips	False
	Agent 2.c	70%	Random Biases	Mistakes	False
	Agent 2.d	70%	Random Biases	None	False
3	Agent 3.a	80%	Random Biases	None	False
4	Agent 4.a	80%	Random Biases	None	True

Table 3.1: Summary of all agents used in our different studies with their associated levels of reliability and differences in the way they behaved and made errors.

3.3.2 Aiming & accuracy

Despite their differences in terms of reliability and behaviours, the process that each agent went through in order to acquire targets and aim at them was kept the same throughout all studies. The agent aiming accuracy or behaviour was modified depending on the research focus of each experiment, for instance by introducing different levels of reliability and predictability in Chapter 4 or varying the way the agents make errors in Chapter 5. This section presents the targeting process of each agent. The following steps take place when an agent registers a target:

- 1. Retrieves information about the target including its position and speed.
- 2. Retrieves information about the gun-tower's position and its projectiles' speed.

Information regarding positions are taken as 2D Vectors (with X and Y values), whereas information regarding speed are computed as integers (speed of missiles as pixels per second). Using the aforementioned information, the agent will then compute where best to aim in order to hit the target. A pseudo-script presented in Algorithm 1 details the procedure.

Algorithm 1 How an agent computes an optimal targeting position. The algorithmn is largely based on Pythagoras' theorem, as calculations are done in a 2D space.

^{1:} procedure GETOPTIMALAIMINGPOSITION(target, targetSpeed, projSpeed, projRotation, delta)

^{2:} Get the future position of the target at the next delta using its speed and rotation.

^{3:} Get the distance to the future position of the target.

^{4:} Compute the travel time for a projectile to hit the target given the distance to the future position of the target and the speed of the projectile.

^{5:} **Return** the position to move the crosshair to hit the target given the future position of the target and the travel time of the projectile.

Once the optimal target to hit a specific missile has been computed, a fixed bias can be applied in order to skew the agent's aiming in a random or biased direction, which in turn controls the agent's performance in targeting accuracy. This variance in the agent's performance was calculated using a random Gaussian distribution. Using a Gaussian distribution (also called normal distribution) ensures that the accuracy of the agent is controlled throughout the task in a consistent fashion. When computing new coordinates for the agent to move to, a fixed σ (randomly determined using a Gaussian distribution) is used to determine the level of the agent's performance. The greater the variance/ σ , the less accurate the agent's aim is, resulting in lower aiming accuracy. Figure 3.7 illustrates these changes.



Figure 3.7: Simplified representation of how different sigma values (noted σ) skew the agent's error margin, resulting in different levels of accuracy. The greater the value, the lower the agent's performance.

3.3.3 Target Awareness & Prioritisation

In addition to an agent's aiming accuracy, another key component of the agent's performance is its ability **to register and assign priorities to targets of importance**. For studies where all targets represented a threat (see Chapters 4, 5 and 6), the agents' errors were manufactured by varying the accuracy of each agent, but not their capacity to detect targets. In our final study (see Chapter 7), we introduced False Positive and False Negative errors, where the agent would register targets as non-threats when they were threats (False Negative error) and threats when they were non-threats (False Positive error). Figure 3.8 shows how these types of errors are distributed given a probability of 80% of missiles spawning as threats and 20% as non-threats.

Assigning priorities to targets is the other critical component of the agent's targeting system, and directly influences how reliable and predictable the agent is perceived to be by users. The first study (see Chapter 4) was designed with a simple priority system where the agent would target *the oldest targets* as each missile spawned one at a time, travelled at the same speed and was always aimed at a single city within one round. While this simple prioritisation system worked for the first study, it had to be modified in subsequent studies to take into account the presence of multiple missiles with varying speed.

For other studies (see Chapters 5, 6 and 7), multiple missiles spawned at the same time with varying speed, which called for changes in the way the agent prioritised targets. At this point, we implemented an algorithm that would assign priorities to targets based on their distance from the crosshair and their distances from the cities.

For the final study (presented in Chapter 7), we introduced false positive (missiles that do not hit cities). We decided to introduce an option to control how frequently (or not) the agent would aim at a false positive target. More details are available in Chapter 7. Below is a summary of all options that influenced the targeting awareness and prioritisation strategies of the agent, along with the Chapter where these options were present.

- Multiple missiles spawning at once (see Chapters 5, 6 and 7).
- Semi-randomised missiles' speed and targets (see Chapters 6 and 7).
- Non-threatening missile (i.e. not heading toward a city, to add False Positives to the task) (see Chapter 7).



Figure 3.8: Distribution of agent's errors per missile when manually inputting a target number of False Positive and False Negative errors. Here, the examples shows a balanced distribution of 50% False Negative and 50% False Positive errors.

3.4 Evaluation

3.4.1 Experimental Procedure

In this thesis, we used different dependent metrics to assess human-agent collaborative scenarios in real-time by modifying the difficulty of the task(s) and the behaviours of agents. We also defined dependent variables to monitor and record changes in the human-agent relationship via log-based metrics (recorded during the task) or survey instruments. Figure 3.9 presents an overall diagram of the procedure used for all studies presented in this thesis. In all our studies (presented in Chapters 4, 5, 6 and 7), participants were briefed on the experiment and asked to provide consent prior to undertaking study. After completing a demographic questionnaire and questions related to their gaming experience and trust in automation, participants were first given a short tutorial on how to play the game. In all studies, participants were instructed that their goal was to protect cities by destroying all incoming missiles and that agents were there to assist with aiming. Participants were informed that they could always correct the agents' aim if they so desired. To reduce the learning effect, the sequence in which participants interacted with each agent was randomised using a William Square design [193] in order to ensure consistency in all studies. At the end of the study, participants were compensated for their time with either a shopping voucher for lab-based studies or online payment for remote studies.



Figure 3.9: Experimental procedure using this framework. Both survey instruments (in blue) and the logging of behavioural metrics (orange) were used to assess the human-agent relationship at different points in the study.

3.4.2 Survey-based assessment

Most reported measures designed to assess the quality of the human-agent relationship were collected using an integrated survey system. Figure 3.10 provides an example of a question displayed during a task. The template can be modified and adapted to fit any other rating-scale based instrument. More details on the nature of analogue scale instruments are available in Section 2.6.1.2.

3.4.2.1 Trust

To measure reported trust in agents in our studies, we mostly relied on the questions designed by Jian et al. [90], available in Appendix E.2. In this survey, a series of statements has to be rated by participants in order to elicit different attitudes relating to trust in automation. In our work, we used either parts of Jian's scale (see Chapter 4 and 5) or a single item scale (see Chapter 6 and 7). In all studies, statements were presented at the end of tasks and required participants to report their trust on Likert scales with verbal anchors that denoted a "complete distrust in the agent" at worst (left) and a "total trust in the agent" at best (right).



Figure 3.10: Example of the in-game prompt used for self-reported measures (in the example: trust). When prompted, questions are displayed (one at a time) and participants have to rate statements according to verbal anchors. Numbers and extra labels can be added when needed. The pointer has to have been clicked at least once before the user can click on Continue.

3.4.2.2 Cognitive Workload

To evaluate cognitive workload, we used the validated NASA TLX survey instrument, available in Appendix E.3, which consists of 6 individual rating scales relating to Mental Demand, Physical Demand, Temporal Demand, Performance, Effort and Frustration. Each scale can be rated from 0 (low) to 21 (high). This method is the most widely used means of measuring cognitive workload [75]. In all studies, we focused on reporting the "RAW TLX" score, which consist of aggregated Nasa TLX transposed on a 100 point scale. This technique was shown to be as useful as weighted scores in a meta-review article by Cao et al. [18].

3.4.2.3 Situational Awareness

Situational Awareness can be described as the capacity to locate and remember important information to carry out a task and make predictions about future outcomes [48]. There are three commonly used methods to measure situational awareness, namely:

- 1. SAGAT (Situation Global Assessment Technique) [48],
- 2. SART (Situation Rating Technique) [173]
- 3. **SPAM** (Situation Present Assessment Method) [45].

All methods have pros and cons. While some require freezing the task (SAGAT), others require multi-tasking (SPAM), or the completion of a post-hoc survey (SART). A recent comparison of all these methods [52] found the SAGAT method to be the least intrusive, and least harmful in terms of users' task completion performance. In this thesis, we used the SAGAT method in Chapter 6 and SPAM method in Chapter 7. More details are presented in the relevant Chapters (6 and 7) regarding their choice and implementation.

SAGAT [49] consists in designing a set of SA related queries related to a specific task. During the experiment, the task is frozen, giving participants time to answer the queries. A numerical value is then derived from each question and aggregated into an overall score evaluating participants' overall SA. Figure 3.11 shows the SAGAT instrument as used in our framework.

SART [173] consists in administering a pre or post-hoc survey designed by Taylor et al. [173]. An example of the survey is available in Appendix E.4. The survey is composed of 10 questions spread across three domains: "Attentional demand" (3 questions), "Attentional supply" (4 questions) and "Understanding" (3 questions). Participants have to rate each statement on a 7-point Likert scale from "Low" to "High". There is also a 3-item version of SART called "3D SART" which comprises a single statement for each of the three dimensions.



Figure 3.11: Implementation of the SAGAT survey instruments to evaluate Situational Awareness at the first level (perception of data). When prompted, the task is immediately frozen and hidden in the background.

3.4.3 Behavioural metrics

3.4.3.1 Performance

During gameplay, various game-specific variables are monitored to record task-performance, both during sessions with an agent (team performance) and without (user-only performance). The variables we are interested in are computed with the amount of shots fired, missiles destroyed and shots missed. These variables are directly linked to the success of the task, with more missiles hit leading to a greater overall performance. Below are the metrics used to study performance with Recall, Precision and F1.

$$\mathbf{Precision} = \frac{\# Missiles Destroyed}{\# Shots Fired}$$
(3.1)

$$\mathbf{Recall} = \frac{\#MissilesDestroyed}{\#IncomingMissiles}$$
(3.2)

$$F1 = 2 * \frac{Precision * Recall}{Precision + Recall}$$
(3.3)

Higher precision indicates greater accuracy (fewer attempts to hit a target), while higher recall indicates greater task performance (less damage being sustained by the cities). F1 is the harmonic means of precision and recall and provides a combined measure of performance. The user control time was computed as the number of seconds participants were controlling the crosshair during each round (a greater user control time indicates less reliance on the agent).

3.4.3.2 Reliance

In our framework, reliance is measured via user control time, which is the amount of time for which the user controlled the crosshair in a scenario where an agent was present. The greater the time, the lower the reliance on the agent. Similar ways of measuring reliance have been used in other human-automation and human factors studies [14].

3.4.4 Data Visualisation and Statistical Testing

This section discusses the methodology and choices for the visualisation of quantitative data and statistical analysis performed in all of our experimental studies. Analysis was conducted using the Python 3 programming language [181] and several statistical libraries such as statsmodel [158], Scipy [184] and Pingouin [178].

Throughout our experimental studies, from Chapter 4 to 7, figures and tables were used to present quantitative results. In order to make data visualisation easier, boxplots were used to present quantitative results as they excel at clearly showing differences between experimental conditions [195]. An example of a boxplot with added explanations of its components is presented on Figure 3.12.



Figure 3.12: Example of a boxplot with a description of each of its components.

The remainder of this section presents the statistical methods used to compare experimental conditions. Before conducting any statistical comparisons, the Shapiro-Wilk test was used [162] to test the null hypothesis that samples came from a normally distributed population with a threshold for significance set at p = 0.05.

In the case of a normally distributed population, the following tests are conducted depending on the design of the study. For a between-groups design, a Levene's test [104] is used to determine whether the population is homoscedastic (equal variance) or not. If the homoscedasticity test is positive, an ANOVA [67] test is performed with follow-up pairwise Tukey-HSD tests [175], whereas if the homoscedasticity test is negative, a Welch ANOVA test [112] is conducted with follow-up pairwise Games-Howell [64] comparisons. For Within-group studies, repeated-measures ANOVAs [67] are used with follow-up pairwise T-tests. For mixed design studies (within and between groups), Mixed ANOVAs [67] are used with follow-up pairwise T-tests [199].

In the case of a non-normally distributed population, the following tests are conducted depending on the study design employed. For comparisons involving 2 groups, Wilcoxon signedrank [192] tests are used in the case of repeated measures, and Mann-Whitney U tests [123] in the case of non-repeated measures. For comparisons involving more than 2 groups, Friedman tests [164] were employed for repeated-measures, and Kruskal-Wallis tests [122] for non-repeated measures. In both cases, follow-up pairwise comparisons were realised using Wilcoxon signedrank tests for paired samples [199].

Regardless of how studies were designed, subsequent pairwise comparisons were only conducted when statistically significant results were found during previous statistical modelling. When statistically significant results are reported, Bonferroni corrections [9] were always applied. In addition, effect sizes were always reported using The Common Language Effect Size (CLES) for both parametric and non-parametric comparisons [121].

3.4.5 Coding

During most of our user studies (see Chapters 5, 6 and 7), we asked participants for feedback at the end of the experiment. We collected this feedback in accordance with the Critical Incident Technique [62] where participants were asked to describe one or more positive and negative aspects of their interaction in the study. Participants' written feedback was then given to three different PhD researchers to perform coding analysis on. For each coding task, specific codes were created by the lead researcher in accordance with the research questions and focus set for the study. When presenting results of coding analysis, Kappa scores [183] were used to assess the internal agreement between coders.

3.5 Study Modalities

3.5.1 Lab-based

Lab-based studies were conducted in the University of Strathclyde's CIS department, and involved one-on-one meetings with participants in offices equipped with computers to run the framework on, and xbox360-type controllers to permit interaction with the framework. Studies presented in Chapters 4 and 5 were conducted in a lab. For lab-based studies, a researcher was present to verbally explain and respond to any questions a participant might have, in addition to indicating when they had to answer certain post-hoc survey instruments.

3.5.2 Online

Online studies were conducted due to the worldwide COVID19 pandemic and our ensuing inability to conduct lab-based studies. Online studies involved participants recruited on the online crowd-sourcing Prolific[©] platform [33]. The Prolific platform was chosen thanks to its large participant pool and strict regulations to promote high quality data collection and attention to the task. Every online study was designed to be taking place without the need for a researcher to be present. For these studies, the game was exported to WebGL which most modern browsers can load and execute locally (via HTML5 and Javascript) by connecting to an executable of our study apparatus located on the Strathclyde's CIS servers. Contrary to lab-based studies where the same hardware was used for all participants, online-based studies dictated that participants' use their own computers, which led us to implement hardware checks to ensure consistent experimental conditions between all participants. A one minute benchmark-test was provided for free before advising participants to take part in the study. Only participants who experienced more than 24 frames per seconds (FPS) at a resolution of 720p (1280 x 720 pixels) or above were kept for further data analysis in our online studies. The frame-rate threshold was set based on audiovisual standards [191] while the resolution cap was set based on the first draft of the framework (see Chapter 4), which was designed at a signal format of 720p.

Compared to the lab-based studies, the online version required a more streamlined process where participants could easily navigate between the different steps of the experiment, as researchers were not present to assistant participants in tackling the study. Most of the time, the duration of experiments was shortened in order to foster better quality of data, and a benchmark was added to allow participants to test whether their machine could run the game properly, before taking part in the study. In addition, in-game performance was recorded to make sure that all participants experienced the game in comparable settings. Table 3.2 describes the main differences between the lab-based and online-version of our study.

Table 3.2: Comparison of the lab-based and online modalities of the framework.

Feature	Lab-based	Online
Access to pre,post-hoc and in-game survey	YES	YES
Access to the same game scenario	YES	YES
Logging of in-game performance	YES	YES
Benchmarking tool	NO	YES
Online logging of participants data	NO	YES
Researcher available to answer question(s)	YES	NO
Semi-structured interviews	YES	NO
Xbox controls	YES	NO
Keyboard controls	NO	YES

Part II

Investigating Collaborative Human-Agent relationships

Chapter 4

Agent Reliability and Predictability

4.1 Motivation

In this chapter, we explore how different levels of agent predictability and reliability influence users in a real-time task. As we have seen in Section 2.1, many features influence the propensity of a human operator to trust and rely on an automated agent. Past work has shown that an agent's performance (in terms of reliability) as well as an agent's behaviour (in terms of predictability) are positively correlated with trust [42, 129]. However, such studies have largely been conducted in turn-based settings [34, 131] where operators and agents interact asynchronously. Human-agent teams often work together in real-time scenarios where the trust relationship evolves over time and is affected by various factors such as task performance and agent behaviours [85]. Currently, there is a limited amount of work exploring the relationship between performance, predictability and trust when agents and humans work together in realtime collaborative settings. The study presented in this chapter answers our first Research Question, formulated in Section 1.3: How do changes in agent predictability (how easy it is to guess its next actions) and reliability (how good the agent is at the task) impact the human-agent relationship? Specifically, this work seeks to address the following sub research questions: How, at the same level of agent reliability, do changes in the agent's predictability affect:

- RQ1.a: the users' task performance when interacting with agents?
- **RQ1.b:** the users' reliance on the agent?
- RQ1.c: the users' cognitive workload when interacting with the agent?
- **RQ1.d:** the users' reported trust in the agent?

The work presented in this chapter is based on a previously published article entitled Inferring Trust From Users' Behaviours; Agents' Predictability Positively Affects Trust, Task Performance and Cognitive Load in Human-Agent Real-Time Collaboration [41]. Using the collaborative framework detailed in Chapter 3, we conducted a lab-based 5 (agents) \times 3 (levels of difficulty) within-subjects study with 30 participants who interacted with five agents exhibiting different levels of reliability (more or less reliable) and predictability (more or less predictable) in tasks of varying difficulty (easy, medium and hard). We found that, at the same level of agent reliability, the more predictably the agent makes errors, the greater the reported trust and task performance, and the lower the cognitive load.

4.2 Related Work

This Section presents relevant past work that has studied systems biases and trust in automated systems. An extensive number of studies have explored ways of measuring trust in systems (see the work of Schaefer et al. [155] for a comprehensive review). As we have seen in Section 2.1 and 2.3, most prior work has focused on assessing trust in relation to the agent's reliability. Less attention, however, has been paid to examining the effects of agent reliability *and* predictability in real-time human-agent collaborative tasks.

In HAC scenarios, agents are generally introduced to reduce users' cognitive workload, while trying to improve users' situational awareness and overall task performance [44, 56, 92, 170]. Past work has shown that an agent's *reliability* and its task *performance* heavily influence users' willingness to trust and rely on it. In their seminal work on the trustworthiness of command and control systems, Sheridan et al. [165] posit that the "effectiveness" (and subsequent reliability) of a system will influence how the operator perceives and trust a systems, and that it is possible to quantitatively measure the development of trust. The work of Robinette et al [146] showed that, in the case of human-robot coordination tasks, even a single error from a robot can strongly impact the development of a person's trust, regardless of how the robot fails at the task. A different study by Hoc et al. [84] on human-agent cooperation in a driving task showed that the type of automation error as well as its timing will have the most impact on how likely operators are to understand, trust and subsequently rely on the system. In their study, Hoc et al stress the importance of understanding the cognitive process involved in cooperating with automation, and how a system's failure can alter it.

In another command and control task focused on threat assessment, Fan et al. [57] tested different levels of agent variability (using systematic biases). They found that informing participants about the agent errors helped users to calibrate their trust accordingly, which led to higher task performance. However, too much information related to the agent's errors can quickly overload users. In a related work, Chavaillaz et al. [24] investigated different levels of agent reliability on trust, reliance and overall task performance in a turn-based X-ray scanning scenario. Their results showed that, as agent reliability decreased, so too did trust in agents. Furthermore, they found that perceived reliability (how much a person is willing to rely on an agent's input) is also affected by the capabilities of the automated system. In their studies, users' perception of the reliability of agents was more accurate when interacting with low-performing agents, compared to high-performing ones. In addition to studies focusing on different degrees of reliability, the work of Shirado et al. [166] explored turn-based coordination problems and found that error-prone agents (up to 30% loss in accuracy) could be beneficial to collaborative performance as their presence reduced the probability of a complacent attitude developing towards the agent.

Given the evidence of past research, it is clear that the performance of an agent (its reliability) as well as the agent's behaviour (its predictability) impact trust, but it is unclear how precisely both of these concepts influence it. Past work in human-agent interaction [24, 57] linked higher predictability and agent reliability to higher reported trust in the agent. In this Chapter, based on these previous insights, we are going to study how human-agent collaboration evolves based on the following working hypotheses:

- H1 At the same level of agent reliability (performance), agents exhibiting systematically biased behaviours (errors committed in a more predictable and consistent fashion) will be trusted more than agents exhibiting randomly varied behaviours (errors that are unpredictable and committed in an inconsistent way).
- 2. H2 As seen in previous HAI work, we hypothesise that it is possible to use behavioural data from human-agent interactions to model and infer users' perceived trust in agents.

The main contribution of our work lies in testing the impact of different degrees of agent reliability on the human-agent trust relationship in real-time scenarios. We use interaction data to model and determine how accurately reliance, agent reliability and performance can help us predict trust in automation.

4.3 Method

To study agent reliability and predictability, we used our interactive human-agent framework described in Chapter 3 in a lab-based study the modality of which is outlined in Section 3.5.1. As the focus of this study is on the explicit impact of predictability and reliability, we designed five different agents that each varied in their *targeting behaviours*. These behaviours were controlled to create different levels of reliability and predictability. The resulting experiment is a 2x2 within subjects design with 2 levels of agent reliability (low and high) and 2 levels of agent predictability (more or less predictable). We also included "no agent" and "perfect" agent conditions serving respectively as the baseline (no support from an agent) and upper bound (highest agent reliability) for the experiment.
4.3.1 Agent Reliability

Agents were designed with different level of reliability, which directly affected how good they were at hitting targets. All agents had a certain degree of variance in their aiming accuracy such that, for a given target, a certain degree of error would be applied to the targeting. This variance in the agent's performance was calculated using a random Gaussian distribution with a fixed σ (sigma) for each level of the agent's performance. The greater the variance (and thus the σ), the less accurate the agent's aim, leading to worse reliability and task performance (see Figure 4.1).

4.3.2 Agent Predictability

In addition to variance, some agents had their aim systematically biased towards a particular direction: (i) always above and to the right of their target, (ii) always below and to the left, (iii) always above and to the left, (iv) always below and to the right. The direction of the systematic bias was randomly selected for each participant at the beginning of the experiment and kept constant during the session. By randomly selecting the direction of the bias for each participant, we ensured that our findings were not constrained by a specific type of systematic bias. By setting specific fixed biases, participants could learn to anticipate the error committed by the agents. This systematic bias impacted the agents' targeting behaviours, but *not* their performance, which were only impacted by random variance.

Agent performance was calibrated using simulations where agents completed the task by themselves (e.g. the same task without users). In these simulations, we calculated the agent's performance based on the Recall scores described in Section 3.4.3.1. We then used T-tests to ensure that the performance of agents with a similar level of reliability were not significantly different. This was to make sure that high or low degrees of predictability would not impact agents' reliability, thus allowing comparisons. While comparing the Recall scores of agents Alpha and Gamma (low performing agents), a t-test yielded p > 0.05. Similarly, t-tests performed using the Recall scores of agents Beta and Delta also yielded p > 0.05. Agents Beta and Delta were tuned to be high performance (approx. 0.7 Recall scores or 70% of the targets being hit), while agents Alpha and Gamma were tuned to be low performance (approx. 0.3 Recall scores, or 30% of the target beings hit).



Figure 4.1: Visualisation of the different biases applied to the agents in the study (not to scale). The greater the bias, the lower the accuracy of the agent. For the systematic bias, a quadrant is randomly chosen for each participant at the beginning of a session. Low systematic bias and low random variance or high systematic bias and high random variance result in the same performance output.

4.3.3 Agent configurations

Using different combinations of levels of agent reliability (low and high) and predictability (more or less predictable), we designed five different agents. Agent names were introduced to make it easier for participants to refer to a particular agent. Agents *Alpha* and *Beta* were designed to be more predictable with respectively a high (Alpha) and low (Beta) level of performance. Agents *Gamma* and *Delta* were designed to be less predictable with respectively a high (Gamma) and low (Delta) level of performance. In addition to the aforementioned agents, we also included a *perfect* agent: "Epsilon" which exhibited no bias and no variance – and thus had the highest reliability and predictability out of all of the agents.

Figure 4.2 shows the different combinations of agents used, which we refer to as: Alpha, Beta, Gamma, Delta and Epsilon (A,B,C,D,E). Here is a summary list of the agent configurations used in this study:

- No Agent.
- Agent Alpha: high performance, low predictability.
- Agent Beta: low performance, low predictability.
- Agent Gamma: high performance, high predictability.
- Agent Delta: low performance, high predictability.
- Agent Epsilon: Highest performance and predictability.



Figure 4.2: Simplified representation of all agents used in this study and their respective aiming patterns. Both agents displaying either HIGH or LOW performance are programmed with the same accuracy, regardless of their aiming patterns.

4.3.4 Task Difficulty

During each interaction with an agent, participants went through three rounds which lasted for 90 seconds each. This duration was set so that participants had enough time to familiarise themselves and adapt to the agents, while ensuring that the experiment could be completed within an hour (reducing participants' fatigue). Each round increased in difficulty (going through "Easy", "Medium" and "Hard" levels of difficulty). Here are the details about task difficulty in this study:

- In the "Easy" level, missiles spawned every 4 seconds at a speed of 100 pixels per second for a total of 22 missiles.
- In the "Medium" difficulty level, missiles spawned every 2 seconds with a speed of 150 pixels per second for a total of 45 missiles.
- In the "Hard" difficulty level, missiles spawned every second with a speed of 200 pixels per second for a total of 90 missiles.

The speed and number of missiles in each level were calibrated during pilot testing with 10 participants, to make sure that changes in difficulty were noticeable without completely overwhelming participants (see Section 4.3.5 for a more detailed description of the pilot study).

4.3.5 Piloting

Before conducting the main study, a formal pilot experiment was carried out. Ten participants were recruited from our local Computer Sciences department. This pilot experiment focused on calibrating the single player (no agent) experience, as well as core gameplay elements such as the controls, visuals and overall difficulty of the game.

To evaluate participants' performance, we used F1 scores. F1 is a metric related to participants' overall task performance and is computed using the number of missiles participants hit, the number of shots fired and the total number of missiles present in each level. For more information, all of the performance metrics are detailed in Section 3.4.3.1. F1 scores varied between **0.88** for the "Easy", **0.77** for the "Medium" and **0.46** for the "Hard" difficulty levels. As the difference in terms of participants' performance between the "Easy" and "Medium" levels was low (a loss of **0.11** for F1 scores), we decided to increase the speed of missiles in the "Medium" difficulty level to intensify its difficulty.

4.3.6 Independent and dependent variables

In this section, we summarise the independent and dependent variables used in this study and as motivated by our experimental method and research questions.

Our independent variables are the following:

- Task difficulty, as defined by the amount and speed of missiles in each level.
- Aiming agent reliability, how accurate the agent is in its predictions (low, high or nearperfect reliability).
- Aiming agent predictability, how predictable the agent's actions are (low or high).

Our dependent variables are the following:

- Task Performance, in terms of missiles hit, shots fired and missile missed.
- *Reliance*, expressed by the duration for which participants relied on the aiming agent's help.
- Trust, as reported by participants.
- Cognitive Workload, as reported by participants.

4.3.7 Experimental Procedure

Ethics approval for this study was obtained from the University of Strathclyde's Department of Computer and Information Sciences (Approval No. 793). The study took place in a lab located in the University of Strathclyde Computer Sciences Department. The experimental procedure is outlined in Section 3 and in Figure 3.9. The study lasted for approximately an hour, and participants were compensated for their time with a shopping voucher worth £10. After being briefed on the experiment and asked to provide consent, participants went through the following steps:

- 1. Demographic questionnaire and pre-hoc surveys. (5 minutes).
- 2. Tutorial with and without an agent. (2 minutes)
- 3. Session without an agent, to record individual performance. (4 minutes 30 seconds).
- 4. One session with each agent (Alpha, Beta, Gamma, Delta, Epsilon). (4 minutes 30 seconds each).
- 5. Post-hoc surveys. (5 minutes).

The sequence in which participants interacted with each agent was randomised using a William Square design in order to mitigate learning effects [193]. During each session, participants played through three rounds of *low* to *medium* and *high* levels of difficulty. At the end of each round, participants were asked to rate their trust in the agents using a subset of the "Checklist for trust in Automation" created by Jian et al [90]. At the end of each session, participants were asked to complete the NASA Task Load Index (TLX) questionnaire [75]. Survey instruments are present in more detail in Section 3.4.2.

4.3.8 Demographics

Participants were recruited through mailing lists and flyers posted on our university campus. We recruited a total of 30 participants (14M, 16F) with ages ranging from 19 to 38 years old ($M = 27 \pm 5.19$). Most participants were enrolled as postgraduate students. When asked how often they played video-games on a 6 point Likert scale from 0 (never) to 6 (Very Frequently), participants scored on average 3.63 (±1.85), indicating that they played "occasionally". Ratings from the Complacency Potential Rating Scale (CPRS) [167] were used to evaluate general attitudes toward automation. CPRS scores ranged between 55.57 and 90.84 ($M = 72.55 \pm 9.3$) which indicates that our sample consisted of participants who were more likely to rely on automation than not [167]. Overall, the distribution of scores was homogeneous enough that our sample could *not* be divided into different groups representing distinct attitudes toward automation. The pre- and post-hoc survey instruments presented to participants on the Qualtrics platform are available in Appendix A.

4.4 Results

In this section, we present results regarding task performance, users' reliance on agents, reported trust in agents and cognitive workload. We used the overall scores participants obtained at the end of each session, across all levels of difficulty. More details on the inclusion of different difficulty levels are available in Section 3.2.4. The statistical methods we used to compare and report results are detailed in Section 3.4.4.

Table 4.1: Metrics related to performance (Recall, Precision and F1, higher scores = better performance) and reliance (User control time (in seconds) higher = less reliance on the agent). Superscript letters next to the results indicate which agents yielded significantly worse scores (p < 0.05). Highest values are highlighted in bold.

	No Agent	$\begin{array}{c} {\rm Agent \ Alpha} \\ {\rm rel} + / {\rm pred} + \end{array}$	Agent Beta rel-/pred+	Agent Gamma rel+/pred-	Agent Delta rel-/pred-	Agent Epsilon highest reliability
Recall	0.64 ± 0.03	0.82 ± 0.02	0.60 ± 0.03	0.72 ± 0.02	0.58 ± 0.03	0.98 ± 0.01
Precision	0.57 ± 0.02	0.60 ± 0.02	0.50 ± 0.03	0.53 ± 0.02	0.47 ± 0.02	0.86 ± 0.01
F1	0.60 ± 0.02	0.68 ± 0.02	0.54 ± 0.03	0.60 ± 0.02	0.51 ± 0.03	0.91 ± 0.01
User Ctrl Time	25.12 ± 0.96	5.34 ± 0.83	24.18 ± 1.16	10.61 ± 1.09	27.68 ± 1.29	1.02 ± 0.43

4.4.1 Performance

To measure task performance, we computed Recall, Precision and F1 scores based on the number of shots fired, missiles hit and total missiles present in each level of our experiment. Recall, Precision and F1 scores are detailed in Section 3.4.3.1. Table 4.1 and Figures 4.3, 4.4 and 4.5 show the average task performance achieved by participants with each individual agent. These scores are averaged over all three levels of difficulty. If we consult Table 4.1, we can see that participants achieved better Recall scores while interacting with high reliability agents (Alpha and Gamma) than on their own (without an agent). Participants performing poorly in the no agent session benefited the most from this increase in performance. As expected, participants performed the best with agent Epsilon (highest reliability and predictability) than with any other agents, across all measures. When interacting with Alpha (high reliability, high predictability) and Gamma (high reliability, low predictability), participants were able to achieve higher precision scores than on their own (no agent), but performed worse with Beta (low reliability, high predictability) and Delta (low reliability, low predictability) across all performance measures, yielding lower Recall, Precision and F1 scores. A repeated-measure ANOVA yielded significant results for Recall scores (p < 0.0001, F = 51.26, $np^2 = 0.64$), Precision scores (p < 0.0001, F = 59.01, $np^2 = 0.67$). Follow-up pairwise comparisons using pairwise T-tests yielded significant results between Alpha (high reliability, high predictability) and Gamma (high reliability, low predictability) for Recall (p < 0.0001, T = 6.36, CLES = 0.72), Precision (p < 0.0001, T = 4.63, CLES = 0.64) and F1 (p < 0.0001, T = 5.42, CLES = 0.67).



Figure 4.3: Recall scores for each session. A higher score indicates a better performance. Recall scores give a measure of how many missiles participants and agents hit. We can see that agent Alpha (high predictability and reliability) led participants to achieve higher median Recall scores than agent Gamma (low predictability and high reliability).



Figure 4.4: Precision scores for each session. A higher score indicates a better performance. Precision scores give a measure of how effective, in terms of shots fired per missile hit, participants were at the task. Overall, precision scores are higher for more predictable agents.



Figure 4.5: F1 scores for each session. A higher score indicates a better performance. F1 is a harmonised mean of Recall and Precision scores. Overall, F1 scores are higher for high predictability agents.

4.4.2 Reliance

To measure to what extent participants relied on an agent, we computed the duration for which each participant controlled the crosshair. Participants controlling the crosshair for a longer period of time suggested that they relied on the agents less (and vice versa). Table 4.1 and Figure 4.6 show the average amount of time (in seconds) participants spent in control of the crosshair (denoted as *User Ctrl Time*). As expected, we observed that participants spent less time controlling the crosshair when working with Epsilon (highest reliability and predictability) compared to any of the other conditions, with or without agents. In addition, participants spent more time controlling the crosshair when collaborating with low reliability agents (Beta and Delta) compared to high performance agents (Alpha and Gamma). A Friedman test yielded significant results (p < 0.0001, W = 0.85) when comparing the overall user control time between sessions. Follow-up non parametric pairwise T-tests indicated that participants relied on the agent significantly more when interacting with agent Alpha (high reliability, high predictability) than with agent Gamma (high reliability, low predictability) with p < 0.0001, U = 250, CLES = 0.28.



Figure 4.6: Amount of time each participant spent correcting the agents. A longer duration indicates less reliance on the agents. At at high level of agent reliability, participants corrected the more predictable agent (Alpha) less than the less predictable agent (Gamma).

4.4.3 Trust

To measure trust, we asked participants to rate their perceived trust in the agent with a series of trust-related questions graded on a scale of 1 to 11, with a lower score indicating a lower reported trust in the agent. While multiple elements were used, we focused our analysis of trust on the ratings associated with the statement "I can trust the agent" presented in Table 4.2 and Figure 4.7. On consulting the results we notice that, on average, participants trusted agent Epsilon (highest reliability and predictability) more than any of the other agents, which was expected as its reliability was the highest. In addition, trust ratings of agents with low reliability (Beta and Delta) were on average much lower than agents with high reliability (Alpha and Gamma). When comparing answers pertaining to the reported trustworthiness of agents, a Friedman test yielded significant results (p < 0.0001, W = 0.90). While performing follow-up comparisons using Wilcoxon signed-rank tests, we found that participants rated Alpha

(high reliability, high predictability) significantly higher than Gamma (high reliability, low predictability) with $\rho < 0.0001$, U = 650, CLES = 0.72. However, no significant results were found when comparing Beta (low reliability, high predictability) with Delta (low reliability, low predictability). These results indicate that, at the same high level of agent reliability, participants were more trustful of an agent with high predictability (Alpha) than an agent with low predictability (Gamma).



Figure 4.7: Average reported trust in the agents. Higher scores indicate greater trust in the agents. At a high level of agent reliability, participants reported a higher trust in the more predictable agent (Alpha) than the less predictable one (Gamma).

Table 4.2: Metrics related to reported trust and cognitive load. Higher scores indicate greater agreement with the statements. Highest values are highlighted in bold.

Question	Bot Alpha	Bot Beta	Bot Gamma	Bot Delta	Bot Epsilon
I can trust the system	4.80 ± 0.26	1.27 ± 0.10	3.57 ± 0.27	1.47 ± 0.16	6.70 ± 0.13
I enjoy interacting with the system	5.20 ± 0.26	1.87 ± 0.22	4.53 ± 0.29	2.03 ± 0.26	6.13 ± 0.27
I am suspicious of the systems intent, action or outputs	3.37 ± 0.32	5.13 ± 0.37	3.83 ± 0.33	5.67 ± 0.35	2.30 ± 0.42
The system's actions will have a negative outcome	3.13 ± 0.28	5.70 ± 0.32	4.00 ± 0.27	6.20 ± 0.19	1.90 ± 0.34
The system provides security	4.30 ± 0.24	1.37 ± 0.13	3.80 ± 0.28	1.53 ± 0.16	6.37 ± 0.24
The system is reliable	4.13 ± 0.30	1.33 ± 0.11	3.47 ± 0.30	1.70 ± 0.23	6.67 ± 0.14
The system is very unpredictable, I never know how it's going to act from one moment to another	3.33 ± 0.33	5.80 ± 0.34	4.17 ± 0.30	5.53 ± 0.36	2.07 ± 0.35
How mentally demanding was the task?	51.90 ± 4.43	71.27 ± 3.99	62.86 ± 3.68	79.05 ± 2.81	23.02 ± 3.24
How hurried or rushed was the pace of the task?	51.43 ± 3.95	73.02 ± 2.77	60.32 ± 3.05	78.25 ± 2.36	34.44 ± 4.78
How successful were you in accomplishing what you were asked to do?	65.56 ± 3.50	44.29 ± 5.26	57.46 ± 4.08	41.11 ± 4.95	92.38 ± 2.59
How hard did you have to work to accomplish your level of performance?	52.38 ± 3.65	76.83 ± 3.25	62.06 ± 2.73	80.79 ± 2.31	24.60 ± 4.10
How insecure, discouraged, irritated, stressed and annoyed were you?	34.60 ± 3.68	66.51 ± 5.22	42.86 ± 4.80	66.67 ± 5.29	19.52 ± 4.51
Overall RAW TLX score	51.17 ± 2.37	66.38 ± 2.15	57.11 ± 2.22	69.17 ± 1.99	38.79 ± 2.67

4.4.4 Cognitive Load

To measure Cognitive load, we used the Nasa TLX survey instrument detailed in Section 3.4.2.2. Higher scores indicate a greater reported workload. As presented in Table 4.8 and Figure 4.2, we observed that participants reported a much lower cognitive load (NASA TLX scores) when interacting with agent Epsilon (highest reliability and predictability) than with any of the other agents. Furthermore, participants reported a much higher cognitive load when interacting with low reliability agents (Beta and Delta) than with high reliability ones (Alpha and Gamma). When comparing overall Raw Nasa TLX scores, a repeated-measure ANOVA yielded significant results (p < 0.0001, F = 85, $np^2 = 0.75$). While performing pairwise T-test comparisons, we found that participants perceived the high reliability, high predictability agent (Alpha) as significantly less cognitively taxing than the high reliability, low predictability agent (Gamma) with p < 0.0001, T = -2.96, CLES = 0.40. In addition, participants found the agent with low reliability and low predictability (Delta) to be significantly more cognitively taxing than the agent with low reliability and low reliability and high predictability (Beta) with p < 0.0001, T = -2.07, CLES = 0.40.



Figure 4.8: Raw NASA TLX ratings for each session with agents. Higher scores indicate greater cognitive loads. Raw TLX scores are lower for participants that interacted with the high reliability and high predictability agent (Alpha) compared to the high reliability and low predictability agent (Gamma).

4.4.5 Predicting Trust

In our Research Questions presented in Section 4.1, we sought to understand how different variables influenced reported trust. According to our working hypothesis presented in Section 4.2 which posited that trust can be predicted using behavioural information, we analysed correlations between trust ratings, task difficulty, the reliance metric (user control time), cognitive workload (NASA TLX scores) and performance metrics (Precision, Recall and F1 scores). From Table 4.4, we can see that participants' reliance on the agents (as measured by user control time) led to the highest correlation ($\rho = -0.801$, p < 0.001) followed by Cognitive Load (Raw TLX scores) with $\rho = -0.730$, p < 0.001, whereas performance metrics (Recall, F1 and Precision) resulted in lower correlations ranging from 0.50 to 0.61.

In addition to analysing correlations between our main variables, we created multiple linear regression models to determine which combinations of factors led to the best predictions of users' trust ratings. The selection criteria for the variables used in our models were based on the work of Hoff et al. [85], where elements related to the development of trust are categorised according to their impact on trust prior or during the interaction with an agent. Table 4.3 shows the combination of factors, mean square error, and adjusted correlation coefficients for each models. Our results show that the best performance for predicting trust ratings ($R^2 = 0.915$) were achieved by combining measures related to reliance (user control time), performance (the number of shots fired, missiles destroyed and misses), task complexity and information related to the participants' age and reported gender. These results corroborate the findings from [85] where elements captured during the interaction (such as performance and reliance related to "Dynamic Learned Trust" [85]) coupled with elements captured prior to the interaction (such as age and gender related to "Dispositional Trust" [85]) help us understand and be more accurate in our prediction of reported trust in the agent, even though each variable does not contribute equally to the overall quality of the predictive models.

Table 4.3: Linear regression results when predicting participants' trust ratings by using contextual (difficulty) and behavioural measures (performance and reliance). Only the most important results are presented. A higher \mathbb{R}^2 value indicates more accurate predictions.

Parameters	Total Mean Square Error	Adjusted \mathbb{R}^2
User Ctrl Time, Precision, Recall, F1 Difficulty, Raw TLX, Gender, Age	2491.9	0.915
User Ctrl Time, Precision, Recall, F1, Difficulty, Raw TLX	3244.1	0.894
User Ctrl Time, Precision, Recall, F1, Difficulty	3890.0	0.893
User Ctrl Time, Precision, Recall, F1	4717.9	0.867
Recall	17253.2	0.793
F1	16994.9	0.781
Precision	16666.6	0.766
Age	14927.5	0.686
Gender	13634.2	0.626
Difficulty	12517.2	0.575
Raw TLX	7796.8	0.357
User Ctrl Time	1830.2	0.082

Parameter 1	Parameter 2	ρ	p-value
User Control Time	Trust ratings	-0.801	< 0.001
Raw TLX	Trust ratings	-0.730	< 0.001
Recall	Trust ratings	0.614	< 0.001
F1	Trust ratings	0.553	< 0.001
Precision	Trust ratings	0.501	< 0.001
Age	Trust ratings	0.080	0.092
Difficulty	Trust ratings	-0.079	0.094
Gender	Trust ratings	-0.018	0.698

Table 4.4: Spearman's correlation tests between behavioural or reported metrics and trust ratings. A higher ρ scores indicates greater correlation.

4.5 Discussion

In this study, we have explored how agent predictability and reliability influence users' perception of agents in terms of cognitive workload and trust, as well the implications on task performance. By conducting this study, we have provided a better understanding of the respective roles of agent reliability and predictability in human-agent collaboration and we have added to past work that mostly focused on testing different levels of systems reliability in collaborative tasks. This work helps us to answer our first Research Question: **How do changes in agent predictability (how easy it is to guess its next actions) and reliability (how good the agent is at the task) impact the human-agent relationship?** from the overall research questions elicited in Section 1.3. We sought to answer the sub-research questions presented in Section 4.1: **How, at the same level of agent's reliability, do changes in the agent's predictability affect**:

- RQ1.a: the users' task performance when interacting with agents?
- **RQ1.b:** the users' reliance on the agent?
- **RQ1.c:** the users' cognitive workload when interacting with the agent?
- **RQ1.d:** the users' reported trust in the agent?

With this work, we have found that interacting with more predictable agents at a high level of agent reliability positively contributes to the human-agent relationship in terms of task performance, reliance on the agent, reported trust in the agent and cognitive load. Additionally, we have also explored how different metrics correlate with reported trust in the agent. The next sections discuss the implications of our findings related to the sub-research questions outlined in Section 4.1.

4.5.1 Predictability in the agent's actions

We hypothesised that, at the same level of agent reliability, more predictable agents would be perceived as more trustworthy than less predictable ones. We found this hypothesis to be true, but only for agents with high levels of reliability. When looking at task performance, we found that interacting with a nearly perfect agent (agent Epsilon) led participants to achieve higher performance and to view the agent more positively in a general sense, which was to be expected from a highly reliable agent. When comparing the rest of the agents, however, clear differences in users' behaviours and perceptions were found.

With our sub-research questions presented in Section 4.1, we set out to explore how agent predictability impacted performance, reliance, workload and trust. When comparing the agents, we noticed that participants interacting with low reliability agents (Beta and Delta) yielded poor overall task performance, even worse than when participants carried on with the task without any agent (which informs RQ1.a). These results were the lowest across all performance indicators: F1, Recall and Precision (see Table 4.1). Moreover, participants had to compensate more for the agents' inaccuracy, as is evidenced by higher user control times, greater reported workload and lower trust ratings (which informs RQ1.b and RQ1.d). Nevertheless, when comparing agent Beta (low reliability, high predictability) to agent Delta (low reliability, low predictability), we found that participants performed slightly better with agent Beta, in addition to spending slightly less time correcting the agent and reporting significantly lower cognitive workloads (which informs RQ1.c), even-though this was not statistically significant. This suggests that when an agent makes errors in a systematic, predictable way, participants are able to compensate for its inaccuracy better.

When comparing agent Alpha (high reliability, high predictability) to agent Gamma (high reliability, low predictability), we found that participants achieved significantly higher performance with Alpha. They also corrected agent Alpha significantly less and reported significantly lower workload. These results further suggest that when an agent's behaviour is more predictable, participants could not only better compensate for the agents' imprecision, but also *adapt* and *work* with the agent better, resulting in an overall better task performance.

Overall these findings suggest that, in the case of imperfect automation, predictability in the way an agent makes errors is important. That is to say, even if it makes a number of errors, an agent with high predictability allows users to adapt better and quicker to its behaviour, which results in higher reported trust in the agent, as well as better task performance and reduced cognitive load.

4.5.2 Factors influencing trust

We further hypothesised that it is possible to infer trust in an agent using information collected during human-agent interactions. To investigate this area, we first sought to determine which factors were the most important to predict participants' perceived trust in agents. Table 4.4 shows correlations between trust ratings and other variables monitored in our study. While previous work hypothesised that performance is the most important predictive factor regarding users' trust in agents [85], our results show that the different performance indicators used in our study (F1, Recall and Precision) correlate only moderately with trust ratings. Moreover, our findings reveal that reliance, expressed by the amount of time users spent correcting the agents, was the metric most correlated with trust, which is in line with previous work [46,103]. However, we found that cognitive load (in the form of Raw Nasa TLX scores) was more strongly correlated with users' reported trust in the agents than task performance. This finding is consistent with other work that focused on predictive decision-making, where cognitive load was found to be affected by trust, reliance and the overall difficulty of the task [6, 201]. To further explore which combinations of factors could predict trust ratings best, we performed several multilinear regressions. We achieved the best results (see Table 4.3) by using data related to users' reliance on the agents, performance scores and the difficultly of the task. These findings suggest that it is important to consider both performance and reliance metrics in order to infer users' trust in an agent more effectively. Moreover, we demonstrated that it is possible to predict users' trust ratings with a high degree of accuracy.

4.6 Conclusion

In this study, we set out to explore the relationship between trust, agent predictability and agent reliability in a real-time collaborative scenario. To achieve this, we designed a withingroups study where participants completed a series of aiming tasks with the help of different collaborative agents. Our findings were aimed at answering our first Research Question: "How do changes in agent predictability (how easy it is to guess its next actions) and reliability (how good the agent is at the task) impact the human-agent relationship?"

We found that, at the same level of performance, participants reported higher levels of trust in agents that were more predictable than less predictable agents. However, as the agents' reliability decreased, participants were less trusting of them, regardless of their predictability. In addition, participants achieved better performance and reported lower cognitive load with systematically biased agents compared to agents with more variance, especially when agents' performance level was high.

These findings further highlight the importance of predictability and consistency in the design of potentially error-prone agents, and how it impacts human-agent collaboration in realtime. Furthermore, our study investigated whether it was possible to infer trust ratings based on participants' interactions. Our findings show that while performance indicators are important, in the context of real-time collaboration, participants' reliance on agents is a better predictor of trust. These findings suggest that it is possible to develop methods that can monitor trust in automation over time, and that such methods could be used by agents to better adapt to individual users. However, to develop trust-aware agents, we need to find ways of recording real-time operationalised metrics of reliance, and explore what other factors, beyond system reliability, affect users' perceptions of agents.

Chapter 5

Agent Errors and Behaviours

5.1 Motivation

In this Chapter, we are focusing on how different types of agent errors can influence human-agent collaboration. We designed a study where participants interacted with different automated agents that performed similarly, but made errors in different ways. We then analysed how different types of agent errors, at the same level of agent reliability, influence the human-agent partnership.

Our understanding of imperfect agent behaviours and its impact on trust is limited. As we have seen in Chapter 4, the behaviour of an agent and in particular the predictability of its actions can greatly influence how users will perceive and be willing to engage with it. When studying agent behaviours, most past work has focused on agent reliability and its impact on task performance, but less on the different ways in which an agent can fail at a task. The work of Marinaccio et al. [118] defined different error types, describing how each was identified and investigated in past work, and how best to mitigate their negative effect on users. Largely inspired by their taxonomy, we decided to run a follow-up study to the work presented in Chapter 4 and explore how different agent behaviours impact the human-agent relationship within our human-agent collaborative framework. More details on our usage of their taxonomy is available in Section 5.3.1.

With this work, we address our second Research Question, presented in section 1.3: How do different types of agent errors defined from previous related work such as slips, mistakes and lapses affect the human-agent relationship? As in our study on agent reliability and predictability in Chapter 4, we sought to investigate how different types of agent errors influence:

- RQ2.a: the users' task performance when interacting with agents?
- **RQ2.b:** the users' reliance on the agent?
- **RQ2.c:** the users' cognitive workload when interacting with the agent?

• **RQ2.d:** the users' reported trust in the agent?

The work presented in this chapter is an extension of the paper entitled Impact of Agents' Errors on Performance, Reliance and Trust in Human-Agent Collaboration [37].

We conducted a lab-based 4 (types of agent errors - no error, slips, mistakes and lapses) x 2 (difficulty levels - easy or hard) within-subjects experiment with 24 participants. The main contribution of this work lies in testing whether, at the same level of agent reliability, different types of agent errors have a noticeable impact on users. Our results show that, when agents perform the same, agent errors are perceived differently and change the way participants interact with agents. For instance, slips and mistakes are more harmful to performance than lapses while slips are more harmful to reliance than mistakes.

5.2 Related Work

5.2.1 Agent Reliability and Errors in HAI studies

This Section presents relevant past work that has studied trust in agents and, in particular, how to categorise the way automated systems can fail at a task. In most human-agent collaboration scenarios, particularly safety-critical scenarios, it is assumed that agent reliability is one of the most important factors that will contribute to an individual's propensity to trust and rely on an agent, alongside other factors such as "stress levels" or "task complexity", as described by Grawbowski et al. [73]. The work of Hoc et al. [84], investigated agent support in a driving scenario and stated that "if human-machine interference is perceived as costly, without gain in terms of performance, the driver may unconsciously by-pass automation" [82]. The relationship between agent performance and trust has been investigated by manipulating agent behaviours in several ways. For instance, while Correia et al. experimented with an agent that suddenly stop working in a collaborative game assignment [34], other studies experimented with falsealarms errors in a human-agent collaborative X-ray scanning task [125], or introduced systematic biases in an agent's decision-making capabilities during a collaborative control-and-command scenario. [57]. Despite taking place in different environments with dissimilar populations, both the aforementioned studies discussed the idea of "trust aware" agents, where transparency goes both ways and where the user and agent calibrate their expectations iteratively, over time. However, little attention was brought to specific types of agent failure, and how they impact the human-agent team decision-making process.

5.2.2 Defining errors

Any flaw in the decision-making capabilities of an agent that results in loss of performance can be understood as an "error". The work of Salem et al. [150] found that the type of error an agent makes (breach of privacy, violations etc.) has an impact on the way users perceive the agent, and will affect the extent to which users are willing to rely on it in subsequent interactions. In the context of Human-Agent interactions, "error" is a broad term as agents can err in different ways and for different reasons. The work of Marinaccio et al. proposes four distinct types of errors: mistakes, lapses, slips and violations [118] (see Table 2.1) derived from human-human interactions studies [145] in which different types of errors are defined by the violation they represent. While these errors all stem from studies focusing on human-human interactions, they were also conceptualised in the context of human-agent interactions, notably in healthcare by Kim et al. [94]. The framework proposed by Marinaccio et al. [118] is an important step towards a better understanding of users' relationships with automation systems for the following reasons:

- It provides definitions and examples of known automation failures based on prior work in both human-human and human-agent settings.
- It defines automation error in a context-specific manner, which provides researchers with a template for defining other agent-related incidents in different domains.
- It lists potential effective trust "repair techniques" to prevent further damage caused by automation failures. Each repair technique is justified by prior work, such as the healthcare study by Kim et al. [94].

In the HAI literature, a number of studies have empirically tested the effect of automation errors on human participants in an effort to try and understand how to best define and prevent their failures. The work of Baker et al. [11] provides a comprehensive review of the potential future avenues in HRI and HAI, and makes use of the framework by Marinaccio et al. [118] to highlight the necessity of studying context-specific definitions of automation errors and effective repair mechanisms. In their article, Baker et al. make relevant recommendations for future research, including "Adapting existing trust research to investigate how robots features affect trust" [11, p. 20]. Following on from this recommendation, we believe that more research should look at the explicit impact of different kinds of agent errors on the human-agent relationship, and how users react and adapt to these errors, over time. Our work seeks to investigate this area by using a framework that allows us to manipulate agent errors and monitor changes in reliance and performance during human-agent collaborative tasks.

Given evidence from past work that defined and assessed different kinds of agent errors as well as methods to repair their damage [118, 145], we expect error types to have an influence on users' behaviour and perception of the agent despite comparable levels of agent reliability. More precisely, we are going to study human-agent partnership under the following working hypotheses:

1. **H1** Agent errors will always have a negative impact on task performance, no matter their type.

2. H2 Agent errors that are harder to detect (mistakes) will have a greater impact on task performance and reliance than agent errors that are easier to anticipate (lapses).

The main contribution of our work lies in testing different types of agent errors and assessing their influence on human-agent collaboration in terms of behavioural changes and perception of the agents.

5.3 Method

As with all studies conducted in this thesis, we used the human-agent collaborative framework presented in Section 3 in its lab-based form, as described in Section 3.5.1 where we modified and created new agent behaviours and errors to answer our research questions. In this iteration of our framework, each participant interacted with 4 different agents (within-groups study) in 2 levels of difficulty, where each agent displayed one particular type of erroneous behaviour defined as slips, lapses, mistakes or none. More information about agent errors is available in Section 5.3.1).



Figure 5.1: Abstract representation of the different errors each agent made. Mistakes are errors of priority, Lapses result in no response from the agent and Slips are experienced as an extreme loss of accuracy.

5.3.1 Agent Errors

The focus of this study is in testing the effect of agents displaying different behaviours and making different errors defined as "slips", "mistakes" and "lapses", all derived from the taxonomy of errors provided in Table 2.1, which is inspired by the work of Marinaccio et al. [118]. As further detailed in Section 5.2, we rely on the taxonomy of agent error by Marinaccio et al. which consists of 4 main error types: "Slips", "Lapses", "Mistakes" and "Violations". We decided against creating a "violation" type of error, as our task takes place in an explicit human-agent collaborative scenario, where the agent is assumed to be acting in the best interests of the user.

All agents created for this study were derived from a baseline agent named "Gamma" that did not display any specific errors, and achieved an average of 80% Recall on the overall task. The 20% loss in Recall was achieved by adding random biases in its aim, as presented in Section 3.3.2. For the rest of the error-prone agents presented below (agents Delta, Epsilon and Zeta), errors were triggered once every two rounds for a total duration of 30 seconds (half the duration of a level) so that they would be noticeable. To ensure consistency, all agents triggered their errors at the same time, for each participant. We verified that the types of error each agent displayed did not result in changes in reliability by ensuring that they did not significantly affect agents' Recall scores during simulations. During comparisons of Recall scores, an ANOVA yielded p = 0.68 when comparing the performance of all error-prone agents in the Easy difficulty, and p = 0.15 in the Hard difficulty levels, indicating that there was no significant difference between error types in terms of agent reliability. A simplified diagram of the different types of agent behaviour is available in Figure 5.1.

5.3.1.1 Mistakes - Agent Delta

To test the effect of mistakes, we designed agent "Delta". When the behaviour of this agent is triggered, the agent becomes incapable of focusing on one target at a time and instead "bounces" back and forth between available targets while never completely reaching them. This behaviour was created to simulate an explicit error of "planning", most commonly defined as "mistakes" [118, 145].

5.3.1.2 Lapses - Agent Epsilon

To test the effect of lapses; we designed agent "Epsilon". When the behaviour of this agent is triggered, the agent simply becomes unresponsive and stops working, showing no sign of activity at all. This behaviour simulates an error of "omission", commonly linked to "lapses" [118,145].

5.3.1.3 Slips - Agent Zeta

To test the effect of slips; we designed agent "Zeta". When the behaviour of this agent is triggered, this agent becomes extremely inaccurate, barely capable of hitting any target as its aim would always be too far off. This behaviour simulates an error of "commission", commonly linked to "slips" [118, 145].

5.3.2 Task Difficulty

Each participant played with all agents (within-groups study) in sessions lasting for 60 seconds per level. Task difficulty in terms of the number of missiles to hit and their speed was fixed for the "Easy" and "Hard" levels across all sessions, with or without agents. In all of the sessions, the first levels (1 and 2) were set to have an "Easy" difficulty while levels 3 and 4 were set to have a "Hard" difficulty (with a total of 2 minutes per difficulty level). Here are details about the difficulty settings used in this study:

- In the "Easy" level, 2 missiles spawned every 6 seconds at a speed of 60 pixels per second for a total of 30 missiles.
- In the "Hard" level, 2 missiles spawned every 3 seconds with a speed of 120 pixels per second for a total of 60 missiles.

Each error-prone agent triggered their respective erroneous behaviour (slips, lapses or mistakes) during one of the "Easy" and "Hard" levels from T=30s to T=60s. In order to mitigate the learning effect, a William Square design [193] was used to control the order in which participants interacted with the agents.

5.3.3 Independent and dependent variables

In this section, we summarise the independent and dependent variables used in this study and as motivated by our experimental method and research questions.

Our independent variables are the following:

- Task difficulty, as defined by the amount and speed of missiles in each level.
- Aiming agent reliability, which remained high (80% accuracy) for each agent when no error was triggered.
- Aiming agent error types, which were all triggered at the same time and were defined as "slips", "mistakes" and "lapses".

Our dependent variables are the following:

- Task Performance, in terms of missiles hit, shots fired and missiles missed.
- *Reliance*, expressed by the duration for which participants relied on the aiming agent's help.
- *Trust*, as reported by participants.
- Cognitive Workload, as reported by participants.

5.3.4 Experimental Procedure

The study was approved by the University of Strathclyde Computer and Information Sciences Departmental Ethics Committee (Ethics Application No. 1029). The experiment took place in a lab, located in the University of Strathclyde's Computer and Information Sciences Department. We used a lab-based 4 (types of agent errors - no specific error (baseline 80% accuracy), slips, mistakes and lapses) x 2 (difficulty levels - easy or hard) within-subjects design including 2 baseline conditions: (1) no agent (individual performance), and (2) agent with constant performance (upper bound). The general experimental procedure is outlined in Section 3 and in Figure 3.9. At the end of the study, participants received a £10 shopping voucher. After consenting to take part in the study, participants went through the following steps:

- 1. Pre-hoc surveys and demographic questionnaire. (5 minutes) (see Appendix B).
- 2. Tutorial with and without an agent. The agent purposely stopped working so participants could understand that they might need to watch out for errors and take control when required (2 minutes).
- 3. One session without any agent (4 minutes).
- 4. One session with each agent (4 minutes each).
- 5. Post-hoc surveys (5 minutes) (see Appendix B).

After each level, participants were presented with a subset from the Checklist for Trust between People and Automation questionnaire [90] which contained questions pertaining to trust, dependability, reliability, deceptiveness, wariness, confidence and security. In addition, after each session, participants had to complete NASA TLX rating scales [76]. More details about each survey instrument can be found in Section 3.4.2. Prior to the experiment, participants were told that each agent had different behaviours, but no further detail was provided in order to avoid biases. The information sheet can be consulted in Appendix B.

5.3.5 Demographics

24 participants (13 M, 11 F) took part in our lab-based study, with ages ranging from 18 to 40 years old ($M = 26 \pm 5.01$). Most participants (13) had completed a bachelor's degree, while 6 had obtained a Master's degree and the remaining 5 were high school graduate or equivalent. When asked about how often participants played video-games on a 6 points Likert scale from 0 (never) to 6 (Very Frequently), participants scored on average 4 (± 1.82 , indicating that they played "occasionally". The pre and post-hoc survey instruments presented to participants on the Qualtrics platform are available in Appendix B.

5.4 Results

In this section, we present results regarding task performance, users' reliance on agents, reported trust in the agents and cognitive workload. We used the overall scores participants obtained at the end of each session, across all levels of difficulty. More details on the inclusion of different difficulty levels are available in Section 3.2.4. The statistical methods we used to compare and report results is detailed in Section 3.4.4.

Table 5.1: Metrics related to task performance and reliance on the agents. Higher Recall, Precision and F1 scores indicate a better performance while higher User Control Time indicate less reliance on the agent. Highest values are highlighted in bold for each element.

	No Agent	Gamma (baseline)	Delta Mistakes	Epsilon Lapses	Zeta Slips
Recall	0.79 ± 0.02	0.89 ± 0.01	0.85 ± 0.01	0.87 ± 0.01	0.86 ± 0.01
Precision	0.63 ± 0.01	0.60 ± 0.01	0.60 ± 0.01	0.64 ± 0.01	0.61 ± 0.01
F1	0.69 ± 0.02	0.71 ± 0.01	0.70 ± 0.01	0.73 ± 0.01	0.71 ± 0.01
User Control Time	24.88 ± 0.61	10.40 ± 1.04	15.25 ± 0.91	13.15 ± 0.78	13.44 ± 0.85

5.4.1 Performance

Figures 5.2, 5.3, 5.4 and Table 5.1 present summary statistics for each session, where Recall, Precision and F1 scores were used to assess the performance of the participants and agents during each session as described in Section 3.4.3.1. Looking at Figure 5.2 and Table 5.1, we can see that participants performed better while interacting with agents than when completing the task themselves, a fact made especially apparent with Recall scores. In addition, Recall scores are noticeably higher for sessions with an agent not displaying any particular type of error, but these differences are not significant. A Friedman test yielded significant results for Recall (p < 0.0001, W = 0.34) scores, but further comparisons using Wilcoxon signedrank tests did not yield any significant results. When performing a repeated measure ANOVA test on F1 scores, significant results were found (p < 0.0001, F = 3.917, $np^2 = 0.08$). Subsequent pairwise T-test comparisons indicated that F1 scores with agent Gamma (mistakes) were significantly lower than Epsilon (lapses) with p = 0.0479, T = 2.08, CLES = 0.63, while F1 scores with agent Delta (mistakes) were also found to be significantly lower than with agent Epsilon (lapses) with p = 0.044, T = -2.12, CLES = 0.37. Tests on Precision scores are not included as they did not yield important information that tests on Recall or F1 scores do not already provide.



Figure 5.2: Recall scores for each session with or without agents. A higher score indicates better performance. Overall, participants benefited from the addition of an agent, regardless of its error type.



Figure 5.3: Precision scores for each session with or without agents. A higher score indicates better performance. Overall, the presence or absence of an agent, regardless of its error type, did not affect participants' Precision scores in any important way.



Figure 5.4: F1 scores for each session with or without agents. A higher score indicates a better performance. Participants interacting with agent Epsilon (lapses) managed to achieve higher performance than in any other condition.

5.4.2 Reliance

Participant control times were compared, that is, the amount of time participants chose to control the crosshair themselves instead of leaving the agents in charge of aiming, correcting the agents' inputs. The longer the participant control time, the lower the reliance on the agents. From consulting Figure 5.5 and Table 5.1, we notice that participants, unsurprisingly, controlled the crosshair much less when interacting with agents than without, no matter whether the agent displayed errors or not. Furthermore, agent Gamma (no error) was corrected less than any other erroneous agent, with agent Delta (mistake) being corrected more than agent Epsilon (lapses) and agent Zeta (slips). When analysing User Control Times, a repeated-measure ANOVA test yielded significant results ($p < 0.0001, F = 61.63, np^2 = 0.74$). Follow-up pairwise T-tests indicated that participants relied on agent Delta (mistake) significantly less than they did on agent Gamma (no error type) with p < 0.0001, T = 5.52, CLES = 0.71, Epsilon (lapses) with p = , U = , CLES = or Zeta (slips) with p < 0.0001, T = 2.22, CLES = 0.61. In addition, participants relied on agent Zeta (slips) significantly less than they did on agent Gamma (no error type) with p < 0.0001, T = 2.32, CLES = 0.33.



Figure 5.5: User Control Time for each session with or without agents. A higher score indicates lower reliance on the agent. As expected, the agent without any error type (Gamma) induced more reliance on its inputs than any other agent.

Table 5.2: Ratings given to statement related to cognitive load and reported trust in agents. Higher values indicate greater agreement with the statement or higher reported rating for questions. Highest values are highlighted in bold.

Question	No Agent	Gamma (baseline)	Delta Mistakes	Epsilon Lapses	Zeta Slips
I can trust the agent	n/a	64.15 ± 2.35	44.65 ± 2.72	44.19 ± 2.69	46.67 ± 2.47
The agent is deceptive	n/a	44.96 ± 4.58	48.50 ± 5.74	58.96 ± 5.06	47.46 ± 4.17
I am wary of the agent	n/a	53.42 ± 5.03	59.04 ± 5.58	61.17 ± 5.54	60.88 ± 4.90
I am confident in the agent	n/a	52.92 ± 4.67	34.62 ± 4.36	32.96 ± 3.06	37.54 ± 4.02
The agent provides security	n/a	55.71 ± 4.31	40.38 ± 5.30	36.50 ± 4.35	42.21 ± 5.16
The agent is dependable	n/a	61.29 ± 3.13	40.27 ± 3.51	40.35 ± 3.21	44.96 ± 3.43
The agent is reliable	n/a	63.00 ± 3.26	40.33 ± 3.49	44.06 ± 3.46	43.79 ± 3.41
How mentally demanding was the task?	62.90 ± 4.70	54.96 ± 5.20	57.34 ± 6.12	52.98 ± 5.83	51.19 ± 5.02
How physically demanding was the task?	45.83 ± 6.12	41.27 ± 6.36	47.42 ± 6.37	44.05 ± 6.21	41.87 ± 6.31
How hurried or rushed was the task?	74.60 ± 4.81	55.16 ± 5.13	61.90 ± 5.59	56.55 ± 5.47	62.30 ± 5.18
How successful were you in					
accomplishing your level	64.29 ± 4.35	58.93 ± 3.84	71.63 ± 4.15	65.08 ± 4.05	63.10 ± 4.22
of performance?					
How hard did you have to					
work to accomplish your	71.83 ± 3.76	56.55 ± 4.66	61.71 ± 4.77	59.72 ± 4.73	61.90 ± 4.05
level of performance?					
How insecure, discouraged,					
irritated stressed, and	48.81 ± 5.36	50.00 ± 5.02	58.93 ± 6.05	51.79 ± 6.10	55.16 ± 5.28
annoyed were you?					
Overall RAW TLX score	61.38 ± 3.54	52.81 ± 3.72	59.82 ± 4.54	55.03 ± 4.40	55.92 ± 3.97

5.4.3 Trust

At the end of each session or level, participants had to rate the agents based on their perceived trustworthiness, dependability, whether they provided security or if participants perceived the agent as being deceptive. From looking at Figure 5.6 and Table 5.2 and the ratings given to the statement "I can trust the agent", we can clearly see that participants, on average,

perceived agent Gamma (no error type) as more trustworthy than any of the other error-prone agents. In addition, agent Zeta was rated as the most trustworthy of all error-prone agents. When analysing the ratings given to the statement "I can trust the agent", a Friedman test yielded significant results ($\rho < 0.0001, W = 0.30$). Follow-up non parametric pairwise T-tests indicate that participants perceived agent Gamma as significantly more trustworthy than any other agents with $\rho < 0.0001$ yielded during each test. Table 5.3 presents correlations between reported trust ratings and various other behavioural factors linked to reliance, task performance or reported cognitive load. Overall, correlation scores are weak, with Recall scores (measure of performance in terms of target hit) having the highest ρ value with 0.36 points. In close second is Cognitive Workload, with a correlation to Raw TLX scores of -0.32 points.



Figure 5.6: Participant ratings for the statement "I can trust the agent" for each session with agents. A higher score indicates greater trust in the agent. Participants trusted the agent without any error type (Gamma) the most, followed by the agent with the "slips" error type (Zeta).

5.4.4 Cognitive Load

At the end of each session, participants had to fill in a Nasa TLX workload questionnaire. Raw NASA TLX scores are used in this study as they were proven to be as effective as their weighted counter-part [75]. From consulting Figure 5.7 and Table 5.2, we can observe that, on average, interacting with an agent was perceived to be less cognitively taxing than without any agent at all, no matter the behaviour displayed by the agent. When comparing the agents displaying erroneous behaviours, collaborating with agent Epsilon (lapses) seemed to result in a comparatively low cognitive load, while collaborating with agent Gamma (no error) resulted in the lowest cognitive load of all. When comparing Raw TLX scores, a repeated ANOVA test yielded significant results with p = 0.01, F = 3.52, $np^2 = 0.13$. Follow-up pairwise

T-tests found that participants reported a significantly lower cognitive load when interacting with agent Gamma (no error) than with agent Delta (mistakes) with p = 0.0194, T = 2.51, CLES = 0.59.



Figure 5.7: Raw TLX score for each session. A higher score indicates a greater reported cognitive load. Among the error-prone agents, the agent with the "mistakes" error type led to the highest reported cognitive load.

Table 5.3: Spearman's correlation tests between behavioural or reported metrics and trust ratings. A higher ρ scores indicates a greater correlation.

Parameter 1	Parameter 2	ρ	p-value
Recall	Trust ratings	0.3638	< 0.001
RAW TLX	Trust ratings	-0.3247	0.0012
F1	Trust ratings	0.2715	< 0.001
Task Difficulty	Trust ratings	-0.2604	< 0.001
User Control Time	Trust ratings	-0.2515	< 0.001
Precision	Trust ratings	0.1758	< 0.001
Age	Trust ratings	-0.1514	0.0029
Gender	Trust ratings	-0.0361	0.4808

5.4.5 Participants' Feedback

At the end of each session, in accordance with the Critical Incident Technique [62], participants were asked to write about the positive and negative aspects of each agent, and what improvement(s) they would suggest. We then performed qualitative coding to understand how agent errors were perceived by participants. More details about coding analysis are available in Section 3.4.5. After being presented with definitions of "lapses", "mistakes" and "slips", three PhD students in Strathclyde Computer Sciences department were recruited and given the task of coding the "positive", "negative" and "improvement" feedback given by each participant regarding agents. Internal agreement scores (Kappa scores [183]) were used to interpret participants' feedback in relation to agent behaviour. The Kappa score obtained for coding feedback related to "mistakes" was 0.5243 (perceived as "Moderate" [183]), with an internal agreement of 78.73%. The agent most frequently associated with mistakes was, correctly so, Delta, with the most common occurrences being "target(s)" (91 occurrences), "prioritise" (24 occurrences), "focus" (23 occurrences), "confused" (21 occurrences) or "struggle" (14 occurrences). The Kappa score obtained for "lapses" was 0.5493 (perceived as "Moderate"), with an internal agreement of 91.84%. The agent most frequently associated with lapses was, correctly so, Epsilon with the most common occurrences being "sometimes stops" (37 occurrences), "not working" (17 occurrences) or "stopped" (10 occurrences). The Kappa score obtained for "mistakes" was 0.2109 (perceived as "fair"), with an internal agreement of 77.34%. The agent most frequently associated with slips was, correctly so, Zeta, with the most common occurrences being "aim" (23 occurrences), "accurate" (21 occurrences) or "accuracy" (18 occurrences). Overall, these results indicate that participants clearly and correctly recognised the "lapses" and "mistakes" error types while "slips" was harder to differentiate from the rest.

5.5 Discussion

In this study, we examined the effect of different erroneous agent behaviours designed using past work in human-human and human-automation research [118,145]. This study sought to answer our second Research Question presented in Section 1.3: How do different types of agent errors defined from previous related work such as slips, mistakes and lapses affect the human-agent relationship?. We sought to answer the sub-research questions presented in Section 5.1, namely how different types of agent errors influence:

- **RQ2.a:** the users' task performance when interacting with agents?
- **RQ2.b:** the users' reliance on the agent?
- **RQ2.c:** the users' cognitive workload when interacting with the agent?
- **RQ2.d:** the users' reported trust in the agent?

Our analysis of the results showed that, at the same high level of agent reliability, different types of errors (namely: "slips", "lapses" and "mistakes") affected users' reliance on the agents, as well as task performance, cognitive workload and trust in different ways. From looking at the results presented in Section 5.4, it is clear that participants preferred to interact with agent Gamma (no error), as Gamma was deemed more trustworthy than any other agent in terms of reported trust (which informs RQ2.d) while resulting in a lower cognitive workload than any of the error-prone agents. These results were to be expected, as the agent maintained a constant and predictable level of performance throughout the whole study. Error-prone agents, however, influenced the human-agent relationship in more varied ways.

5.5.1 Lapses, Errors of omission

We expected that agent errors that were harder to detect (mistakes) would have a greater impact on task performance and reliance than agent errors were easier to anticipate (lapses). This hypothesis was correct, as lapses led to most changes in reliance and performance compared to the slips and mistakes error types. When looking at the results, it is apparent that participants' interactions with the agent displaying *lapses* (error of omissions) led to the biggest changes in the human-agent relationship. While agent Gamma (no error) led participants to achieve the best task Recall scores (see Figure 5.2), participants interacting with agent Epsilon (lapses) managed to achieve higher Precision (see Figure 5.3) and F1 scores (see Figure 5.4) on average, indicating that they performed with higher accuracy and needed fewer attempt to hit targets when interacting with an agent making errors of omission (which informs RQ2.a). In addition, we also observed that participants corrected agent Epsilon (lapses) less frequently than Gamma (no error) in terms of individual user corrections (see Table 5.1), thus relying more on agent Epsilon (which informs RQ2.b). However, on average, participants corrected agent Epsilon (lapses) for significantly longer periods of time than they did agent Gamma (see Figure 5.5) (which informs RQ2.b). We hypothesise that, in order to match Gamma's performance (no error), participants had to involve themselves more in the aiming process. This increased involvement resulted in fewer corrections (see Table 5.1) than with the baseline agent Gamma (no error), but for longer periods of time (see Figure 5.5), as the errors caused by Epsilon (lapses) were easier to spot and, in turn, fix. As a result, participants actually managed to perform better with Epsilon than with Gamma (no error). However, while doing so, participants reported a significantly higher cognitive workload (see Figure 5.7) due to the higher number of actions they were obliged to carry out. Participants' overall perception of Epsilon (lapses) was also significantly more negative in all of the reported measures (see Section 5.4.5), which informs RQ2.c. Agent Epsilon (lapses) was still perceived as being helpful, as participants reported significantly lower cognitive workload when interacting with it, compared to sessions without any agent. This result could indicate that participants were more tolerant of an agent committing lapses, as this type of error seems to be easier to notice. When looking at participants' feedback for each of the agents, we can also observe that the agent most frequently associated with "lapses" is agent Epsilon (lapses), with many references to its tendency to "stop" or to suddenly "stop working".

5.5.2 Slips and Mistakes, Error of commission and intention

We expected agent errors to have a negative impact on task performance, no matter their type. This hypothesis was proved incorrect for the "slips" type of agent error. Among the other error-prone agents, noticeable differences were found when studying participants' reactions to agent Delta (mistakes) and agent Zeta (slips). In terms of performance, Delta's F1 score (see Figure 5.4), the lowest among all the agents, is significantly lower than Epsilon's F1 score (which address RQ2.a). This decrease in performance goes with a decrease in reliance, as participants corrected agent Delta (mistakes) for significantly longer periods of time (see Figure 5.5) than any of the other agents; error-prone or not (which informs RQ2.b). These findings indicate that an agent making mistakes is harder for participants to correct than an agent having lapses. Interactions with agent Delta resulted in the worst performance and reliance scores, which in turn led to higher cognitive workloads (see Figure 5.7, the highest among all agents and sessions), with a statistically significant difference when compared to the cognitive workload associated with agent Gamma (no error) (which informs RQ2.c).

5.5.3 Perception of the agents

From the qualitative coding (see Section 5.4.5), most of the negative descriptions associated with agent Delta were adequately coded as resulting from "mistakes", with recurring terms such as "confused", "targets" or "prioritise", highlighting how indecisive the aim of the agents was perceived to be. These observations suggest that participants were well aware of how and when Delta was making mistakes, which led them to rely on it less, and to rate it as less dependable and less reliable than any of the other error-prone agents. The scores obtained by participants interacting with agent Zeta (slips) seem to place it in the middle of the other error-prone agents in terms of performance (F1 scores, see Figure 5.4) and participant reliance (user control time, see Figure 5.5). Nonetheless, the number of participant corrections for agent Zeta (slips) was both the highest among all the agents and significantly higher than for Epsilon (lapses). Participant control time was also found to be significantly lower than for Delta (mistakes) and higher than for Gamma (no error).

5.6 Conclusion

In this study, we explored the impact of different types of agent errors and behaviours on users in a real-time collaborative scenario. To do so, we designed different kinds of agent errors based on a taxonomy from the work of Marinaccio et al. [118], namely: "lapses" (errors of omission), "slips" (errors of commission) and "mistakes" (errors of prioritisation). Our findings were aimed at answering our second Research Question: "How do different types of agent errors defined from previous related work such as slips, mistakes and lapses affect the human-agent relationship?"

While all error types adversely affected the way users interacted with agents, each had a different impact on the dependant variables we examined. For instance, "lapses" made participants more alert and even led them to perform better than the baseline agent, while "slips" were perceived as harder to correct, and "mistakes" had the overall worst impact on both perception of the agent and task performance. These results indicate that participants felt the

need to correct the agent making slips (Zeta) more than any other agent, however these corrections were shorter than those for the agent making mistakes (Delta). Overall, the increase in user corrections for Zeta (slips) came with an increase in cognitive workload, as Zeta was found to be the most cognitively taxing agent. Nonetheless, the slips displayed by Zeta were still perceived as less cognitively taxing than the mistakes displayed by Delta. When looking at participants' feedback, agent Zeta was most often coded as displaying "slips", albeit with the lowest Kappa score (0.21) of all the other agent errors. Participants still perceived differences in the way agent Zeta behaved, mainly mentioning issues in the "accuracy" displayed by the agent. In order to make up for Zeta's accuracy, participants had to constantly adjust the aim themselves. However, out of all the error-prone agents, Zeta was rated as the most dependable, as providing the most security and as being the most trustworthy agent.

Our findings suggest that clearly defining the type of error made by an agent can be useful in anticipating the impact it can have on users. As a result, we posit that when designing collaborative agents likely to give imperfect input, it is best to avoid indecisiveness, and that, in most circumstances, a total lack of input is preferable to indecisive or inaccurate information.

Chapter 6

Visual Uncertainty

6.1 Motivation

In this chapter, we focus on the environment of interaction and its impact on human-agent collaboration. More specifically, we investigate whether limiting the amount of visual information required to successfully complete a task can have an effect on the human-agent partnership. As we have seen in Chapters 4 and 5, the behaviour and reliability of an agent largely affect how users are willing to interact with the system. However, the environment of interaction also plays an important role on the effectiveness of the human-agent collaboration, as restricted access to information can directly affect the outcome of a task, which may be disruptive to the human-agent partnership as well.

Previous research on the topic of trustworthy automation has often focused on the design of the agents themselves, by studying the effect of their reliability on users [57] or by adding more information about their actions and reasoning [93, 124], as an attempt to help mitigate adoption issues. In a study about human-agent teamwork, Van et al. [182] suggest that task complexity changes users' attitudes towards agents in terms of trust and fairness of judgement. As we have seen in Section 2.1, many elements influence how users perceive an agent. While most studies have researched how agent features influence the perceived trustworthiness of a system and how to reduce the uncertainty inherent in interacting with an agent [174, 198], fewer works have researched how uncertainty stemming from the *environment of interaction* affects the development of trust, over time.

This study seeks to answer our third Research Question presented in Section 1.3: How do different types of environmental conditions (static or moving), which impair vision and induce uncertainty, affect the human-agent relationship? Specifically, we sought to answer the following sub-research questions: how Visual Environmental Uncertainty that restricts, occludes or hides visual information from the user influences:

- RQ3.a: how well the human-agent team performs on a task.
- **RQ3.b:** how much the user trusts and relies on the agent.

• RQ3.c: situational awareness in relation to the users' trust in the agent.

The work presented in this chapter is an extension of the paper entitled *Investigating the Impact* of Visual Environmental Uncertainty on Human-Agent Teaming [38].

Using the collaborative framework detailed in Chapter 3, we conducted an online-based 4 (types of visual uncertainty) x 2 (levels of difficulty) mixed-design experiment with 96 participants. In this study, four different types of visual environmental uncertainty covering the screen and hiding information (within-subjects) were used as conditions, between two different levels of task complexity (between-subjects). Participants interacted with the same agent throughout the experiment to complete a goal-oriented task in our interactive human-agent framework. Our findings indicate that human-agent collaboration can be greatly influenced by different types of visual uncertainty. For instance, participants trusted the agent more and relied on it more when visual uncertainty was at its highest, despite a loss in task performance. On the contrary, some visual uncertainty that reduced the amount of time participants had to react actually increased task performance.

6.2 Related Work

In HAI, uncertainty is often studied in terms of the transparency of an agent's actions [198], but uncertainty can also arise from the environment in which the human-agent interaction takes place. As previously discussed in Section 2.4.2, *uncertainty* refers to a lack of certainty about a piece of information or a result. As evidenced by a study on trust under visual clutter by Sacha et al. [149], while a significant amount of work has gone into evaluating different techniques to manage visual uncertainty [160], very few studies have focused on how visual impairments, in terms of visual clutter or occlusions, affect users' attitude towards automated systems [101].

To understand and define visual uncertainty, it is useful to look at other disciplines. For instance, past work in Geographical studies has thoroughly defined and researched artefacts that impede users' vision, such as "blurry surfaces" or "dark patterns", as defined by MachEachren et al. [116]. All these patterns, lead, in turn, to the creation of visual occlusions. These occlusions are composed of a shape and a position the combination of which can, in turn, create "occlusions" as defined by VanLier et al. [180]. The resulting occlusions create a "virtual structure" when two or more shapes merge together [180], effectively preventing vision in a particular part of the visual space. Occlusions and the overall masking of visual information can have different effects on people. For instance, an occlusion that restricts peripheral awareness in such a way that attention is only focused on a specific area is called *tunnel vision* [79,115,172]. Tunnel vision, as explained in the work of Ma et al. [115], reduces information input and increases the amount of effort required to gain information, as people affected by tunnel vision have to scan the visual space while being impaired by a narrower field of view. Tunnel vision in a study by Ma et al. [115] or Mackworth et al. [115, 117], or be the result of deteriorating cognitive or physical capabilities [194].

HAI can occur in many different scenarios of various task complexity where information required to perform the task successfully can be unavailable or partially missing. Such settings often consist of search-and-rescue or monitoring tasks, often involving multiple agents such as UAV survey assignments [4, 29]. In these contexts, visual uncertainty can take many forms, whether it stems from a lack of information or the obstruction of visual information. The concept of Situational Awareness (SA), as described in Section 2.6.3, is useful in evaluating how well users can perceive information to inform their decisions. Situational Awareness is used to evaluate how much information users understand, and how well they can use this information to anticipate future outcomes [49] and carry out their task. Ideally, an effective collaborative agent should increase reported SA by increasing users' task knowledge while not increasing their cognitive workload [26]. In HAI, evaluating SA represents a good way of studying how visual uncertainty derived from occlusions can affect users and their attitude towards an agent.

As little work has investigated the impact of visual uncertainty in a human-agent collaborative task, we would like to address this gap by employing an interactive human-agent framework in which different types of visual uncertainty are present, and study the resulting impact on the human-agent relationship. Given the evidence of past research, it is likely that visual impairments will heavily impact on user performance. It is less clear, however, how the same impairments will influence trust and reliance on an agent. In this Chapter, we look at how human-agent collaboration evolves in the context of visual uncertainty with the following working hypotheses:

- H1 According to past work in trust on automation and transparency, visual uncertainty will negatively affect users' trust in the agent.
- H2 The types of visual uncertainty that hide the most amount of information will have the most significant impact on situational awareness and trust.
- H3 Participants will rely and trust the agent more when a type of visual uncertainty forces them to react more quickly.

The main contribution of our work lies in testing the impact of different types of visual uncertainty on the human-agent relationship in a real-time scenario.

6.3 Method

This study was conducted using our interactive human-agent collaborative framework described in Section 3. As opposed to studies presented in Chapters 4 and 5, this study was conducted online. Modalities for online-based experiments are presented in Section 3.5.2.
6.3.1 Visual Uncertainty

We designed four types of visual uncertainty (one visual uncertainty type per condition). Every type of visual uncertainty obstructs visibility and visual information in a different way. Each impairment lasted for the same amount of time (half the duration of one level) and all were triggered at the same time, for each participant. Figure 6.1 presents an abstract representation of the visual uncertainty used in our study while Figure 6.2 showcases the actual implementation as seen in our framework. We named these visual occlusions as follows:

- **Top Disruption:** 50% of the upper part of the screen is hidden (see Figure 6.2c). This type of uncertainty is *static*, and was designed to place participants in a situation where waiting to see the missiles that had recently spawned would reduce the amount of time they had to react.
- Bottom Disruption: 50% of the lower part of the screen is hidden (see Figure 6.2d). This *static* type of uncertainty was designed to force participants into memorising and making predictions on the future position of missiles based on where they spawned and at which speed.
- Moving Clouds: the bottom and top of the screen are alternatively hidden in a continuous manner. Both of the clouds present in Figure 6.2a move across the sky from left to right, always hiding 25% of the top and 25% of the bottom of the screen (i.e. 50% in total). This type of uncertainty is more *dynamic* in nature, and users were able to achieve a higher level of situational awareness if they decided to wait in order to obtain better visibility.
- **Darkness:** the totality of the screen is obscured (see Figure 6.2b), and only a specific area around the cross-hair is made visible to give users a minimum amount of feedback on their actions. This type of *dynamic* uncertainty was designed to constrain the user's focus on one single spot on the screen, creating a "tunnel vision" effect. This condition was designed to test whether participants would completely rely on what the agent was showing them or would try to scan the area manually to increase their situational awareness.



Figure 6.1: Abstract representation of the different types of visual obstruction used in this study. Each obstruction results in occlusions that are either near-complete or partial and static or fixed.



Figure 6.2: All of the different types of visual uncertainties used in this study. (a) "Top Disruption" and (b) "Bottom Disruption" occlude, respectively, the upper and bottom-most parts of the screen while (c) "Clouds" occludes the top and bottom alternatively and (d) "Darkness" obscures the totality of the screen except for the immediate proximity of the cross-hair, producing tunnel-vision.

6.3.2 Task Difficulty and Complexity

Two groups were recruited for this study. The first group experienced a "low" level of task complexity (fewer and slower missiles over both difficulty levels) while the second experienced a "high" level of task complexity (more and faster missiles in both difficulty levels). We chose to include these two levels of complexity as past studies have shown that situational awareness is affected by the number of elements that need to be monitored [49,52]. Each participant played through all four visual uncertainty conditions in sessions lasting for 90 seconds per difficulty level (defined as "easy" and "high" - relative to the task complexity participants were recruited for). Similarly to previous studies, participants were able to adjust the crosshair themselves and were responsible for firing projectiles. Within each group (low or high complexity), we controlled the *difficulty* of the task based on 2 variables: (1) the speed of each missile and (2) the delay between the spawning of each missile. Here are more details about the difficulty settings used in this study for both task complexity groups:

- 1. Low task complexity group:
 - (a) In the "Easy" levels, 1 missile spawned every 6 seconds at a speed of 60 pixels per second (15 missiles in total).
 - (b) In the "Hard" difficulty levels, 1 missile spawned every 3 seconds with a speed of 140 pixels per second (22.5 missiles in total).
- 1. High task complexity group:
 - (a) In the "Easy" levels, 3 missiles spawned every 5 seconds at a speed of either 30 or 50 pixels per second (random selection) for a total of 54 missiles.
 - (b) In the "Hard" difficulty levels, 3 missiles spawned every 4 seconds with a speed of either 60 or 80 pixels per second (random selection) for a total of 67 missiles.

The number of missiles spawning at once as well as their speed was calibrated using pilot studies, in order to make sure that the task was perceived as "easier" or "harder" depending on the level of task complexity and difficulty. In addition, the performance of the agent was calibrated through test simulations where the agent played by itself to ensure that, no matter the level of complexity, it would always get an average Recall score of 0.7 over the "easy" and "hard" levels of difficulty. Each level (either difficulty) lasted for 90 seconds. This duration was set so that participants had enough time to interact with the agent in each condition, while ensuring that the entirety of the experiment could be completed in about 45 minutes, thus reducing participants' fatigue and fostering higher attention levels throughout the online-based study.

6.3.3 Agent Performance

The exact same agent was used in all conditions where an agent was present, no matter the type of uncertainty, level of difficulty or complexity of the task. The agent's level of reliability was set to an average accuracy of 80%, similar to the agent used in Chapter 5.

6.3.4 Independent and dependent variables

In this section, we summarise the independent and dependent variables used in this study and as motivated by our experimental method and research questions.

Our independent variables are the following:

- Task difficulty, as defined by the amount and speed of missiles in each level.
- *Task Complexity*, which was mainly defined by the amount of missiles spawning at once during a level. We divided the recruitment of participants into two groups, one experiencing a lower and the other a higher level of task complexity.
- Aiming agent reliability, which remained high during all conditions.
- Visual uncertainty, which were occluding visual information and were divided into four different entities referred to as "clouds", "darkness", "top disruption" and "bottom disruption".

Our dependent variables are the following:

- Task Performance, in terms of missiles hit, shots fired and missile missed.
- *Reliance*, expressed by the duration for which participants relied on the aiming agent's help.
- *Trust*, as reported by participants.
- Cognitive Workload, as reported by participants.
- *Situational Awareness*, as inferred using participants' attempts to remember how many important elements (in this study, missiles) were present in the top and bottom half of the screen before it was occluded (see Section 3.4.2.3 for more explanations).

6.3.5 Experimental Procedure

This study was approved by the University of Strathclyde Computer and Information Sciences Departmental Ethics Committee (Ethics Application No. 1177). Recruitment was limited to people residing in the UK, where our experimental apparatus was hosted. Limitations on the minimum hardware required to take part in the study was put in place to ensure that participants experienced equivalent experimental conditions. Only data from participants that experienced an average frame-rate above 24 frames per second with a resolution of at least 720p were kept in our dataset for further analysis (more details available in Section 3.5.2). A one minute bench-marking test was provided before users consented to take part in the study. This test was designed in order to filter out participants that did not meet hardware requirements. Final recruitment was conducted online on the Prolific@web based experiment platform, where participants received £5.50 for undertaking the experiment (45 minutes in length). Once registered, each participant went thought the following steps:

- 1. Demographic and pre-hoc survey. (five minutes) (available in Appendix C).
- 2. Tutorial aimed at understanding the controls of the game and how to interact with the agent. (two minutes).
- 3. Session with an agent. (three minutes).
- 4. Session without an agent. (three minutes).
- 5. Four sessions, with each one having a unique type of environmental visual uncertainty (the order of which was randomised using a latin square design [15]) in which cognitive workload, trust and situational awareness surveys were presented. (three minutes each).
- 6. Post-hoc survey collecting participants' feedback. (five minutes) (available in Appendix C).

During the study, participants completed NASA TLX rating scales, which are 6-item survey instruments extensively used to measure cognitive workload [75]. In this study, RAW TLX [18] scores are reported. To measure trust in the agent, we used a single statement at the end of each round: "I can trust the agent" graded on a 7 point Likert scale from 1 (complete distrust in the agent) to 7 (total trust in the agent) adapted from the work of Jian et al. [90]. To measure Situational Awareness, we used the Situation Awareness Global Assessment Technique (SAGAT) [49], which involves freezing the task, hiding all elements present on the screen and asking participants different questions related to the location of items of interest. Due to the nature of the study, we chose to assess "Level 1" situational awareness only, which consists of knowing *where* to find information [53]. In our study, we asked participants how many missiles were present in the top- and bottom-most parts of the screen. More information about the survey instruments detailed here can be found in Section 3.4.2.

6.3.6 Demographics

96 participants (53M, 41F, 2 Non-Binary) were recruited for this study, with ages ranging from 21 to 31 years old ($M = 25 \pm 2.4$). In terms of education levels, most participants (n = 44) held at least a bachelor's degree. When asked how often they played video-games, most participants (n = 29) reported playing "Occasionally".

6.4 Results

In this Section, we present results regarding task performance, users' reliance on agents, reported trust in the agents, cognitive workload and situational awareness. We used the overall scores participants obtained at the end of each session, across all levels of difficulty and across the two level of task complexity. More details on the inclusion of different difficulty levels are available in Section 3.2.4, while more explanations regarding the inclusion of different task complexity levels can be found in Section 6.3.2. The statistical methods we used to compare and report results is detailed in Section 3.4.4.

Table 6.1: Metrics related to performance and reliance recorded during the task. Higher recall, precision and F1 scores indicate a better performance while a higher user control time indicates lower reliance on the agent. Highest values are highlighted in bold.

	No Agent	Agent	Agent Clouds	Agent Darkness	Agent Top Dis.	Agent Bottom Dis.
Recall	0.75 ± 0.02	0.77 ± 0.01	0.74 ± 0.01	0.72 ± 0.02	0.79 ± 0.01	0.72 ± 0.02
Precision	0.65 ± 0.02	0.57 ± 0.01	0.50 ± 0.01	0.56 ± 0.02	0.57 ± 0.02	0.50 ± 0.02
F1	0.69 ± 0.02	0.64 ± 0.01	0.58 ± 0.01	0.62 ± 0.02	0.66 ± 0.01	0.58 ± 0.02
User Ctrl Time	34.47 ± 0.57	24.01 ± 0.87	23.71 ± 0.95	22.51 ± 0.88	25.13 ± 0.95	22.34 ± 0.94

6.4.1 Performance

To calculate performance, we used Recall, Precision and F1 scores which are all computed using metrics related to the success of the task as described in section 3.4.3.1. Tables 6.1 and Figure 6.3 present an overview of the performance scores recorded in the study. As we can see from consulting Table 6.1 and Figure 6.3 participants performed better while interacting with an agent in terms of Recall scores and slightly worse in terms of F1 scores. When visual uncertainty was added, participants' performance dropped slightly, with the exception of the "Top Disruption" condition which actually saw an increase in performance.

A Mixed ANOVA test for Recall scores yielded significant results at all levels: complexity $(p < 0.0001, F = 7.11, np^2 = 0.07)$, visual uncertainty type $(p < 0.0001, F = 9.07, np^2 = 0.09)$ and interaction effect $(p < 0.0001, F = 1.34, np^2 = 0.01)$. Follow-up pairwise T-tests showed that Recall scores during session with the "Top Disruption" were significantly higher than during sessions featuring other types of visual uncertainty: "Bottom Disruption" (p < 0.0001, T = -6.37, CLES = 0.38), "Clouds" (p < 0.0001, T = -6.23, CLES = 0.39) and "Darkness" (p < 0.0001, T = -4.71, CLES = 0.41).

A Mixed ANOVA test for F1 scores yielded significant results at all levels: complexity ($p < 0.0001, F = 0.77, np^2 = 0.15$), visual uncertainty type ($p < 0.0001, F = 20.42, np^2 = 0.78$) and interaction effect ($p < 0.0001, F = 2.66, np^2 = 0.10$). as with Recall scores, Follow-up pairwise T-tests indicate that participants yielded significantly higher Precision scores under "Top Disruption" than under any other type of visual uncertainty: "Bottom Disruption"

(p < 0.0001, T = -6.49, CLES = 0.37), "Clouds" (p < 0.0001, T = -6.65, CLES = 0.39) and "Darkness" (p < 0.0001, T = -2.33, CLES = 0.45).



Figure 6.3: Recall scores for each session with or without agents. A higher score indicates better performance and a higher number of missiles hit. While a higher complexity level led to worst overall performance, participants in the "Top Disruption" condition performed better than in any other condition with visual uncertainty.



Figure 6.4: Precision scores for each session with or without agents. A higher score indicates a better performance and that fewer attempts were required to hit missiles. Similarly to Recall scores, participants in the "Top Disruption" condition performed better than in any other condition with visual uncertainty.



Figure 6.5: F1 scores for each session with or without agents. A higher score indicates a better performance. Overall, the presence of an agent actually led to overall worse performance in terms of F1 scores compared to sessions without any agent.

6.4.2 Reliance

User control time was recorded as the amount of time for which participants corrected agents for (in seconds). A higher user control time indicates lower reliance on the agent. Tables 6.1 and Figure 6.6 present user control times for all sessions. As anticipated, participants controlled the crosshair longer when no agent was present. In general, participants seem to have corrected the agent for much longer periods of time when experiencing any kind of visual uncertainty, with a peak being reached under the "Top Disruption" condition at a high level of task complexity.

A Mixed ANOVA test for user control times scores yielded significant results at all levels: complexity (p < 0.0001, F = 9.78, $np^2 = 0.11$), visual uncertainty type (p < 0.0001, F = 35.24, $np^2 = 0.30$) and interaction effect (p < 0.0001, F = 2.26, $np^2 = 0.03$). Follow-up pairwise T-tests showed that participants in the "Top Disruption" condition relied on the agent significantly less than in the "Darkness" (p < 0.0001, T = -2.73, CLES = 0.42) and "Bottom Disruption" (p < 0.0001, T = -2.38, CLES = 0.43) conditions.



Figure 6.6: User Control Time for each session. A higher score indicates lower reliance on the agent. Among all of the sessions with an agent, participants relied on the agent the least in the "Top Disruption" one.

Table 6.2: Average scores related to reported Trust and Cognitive Workload. Highest values are highlighted in bold.

Question / Statement	No Agent	Agent	Agent Clouds	Agent Darkness	Agent Top Dis.	Agent Bottom Dis.
I can trust the agent	n/a	3.49 ± 0.13	3.83 ± 0.12	4.11 ± 0.13	3.90 ± 0.13	3.64 ± 0.12
How mentally demanding was the task?	14.08 ± 0.46	14.02 ± 0.45	14.46 ± 0.48	14.67 ± 0.49	14.12 ± 0.48	14.48 ± 0.49
How physically demanding was the task?	45.24 ± 2.52	49.12 ± 2.72	54.46 ± 2.75	53.13 ± 2.72	50.15 ± 2.55	54.17 ± 2.82
How hurried or rushed was the task?	70.63 ± 2.06	71.33 ± 1.95	71.73 ± 2.10	69.97 ± 2.14	69.05 ± 2.09	71.13 ± 2.15
How successful were you in accomplishing your level of performance?	39.73 ± 2.57	42.96 ± 2.52	38.79 ± 2.48	35.84 ± 2.42	43.70 ± 2.71	36.51 ± 2.46
How hard did you have to work to accomplish your level of performance?	72.22 ± 1.69	66.52 ± 2.16	68.60 ± 2.14	68.57 ± 2.09	67.41 ± 1.91	69.10 ± 2.00
How insecure, discouraged, irritated, stressed and annoyed were you?	58.33 ± 2.78	61.95 ± 2.71	65.67 ± 2.68	62.41 ± 2.61	59.82 ± 2.67	64.53 ± 2.50
Overall Raw TLX score	57.23 ± 1.17	58.38 ± 1.30	59.85 ± 1.36	57.98 ± 1.27	58.03 ± 1.17	59.09 ± 1.26

6.4.3 Trust

Between each level and after each session, participants were asked to indicate how much they trusted the agent by rating statements graded from 0 (low trust) to 7 (high trust). These statements and associated ratings are presented on Table 6.2. For the study of trust, we are referring to ratings related to the following statement: "I can trust the agent". Table 6.2 and Figure 6.7 present ratings given to the agent for each session. We can see that trust in both low and high levels of task complexity seems to be higher under visual uncertainty than without any type of visual uncertainty. In addition, Table 6.3 presents correlations between trust ratings (under the column "Parameter 2") and other variables recorded in this study (under the column "Parameter 1"). By consulting Table 6.3, we can observe that, overall, correlations between dependant variables and trust ratings are low. The behavioural proxy for reliance (recorded as

"user control time") holds the best negative correlation score with trust ratings with a ρ score of -0.30.

A mixed ANOVA yielded significant results for the following levels: visual uncertainty type $(p = 0.0004, F = 5.14, np^2 = 0.004)$ and interaction effect $(p = 0.0130, F = 3.20, np^2 = 0.05)$ but not for task complexity $(p = 0.53, F = 0.38, np^2 = 0.03)$. Follow-up pairwise T-tests indicated that participants in the "Darkness" condition reported significantly more trust in the agent than in the "Bottom Disruption" condition (p = 0.001, T = 3.37, CLES = 0.58).



Figure 6.7: Reported Trust scores per session, where a higher score indicates greater reported trust in the agent. Participants reported higher trust levels in the "Clouds" condition compared to any other type of visual uncertainty.

6.4.4 Cognitive Workload

After each session, participants were asked to fill in a NASA TLX questionnaire [75] aimed at understanding their cognitive workload. Tables 6.2 and Figure 6.8 present RAW TLX scores, which are aggregated scores used to estimate the overall cognitive load a person reported after completing a task. Looking at Figure 6.8 we can see that participants reported a higher cognitive workload when interacting with an agent under visual uncertainty.

A mixed ANOVA yielded significant results for the following levels: task complexity (p = 0.0096, F = 6.99, $np^2 = 0.07$), interaction effect (p = 0.0091, F = 3.10, $np^2 = 0.02$) but not the visual uncertainty type (p = 0.1807, F = 1.52, $np^2 = 0.03$). No statistically significant differences were found in follow-up pairwise comparisons.



Figure 6.8: Reported Cognitive Load per session. A greater Raw TLX score indicates a more cognitively taxing experience. Overall, the level of task complexity affected reported Raw TLX scores the most.

6.4.5 Situational Awareness

To study situational awareness, we froze the task mid-way through and asked participants to report how many missiles were present in each half of the screen. We then subtracted their answers from the actual number of missiles present on the screen. Thus, a score of 0 means that they guessed the exact number of missiles, while a score of -2 or +2 means that they, respectively, underestimated or overestimated the missile count by 2. The total difference between guesses and the actual number of missiles is presented in Figure 6.9. We can observe that participants were more likely to make accurate guesses when interacting with an agent than when playing without one. However, the type of error participants made was influenced by the type of environmental uncertainty they encountered; for instance, participants operating under the "Darkness" condition recorded the largest range of under or overestimations. Figure 6.9 presents situational awareness results related to the bottom half of the screen only. We can observe that the "Bottom Disruption" and "Top Disruption" scores are very different, with the "Top Disruption" leading to more overestimations than the "Bottom Disruption", which led to more under-estimations.

A mixed ANOVA test yielded significant results on the following levels: task complexity $(p = 0.0117, F = 6.59, np^2 = 0.05)$, visual uncertainty type $(p = 0.0068, F = 3.24, np^2 = 0.01)$ but nothing significant on interaction effect $(p = 0.5571, F = 0.79, np^2 = 0.01)$. Follow-up pairwise T-tests showed that participants underestimated the number of missiles present in the bottom half of the screen significantly more when the bottom of the screen was hidden under the "Bottom Disruption" (p = 0.0106, T = 2.60, CLES = 0.65)

condition than when the top of the screen was hidden under the "Top Disruption" condition (p = 0.0106, T = 2.60, CLES = 0.65).



Figure 6.9: Situational Awareness scores. The closer a score is to 0, the better the participants' situational awareness is. The top-most plot presents overall situational awareness scores while the bottom-most plot presents SA scores related to elements located in the bottom half of the screen. The "Top Disruption" led to more under-estimations while the "Bottom Disruption" led to more over-estimations.

Table 6.3: Spearman's correlation tests between behavioural or reported metrics and trust ratings. A higher ρ scores indicates greater correlation.

Parameter 1	Parameter 2	ρ	p-value
User Control Time	Trust ratings	-0.3058	< 0.001
F1	Trust ratings	-0.1497	< 0.001
Task Difficulty	Trust ratings	-0.1455	< 0.001
Precision	Trust ratings	-0.1355	< 0.001
Gender	Trust ratings	-0.1179	< 0.001
Raw TLX	Trust ratings	-0.1003	0.0284
Age	Trust ratings	-0.096	0.007
Recall	Trust ratings	-0.0783	0.0153
SAGAT Total Difference	Trust ratings	-0.0773	0.0166

6.4.6 Participants' Perceptions

At the end of the experiment, participants were presented with a description of each type of visual uncertainty they encountered and were then asked to report how each condition affected their "ability to play and rely on the agent". Qualitative coding was performed on the resulting dataset by 3 independent coders with no previous involvement in the study. Codes were created to look for particular changes in the way participants perceived their task performance or reliance on the agent. More details on coding analysis are available in Section 3.4.5. The codes used in this study were the following: "Increased Reliance on the agent", "No Change to Reliance", "Decreased Reliance on the agent", "Better Performance", "No Change to Performance" "Lower Performance". Internal agreement scores (Kappa scores [63, 183]), agreement scores and the number of references used for each code per visual uncertainty condition are presented in Table 6.4.

If we consult Table 6.4, we can observe that the "Increased Reliance on the agent" code was the most commonly used, with the highest number of references across all conditions. In addition, its associated kappa scores were always considered "fair to good", as they were found to be consistently above 0.41 [63]. This denotes that people were aware of being more reliant on the agent to varying degrees for each visual uncertainty they faced. The other code heavily employed is "Lower Performance", with a much higher frequency than any other code pertaining to performance (with "fair to good" Kappa scores), indicating that each visual uncertainty heavily impacted participants' perception of how successful they were at completing the task. When comparing participants' feedback for each session, we can observe differences in terms of the codes most used. For instance, the "Clouds" type of visual uncertainty resulted in the highest amount of references used than any other visual uncertainty type, denoting a more varied reaction.

Table 6.4: Results from qualitative coding analysis on post-hoc surveys data asking participants to report how each condition changed their reliance on the agent and ability to play. The highest Kappa scores for each visual uncertainty type are in **bold**.

Visual Uncertainty	Code	References	Agreement score	Kappa score
	Increased Reliance on agent	213	89.34%	0,54
	Lower performance	118	91.15%	0,48
Donlmood	Better performance	31	94.59%	0.18
Darkness	No change to reliance	22	95.89%	0.04
	Decreased reliance on agent	27	96.39%	0.16
	No change to performance	12	95.89%	0.37
	Increased Reliance on agent	153	87.86%	0.38
	Lower performance	102	92.03%	0.51
Clouds	Decreased reliance on agent	62	93.49%	0.29
Clouds	Better performance	37	96.27%	0.24
	No change to reliance	34	94.49%	0.02
	No change to performance	30	96.21%	0.16
	Increased Reliance on agent	108	93.30%	0.41
	Decreased reliance on agent	72	92.31%	0.34
Ton Dismuntion	Lower performance	68	93.73%	0.45
top Disruption	No change to reliance	49	95.19%	0.25
	Better performance	43	95.63%	0.35
	No change to performance	36	96.31%	0.40
	Increased Reliance on agent	147	88.20%	0.39
	Lower performance	121	88.72%	0.44
Bottom Dismuntion	No change to reliance	43	94.38%	0.15
Dottom Distuption	Decreased reliance on agent	39	94.39%	0.07
	Better performance	24	95.99%	0.25
	No change to performance	21	96.84%	0.46

6.5 Discussion

In this study, we explored how *Visual Uncertainty* was impacted users during an interactive human-agent collaborative task. We designed an experiment comprising four different types

of visual uncertainty, each occluding visual information in different ways, and tested their impact on the human-agent relationship. This study sought to answer our third Research Question: How do different types of environmental conditions (static or moving), which impair vision and induce uncertainty, affect the human-agent relationship? from our overall research questions highlighted in Section 1.3. With our sub-research questions, we were concerned about the role of visual uncertainty regarding:

- RQ3.a: how well the human-agent team performs at a task.
- RQ3.b: how much the user trusts and relies on the agent.
- RQ3.c: situational awareness in relation to the users' trust in the agent.

Our results indicate that visual uncertainty, in addition to having a negative impact on task performance, can also alter the way users are willing to rely on and trust an agent in a collaborative task.

6.5.1 Trust under Visual Uncertainty

By referring to prior work related to trust in automation and transparency, we anticipated that uncertainty in the environment would have a negative effect on how participants trusted the agent (see hypotheses presented in Section 6.2). We found evidence to suggest that this is not the case in all situations. When comparing trust ratings (see Figure 6.7), we found that participants reported significantly higher levels of trust in the agent under the "Darkness" type of uncertainty than any other type of uncertainty (which informs RQ3.b). This finding is surprising, as trust is a construct that is calibrated through interactions [43,85], and should benefit from the added transparency found in an environment with perfect visibility. We believe that participants trusted the agent more in such conditions because of the extensive nature of the "Darkness" type of uncertainty, which effectively covered all the whole screen except for a "halo" around the crosshair. In a situation where participants wanted to obtain greater situational awareness, they could have bypassed the agent's aim completely and manually "scanned" the screen for missiles. Instead they decided to rely, to a greater or lesser extent on what the agent was showing them. In the end, lowering reliance on agents was shown to be an ineffective way of dealing with this type of uncertainty, as performance (as demonstrated by recall scores) was found to be significantly lower in "dark" conditions than in any other context.

6.5.2 Situational Awareness under uncertainty

We posited that the types of uncertainty that hid the most information would have the most significant impact on situational awareness and consequently reduce participants' trust the most (see hypotheses presented in Section 6.2). We found that to be the case for one type of uncertainty (see Figure 4.7). In our study, two types of environmental uncertainty were designed to be dynamic and to obstruct the most on-screen elements: "Darkness", where the whole screen was hidden, and "Clouds" where the top and bottoms halves of the screen were hidden dynamically at different points. We found that despite a tendency for participants to make more erroneous guesses under the "Darkness" condition, there were no statistically significant differences related to these "Darkness" and "Clouds" in terms of situational awareness (which informs RQ3.c). However, we found that participants reported significantly higher trust ratings under the "Darkness" condition than when faced with any other type of uncertainty. As the "Darkness" condition hid the most elements from participants, our findings seem to indicate that forcing participants to focus on a specific area could be enough for them to forget what is happening elsewhere, in the same environment of interaction. This finding highlights how easily complacency can set in when environmental visual uncertainty hides crucial information, despite an agent's performance and reliability remaining the same.

6.5.3 Adapting under uncertainty

With our third hypothesis (see Section 6.2), we expected participants to be more likely to rely and trust agents when they were forced to react more quickly. We found that uncertainty types negatively affected reliance, but not reported trust. In the design of our uncertainty conditions, two types of uncertainty were similar in terms of size and shape but different in their locations on the screen: the "Top" and "Bottom" disruptions, which hid, respectively, the top half and the bottom half of the screen. We found that participants relied significantly less on the agent when the top half of the screen was hidden than when the bottom half of the screen was hidden. In this task, missiles spawn from the top of the screen, making that a very important area to attend to, as it allows participants to assess incoming targets and predict their path using their current bearings. It was surprising to see that despite a lack of visual feedback on this crucial part of the screen, participants were actually significantly more likely to take control of the agent's aim. However, when comparing trust ratings, no significant differences were found between the "Top Disruption" and "Bottom Disruption" conditions, which might indicate a level of cognitive dissonance between how users felt about the agent (same level of trust) and how they interacted with it (clear distrust in one set of conditions compared to the other).

6.6 Conclusion

In this Chapter, we examined the impact of different types of visual uncertainty on participants in a human-agent interactive collaborative task. We used both interaction logs and survey instruments to infer and study how and why participants changed their behaviours when confronted with visual environmental uncertainty. We found that the type of uncertainty has a direct impact on how likely participants are to rely on an agent, assuming the agent's level of performance remains the same. More precisely, we found that a high level of uncertainty (studied through the "Darkness" condition in this paper, producing a *tunnel vision* effect) led participants to trust and rely significantly more on the agent compared to uncertainties that were more static in nature, and only obscured parts of the screen (namely, our "Bottom" and "Top Disruption" conditions). In addition, we found that this complacent behaviour led participants to achieve a significantly lower performance, which was ultimately detrimental to the human-agent collaboration. Nonetheless, some types of uncertainty actually mitigated this complacency and kept users more focused on their tasks. We found that hiding the top half of the screen ("Top Disruption" condition), where important information were shown, led participants to perform significantly *better* than, even, in sessions without any type of visual uncertainty. However, overall situational awareness remained fairly constant throughout the study, albeit this could be explained by the inherent difficulty of the tasks, and not the type of uncertainty experienced.

Our findings show that there is a need to understand exactly how users process information in order to plan future actions and make decisions when this same information is missing, as well as how an agent can help mitigate this environmental uncertainty.

Chapter 7

Visual Explanation and Agent Transparency

7.1 Motivation

In this chapter, we look at how visual agents that display information about the task or an aiming agent's actions can affect the human-agent partnership. We intended for the design of each visual agent to enhance specific SA levels and assess their impact on trust, reliance, task performance, cognitive load and situational awareness. This work was motivated by the work of Chen et al. [26], who proposed the "Situation Awareness-based Agent Transparency model" (SAT) which aims to support users' SA via different visualisation levels intended to help users understand a situation or system's decisions, and prepare for the future outcomes of their interactions. The framework defined three Situational Awareness levels. The first focused on understanding "what" is happening, the second one focused on understanding "why" something is happening and the third on understanding what will happen next.

This study seeks to answer the fourth Research Question defined in Section 1.3: How do different types of visual help (designed to elicit different levels of situational awareness) influence the human-agent relationship? More specifically, we aim to answer the following research questions:

- **RQ4.a:** How beneficial is the introduction of visual agents when users are by themselves (no aiming agent)?
- **RQ4.b:** How beneficial is the introduction of visual agents when users are supported by an aiming agent?
- **RQ4.c:** Which visual agent provides the best overall support?

Using the interactive framework detailed in Chapter 3, we conducted an online-based 6 (types of visual explanations) x 2 (levels of difficulty) mixed-design study where 180 participants interacted with an agent in tasks of various levels of difficulty (within-factor) with different

types of visual agents (between-group factor) based on the SA framework by Chen et al. [26]. In addition to SA level, the design of each visual agent was intended to provide information in a descriptive (highlighting important information) or prescriptive ("telling" users what to do) manner. Our results indicate that participants did not significantly benefit from the addition of visual agents in any of the metrics we recorded, with even lower differences in scenarios where an aiming agent was introduced.

7.2 Related Work

7.2.1 Visualisation modalities

In the following paragraphs, we discuss some of the most common paradigms when it comes to communicating information via visualisations in HAI and other domains. Different modalities of providing relevant task-specific information have been designed and employed to provide better decision-making support in a HCI scenario. In a study focused on the design of Head-up Display (HUD), Charissis et al. state that "a successful human-centred interface should enhance human actions [...] senses [...] and judgement [...]. Furthermore, it should guide the user rather than constrain his/her [...] abilities" [23, p. 2]. While Charissis' work was focused on Human-Machine Interfaces (HMI) in an automotive environment (see Figure 7.1), guiding a user without hindering their abilities is an obligation for successful HCI interactions, especially when users are required to understand changes in the environment and respond to them appropriately and in a timely manner. A range of visualisation techniques have been tested in various studies to communicate information as quickly and efficiently as possible [23, 138, 163].

Alphanumeric (alphabetical and numerical) symbols are one of the oldest and most commonly used ways of presenting information to the user. In a work classifying different types of visual representations, Lohse et al. [114] describe how employing numeric elements is often perceived as "unattractive" when used to emphasise parts of a specific representation, and can often lead to confusion by overloading the user [23]. As a more compact way of displaying information, other systems rely on *icons* or *symbols*, which assign an unambiguous meaning to a picture [163]. Icons are used when the meaning of the icon is apparent to the target audience; for instance, signs with an exclamation mark "!" are commonly used to indicate a potentially hazardous area or incoming danger. Icons were found to be interpreted much more quickly by human operators in fast-changing scenarios, where "iconic displays led to response times three times shorter than responses time with alphanumeric displays" [163, p. 5]. In fast-paced tasks, icons can also be used as "attention indices" [138, 139], where symbols serve only as "pointers to attention processes" so that users know what to focus their attention on.

Most visualisation techniques rely on "alphanumeric" symbols and/or "icons" to display changes on an Human-Machine Interface. While these modalities remain the same, their implementation can vary greatly depending on the task. For instance, some interaction scenarios will require the future state of one or more elements to be displayed. In these situations, a technique named "Conformal Symbology" [68] can be useful, as it highlights elements of importance by overlapping them on the environment of interaction itself, providing a more seamless integration of visual elements. In other, longer, surveillance tasks, attention does not have to be sustained at all times but only in short bursts. In these scenarios, "dead reckoning" [163] techniques are often used to provide navigational information by displaying headings (usually in the form of arrows) above elements, hinting at their most likely future positions and letting the user decide whether they deserve further investigation or not.

There are, in short, many different types of visualisation techniques, all of which can be deployed in different ways, and all of which boast their own strengths and weaknesses. For instance, while icons can provide information quickly, they lack the transparency of alphanumeric information. And while alphanumeric symbols are an excellent means of accurately conveying information, they lose in terms of cognitive workload demand and their potential for cognitive capture. To better understand *when and on which criteria* a visualisation should be chosen, one can refer to the literature related to situational awareness [26, 50, 51], where visualisations are assessed based on the type of information displayed and requirements of the task.



Figure 7.1: Example of an Head-Up Display (HUD) interface from a study by Charissis et al. [23]. A good HUD interface must manage to present task-specific information without obstructing the user's field of view.

7.2.2 Visualisation & Situational Awareness

In a comprehensive report by Chen et al. [26], Situational Awareness-based agents are presented as likely to have a positive impact on trust in agents, as they could improve trust development by providing more information about the system's inner workings in a simplified form, as hinted by Lee and See's work [103]. Situational Awareness is usually studied on 3 different levels [26]:

- SA level 1: "The what": The agent conveys information about the current situation.
- SA level 2: "The why": The agent shows its reasoning process and explains its constraints.
- **SA level 3:** "The what next": The agent indicates what could happen next based on current limitations and/or trends.

Previous work in situational awareness and visualisation techniques has either theorised or empirically tested the benefits of numerous visualisation modalities that aim to present information about an agent or the environment of interaction back to a user. However, finding the "best" type of visualisation to display information is challenging, as it largely depends on the type of task and broader context of interaction. In order to investigate the strengths and weakness of different types of visualisation modalities, we make use of the SAT framework formalised by Chen et al. [26] which classifies visualisation modalities based on the type of support they provide.

Based on previous research related to visualisation and situational awareness, the study presented in this chapter investigates the impact of different types of visual agents on the humanagent relationship in a collaborative scenario. Each visual agent employs different visualisation modalities and was both designed according to past studies related to Situational Awareness levels [26] and informed by the empirical implementation of visualisations from past HAI and Human Factor research. In addition to the SAT framework, we further divide our visualisations into two categories: "descriptive" and "prescriptive". Visualisations that are *descriptive* are intended tp focus on highlighting important elements and letting the users make sense of the information while *prescriptive* visualisations are intended to process more of the information for the user.

In this Chapter, we study how human-agent collaboration evolves when supported by different types of visual agents intended to provide different levels of Situational Awareness support. Specifically, we are starting out with the following hypotheses:

• H1 As greater transparency is linked to better task performance [101], visual agents will have a noticeable impact on the human-agent collaboration no matter whether aiming agents are present or not.

• H2 Visual agents that provide prescriptive information (telling the users "what to do") will have a more positive impact on reliance and task performance than agents that provide descriptive information (highlighting important elements in the environment), as they require less time to be processed.

The main contribution of this work lies in testing different types of visual agent (and subsequent visualisations) on the human-agent partnership during a real-time task. Below we outline the method used to design visual agents and conduct this study.

7.3 Method

This study was conducted using our interactive human-agent collaborative framework described in Section 3. Like the study presented in Chapter 6, this experiment was conducted online. Modalities for online studies are described in Section 3.5.2.

7.3.1 Visual Agents

Previous studies presented in this thesis (see Chapters 4, 5 and 6) used agents designed to help users aim at targets. In the work presented in this chapter, we incorporated agents displaying visualisations to users. Here, these agents are referred to as visual agents, as opposed to previous agents (presented in Chapters 4, 5 and 6) that were only helping with the aiming process and will consequently be referred to as Aiming Agents throughout this Chapter. We developed a total of six different types of visual agent based on the situational awareness framework proposed by Chen et al. [26]. Each visual agent is focused on giving the user more information about either what the agent is doing (SA level 1), the agent's reasoning and prioritising process (SA level 2), or what the agent will do next (SA level 3). These 3 levels of Situational Awareness are further detailed in Section 7.2. In addition, we intended for the design of each visual agent to be either "prescriptive" (telling users what to do) or "descriptive" (letting the user make sense of the information). To make all designs comparable, each visualisation only displayed information about 5 targets at once. Below, we provide descriptions of the visual agents developed for this study (please see Figure 7.2 for an abstract representation). Contrast between colours used by visual agents (especially for the "Threat Shape" visual agent) were designed to be high enough to account for users suffering from deuteranopia [71], the most common form of colour blindness.

7.3.1.1 SA Level 1

• Threat Shapes, presented in Figure 7.2 and implemented as shown in Figure 7.3a. This visualisation indicates which target(s) the agent recognises as threats (red triangle), or non-threats (green polygon). This visualisation focuses on SA level 1: understanding "what" is happening in the current situation. The visualisation was based on notes and

implementations found in the work of Pylyshyn et al. [139], Shekkhar et al. [163] and Mercado et al. [124]. This visualisation can be described as "prescriptive", as the visual agent processes most of the information for the user (which targets are important or not).

• **Priority Numbers**, as presented in Figure 7.2 and implemented as shown in Figure 7.3b. This visualisation displays the results of the agent's prioritisation process via numbers, indicating which missiles the user is advised to focus on. This design was inspired by descriptions and implementations found in the work of Lohse et al. [114] and Chen et al. [26], and can be described as "descriptive", as it describes the order in which the agent should aim at targets.

7.3.1.2 SA Level 2

- Agent's Prioritisation, presented in Figure 7.2 and implemented as shown in Figure 7.3c. This visualisation focuses on SA level 2 and understanding "why" certain actions are being recommended to the user or undertaken by the agent. With this visualisation, targets deemed as threats are highlighted with red squares of different sizes and opacity (the bigger and more opaque, the more important the target, according to the agent) to indicate the priority in which participants are recommended to deal with them. This visualisation combines priority and threat assessment to communicate *why* users have to deal with targets in a certain order. The design was mostly informed by the work of Kilgore et al. [93] who empirically tested an interface where the size and transparency of icons were changed to highlight specific elements. This type of visualisation can be described as "prescriptive", as it parses most of the information for the user (which targets are important and in which order they should be hit).
- Missile paths, presented in Figure 7.2 and implemented as shown in Figure 7.3d. This visualisation focuses on SA level 2 and understanding "why" the agent aims at certain targets based on their current paths. This visualisation consists in displaying the paths of missiles and their trajectories. This design was informed by the work of Iordanescu et al. [88]. This visualisation can be described as "descriptive" at it display missiles' trajectories.

7.3.1.3 SA level 3

• Agent's Plan display, presented in Figure 7.2 and implemented as shown in Figure 7.3e. This visualisation displays paths between targets in the order in which the agent is aiming at them. This gives an explanation as to *why* the agent is heading in a particular direction. The design was inspired by the work of Ramchurn et al. [143]. This visualisation can be described as "prescriptive", as it provides the user with a path that they can choose to follow or not.

• **Performance Graph**, presented in Figure 7.2 and implemented as shown in Figure 7.3f. This visualisation support SA level 3 and gives more general information about the current level of performance as well as the current trend (whether the team is getting better with a green arrow or worse - with a red arrow). This visualisation can be described as "descriptive" as it displays information about the evolution of task performance over time.



Figure 7.2: Abstract representation of the visual agents developed for this study. Each visualisation supports a different level of Situational Awareness.





Figure 7.3: All visual agents intended to support different SA levels: (a) represents SA1 Threat Shape, (b) Priority Number, (c) Threat Prioritisation, (d) Agent Plan display, (e) Missiles' paths and (f) Performance Graph.

7.3.2 Task Difficulty

Each participant, no matter whether an aiming agent and/or a visual agent was present, played in sessions composed of two levels of difficulty and lasting for 120 seconds per level of difficulty. Task difficulty was considered in terms of the number of missiles to hit. Their speed was fixed for the "Easy" and "Hard" levels across all sessions, with or without agents. The details regarding difficulty settings are presented below:

- In the "Easy" level, 3 missiles spawned every 5 seconds at a speed of either 30 or 60 pixels per second (random selection) for a total of 54 missiles. 30% (16 missiles) of the missiles spawned were "False Positives" (not heading toward cities).
- In the "Hard" difficulty level, 3 missiles spawned every 4 seconds with a speed of either 60 or 80 pixels per second (random selection) for a total of 67 missiles. 30% (20 missiles) of the missiles spawned were "False Positives" (not heading toward cities).

Contrary to previous studies presented in Chapters 4, 5 and 6, where participants were asked to hit as many missiles as they could, the experiment presented in this chapter relies on testing different visual agents, supporting different SA levels that should help to identify relevant and non-relevant information. In our study, this difference is reflected by the introduction of True Positive (missiles that are going to hit a city) and False Positive (missiles that are NOT going to hit a city). As a result, 30% of all missiles spawned in each difficulty level (Easy and Hard) were False Positives. For this particular study, therefore, our metrics for task performance will be computed differently to take into account the inclusion of False Positives (missiles not colliding with cities highlighted as a threat by the agent) and False Negatives (non-threatening missiles highlighted as threats by the agent). The following equations detail these changes to the original ones present in Section 3.4.3.1.

Threat Precision =
$$\frac{\#ThreateningMissilesDestroyed}{\#TMissilesDestroyed + \#NTMissilesDestroyed}$$
(7.1)

$$\mathbf{Threat Recall} = \frac{\#ThreateningMissilesDestroyed}{\#ThreateningMissilesSpawned}$$
(7.2)

Threat
$$F1 = 2 * \frac{ThreatPrecision * ThreatRecall}{ThreatPrecision + ThreatRecall}$$
 (7.3)

In addition, this study also makes use of "Relative metrics", which are defined as the relative gain or loss in a single session when compared to a reference one. As this experiment contains a between-groups design, computing Relative metrics helps understand whether a visual agent resulted in a relative improvement or loss. For instance, if participants scored a Recall of 0.6 by themselves and 0.8 with a visual agent, the relative gain would be 0.2. This process is illustrated by Equation 7.4 below:

$$Relative Metric = ScoreSessionA - ScoreSessionRef$$
(7.4)

7.3.3 Agent Reliability

Contrary to previous studies (see Chapters 4, 5 and 6), the aiming agent's reliability was not determined by the accuracy of its aim, but by the type of error it made. For each level, as in previous studies, the agent would have a reliability of exactly 80% with the difference from previous studies being that the remaining 20% error-rate comprised only False Negatives (the agent not aiming at a missile that is going to hit a city) and False Positives (the agent aiming at a missile that is going off-screen, outside the viewport) errors.

7.3.4 Independent and dependent variables

In this section, we summarise the independent and dependent variables used in this study and as motivated by our experimental method and research questions.

Our independent variables are the following:

- Task difficulty, as defined by the amount and speed of missiles in each level.
- Aiming agent reliability, which remained high during all conditions.
- Situational Awareness (SA) groups, where each visual agent was intended to provide information regarding SA level 1, 2 or 3.
- *Prescriptive or Descriptive visual agents*, each visual agent, regardless of its SA group, was designed to be either "prescriptive" (intended to guide users toward specific decisions) or "descriptive" (intended to let users make sense out of the information presented).

Our dependent variables are the following:

- Task Performance, in terms of missiles hit, shots fired and missile missed.
- *Reliance*, expressed by the duration for which participants relied on the aiming agent's help.
- *Trust*, as reported by participants.
- Cognitive Workload, as reported by participants.
- Situational Awareness, as reported by participants.

7.3.5 Experimental Procedure

This online study was approved by the University of Strathclyde CIS Departmental Ethics Committee (Ethics App. No. 1395). The study was completed entirely online using participants' own hardware and via the Prolific platform (for more information about the platform, see Section 3). The experimental procedure was similar to the one described in Section 3. Only participants that experienced an average frame-rate above 24 frames per second were kept in our dataset for further analysis (to ensure "playable" experimental conditions). A minute long bench-marking test was provided for free before participants consented to take part in the study in order to filter out those whose hardware did not meet our requirements. Participants received £5.50 for undertaking the experiment (approximate duration of 45 minutes). Once registered, each participant went though the following steps:

- 1. Demographic and pre-hoc survey. (five minutes) (see Appendix D).
- 2. Tutorial aimed at understanding the controls of the game and getting used to interacting with the agent. (two minutes).
- 3. Session with an agent. (four minutes).
- 4. Session without an agent. (four minutes).
- 5. Session with visual agent only. (four minutes).
- 6. Session with an agent and visual agent. (four minutes).
- 7. Post-hoc survey collecting participants' feedback. (five minutes) (for further details, see Appendix D).

The order of the sessions described above (see items 3 to 6) was randomised using a Latin Square [15] to reduce the learning effect. A presentation of each visual agent was provided at the beginning of sessions with examples in order for participants to understand how to make sense of the information communicated to them. In terms of survey instruments, participants completed NASA TLX rating scales, which are 6 item survey instruments widely used to measure cognitive workload [75]. In this study, RAW TLX [18] scores are reported. To measure trust in the agent, we used a single statement at the end of each round: "I can trust the agent" graded on a 7 point Likert scale from 1 (complete distrust in the agent) to 7 (total trust in the agent) adapted from the work of Jian et al. [90]. To measure Situational Awareness, we used the 3 item Situation Awareness Rating Technique (SART) [159] also called "3D SART", which comprises 3 questions eliciting different elements related to SA such as "Attentional Demand", "Attentional Supply" and "Understanding". Further details on the survey instruments used in this study are presented in Section 3.4.2.

7.3.6 Demographics

180 participants (93M, 87F) participated in this study, where most people (n = 104) indicated being aged from 18 to 24 years old while the remaining participants (n = 76) were aged 25 to 34 years old. In terms of level of education, most participants reported having a Bachelor's degree (n = 77) while the rest reported having a College degree (n = 33), High school diploma (n = 32) or Master's degree (n = 20), or other (n = 38). The "Confidence" dimension from the "Revised Computer Game Attitude Scale" [22] was used to evaluate how confident participants felt about their video-game skills with self reported rating scales ranging from 0 (denoting low confidence) to 5 (denoting high confidence). Participants' average rating was 3.85 ± 0.9 , denoting a population used to playing games and confident in its abilities.

7.4 Results

In this section, we present results pertaining to task performance, users' reliance on aiming agent, reported trust in the aiming agent, cognitive workload and situational awareness. We report results obtained by participants at the end of each session, across all levels of difficulty. More details on the inclusion of different difficulty levels are available in Section 3.2.4. The statistical methods we used to compare and report results is detailed in Section 3.4.4. We are using different metrics from those employed in Chapters 4, 5 and 6 to obtain insights on how users performed with and without an aiming agent or visual agent. These changes are explained in Section 7.3.2. When describing results, we distinguish between scores obtained during sessions without aiming agents and sessions with aiming agents. This distinction helps us to focus on the influence of a visual agent on the human-agent collaboration with and without the assistance of an aiming agent.

7.4.1 Relative Performance

Figures 7.4, 7.5 and Tables 7.1, 7.2 and 7.3 present scores related to performance and relative performance. Overall, participants scored higher for all performance metrics in sessions where an aiming agent was present, compared to sessions where participants were only assisted by a visual agent, regardless of its SA level. Relative Threat Recall and Relative Threat Precision scores provide insights into the number of important targets (true positive - missiles heading toward cities) that participants hit during the tasks.

7.4.1.1 Visual Agent without Aiming Agent

Overall, participants' performance was lower when no aiming agent was present. Looking at the Relative Threat Recall scores on Figure 7.4, we can observe that participants benefited from having the support of a visual agent in all groups except "Threat Shape" (SA1) and "Agent Plan" (SA3), which display the most variance in Relative Threat Recall scores. Threat Precision scores, however, exhibit more significant differences when compared to all other performance indicators. Across all six groups, participants scored high Threat Precision scores when no aiming agent was present compared to sessions with an aiming agent. This difference is particularly important for Relative Threat Precision scores in the "Priority Number" group (SA1) and "Missile Path" group (SA2). While performing between-groups comparisons on sessions with a visual agent and without an aiming agent, a Welch ANOVA yielded significant results for Relative Threat Recall scores (F = 11.78, p < 0.0001, $np^2 = 0.14$). Further pairwise comparisons using Games-Howell tests indicated that participants in the "Agent Plan" (SA3) group performed significantly worse than participants in the "Priority Number" (SA1) group (T = 5.56, p = 0.001, CLES = 0.76), "Missile Path" (SA2) group (T = 5.21, p = 0.001, CLES = 0.75), "Threat Prioritisation" (SA2) group (T = 7.16, p = 0.001, CLES = 0.82) and "Performance Graph" (SA3) group (T = -93, p = 0.0021, CLES = 0.31). In addition, participants in the "Performance Graph" (SA3) group scored significantly lower in terms of Relative Threat Recall than participants in the "Threat Prioritisation" (SA2) group (T = 4.3, p = 0.001, CLES = 0.71).

While performing between-groups comparisons on sessions with a visual agent and without an aiming agent, an ANOVA yielded significant results for Relative Threat Precision scores $(F = 8.41, p < 0.0001, np^2 = 0.11)$. Further pairwise comparisons using Tukey tests indicated that participants in the "Priority Number" (SA1) group performed significantly better than participants in the "Threat Shape" (SA1) group (T = 4.65, p = 0.001, CLES =0.73), "Agent Plan" (SA3) group (T = 4.59, p = 0.001, CLES = 0.72) and "Threat Prioritisation" (SA2) group (T = 3.65, p = 0.004, CLES = 0.68). In addition, participants scored significantly higher Relative Threat Precision scores in the "Missile Path" (SA2) group compared to participants in the "Threat Shape" (SA1) group (T = -4.28, p = 0.001,CLES = 0.29) and in the "Missile Path" (SA2) group compared to participants in the "Agent Plan" (SA3) group (T = 4.22, p = 0.001, CLES = 0.71).

7.4.1.2 Visual Agent and Aiming Agent

Overall, participants scored higher Relative Threat Recall scores in sessions where an aiming agent was present (with or without visual agent) compared to sessions with only a visual agent and no aiming agent. Looking at Relative Threat Recall scores (see Figure 7.4) we can observe increases in sessions with an aiming agent *and* a visual agent compared to sessions with only an aiming agent for the "Priority Number" and "Threat Shape" (SA1) visual agents groups. All other groups (SA2 and SA3) present lower Relative Threat Recall scores when a visual agent and an aiming agent are present, compared to sessions with only an aiming agent. For Relative Threat Precision scores (see Figure 7.5), participants in sessions with an aiming agent and visual agent scored higher than in sessions with only an aiming agent for the "Threat Shape" (SA1) group.

While performing between-groups comparisons on sessions with a visual agent and an aiming agent, a Welch ANOVA test on Relative Threat Recall (F = 2.49, p = 0.0334, $np^2 = 0.04$) yielded significant results, but no further significant results were found during pairwise comparisons.

While performing between-groups comparisons on sessions with an aiming agent and a visual agent, an ANOVA test on Relative Threat Precision (F = 7.85, p < 0.0001, $np^2 = 0.10$) yielded significant results while pairwise comparisons using Tukey tests indicate that participants in the "Threat Shape" (SA1) group performed significantly worse in terms of Relative Threat Precision than participants in the "Missile Path" (SA2) group (T = -5.0, p = 0.001, CLES = 0.26), "Threat Prioritisation" (SA2) group (T = -4.57, p = 0.001, CLES = 0.27), "Agent Plan" (SA3) group (T = -4.38, p = 0.001, CLES = 0.28), "Performance Graph" (SA3) group (T = -5.54, p = 0.001, CLES = 0.23) and "Priority Number" (SA1) group (T = 3.93, p = 0.0014, CLES = 0.69).



Figure 7.4: Relative Threat Recall scores for each visualisation, organised by Situational Awareness (SA) levels. A positive Relative Threat Recall score indicate a better performance in terms of Threat missiles hit (True Positive) compared to the baseline session without the help of any agent or visualisations. Overall, the addition of an aiming agent was the main reason for increases in Relative Threat Recall scores.



Figure 7.5: Relative Threat Precision scores for each visualisation, organised by Situational Awareness (SA) levels. A positive Relative Threat Precision score indicates a more efficient ratio of missiles hit (True Positive) to shots fired compared to the baseline session without the help of any agent or visualisations. As opposed to Relative Threat Recall scores, the addition of an aiming agent led to comparatively poorer Relative Threat Precision scores.

7.4.2 Reliance

Figures 7.6 and Tables 7.1, 7.2 and 7.3 present scores related to user control time, which are used as a proxy for measuring reliance on the aiming agent. User control time is measured as the duration for which participants manually controlled the crosshair. When an aiming agent is present, a higher user control time indicates lower reliance on the agent. As user control time is a useful means of understanding reliance on the aiming agent, following description of results will focus on conditions where an aiming agent was present.

7.4.2.1 Visual Agent and Aiming Agent

From consulting Figure 7.6, we can observe that participants controlled the crosshair less when aided by an aiming agent, with or without a visual agent. Nonetheless, differences in reliance were observed when comparing groups in sessions where an aiming agent was present. For instance, participants relied on the aiming agent more in sessions with a visual agent for the "Priority Number" (SA1) and "Agent Plan" (SA2) groups. All other visual agents (from SA1 to SA3), however, induced a lower reliance on the aiming agent.

While performing between-groups comparisons on sessions with an aiming agent and a visual agent, a Kruskal Wallis tests yielded significant results for user control time (H = 21.99, p = 0.0005). Further pairwise comparisons using paired T-TESTS indicate that participants relied on the aiming agent significantly more in the "Agent Plan" (SA3) group than in the "Threat Shape" (SA1) group (U = 2246, p = 0.0004, CLES = 0.69), "Missile Path" (SA2) group (U = 2113, p = 0.0015, CLES = 0.67) and "Threat Prioritisation" (SA2) group (U = 2030, p = 0.0039, CLES = 0.66). In addition, participants relied on the aiming

agent significantly more in the "Performance Graph" (SA3) group than in the "Threat Shape" (SA1) group (U = 2120, p = 0.0026, CLES = 0.66).



Figure 7.6: Amount of time spent controlling the crosshair during each session, organised by Situational Awareness (SA) levels. Positive scores indicate more manual control and lower reliance on the agent. Unsurprisingly, the addition of an aiming agent led to overall lower user control times.

Table 7.1: Average scores for Performance and Reliance metrics yielded by participants in each group related to Situational Awareness level 1. Scores are presented for each session with or without an aiming agent and/or with or without a visual agent (noted as "VA" in the table). Higher scores indicate a better performance or lower reliance on the agent.

	No Visual Agent		Threat Shap	e (SA Level 1)	Priority Number (SA Level 1)	
	User Only	User + AA	User + VA	User + AA + VA	User + VA	User + AA + VA
Recall	0.56 ± 0.02	0.73 ± 0.01	0.55 ± 0.02	0.73 ± 0.01	0.57 ± 0.02	0.77 ± 0.01
Precision	0.64 ± 0.01	0.74 ± 0.01	0.67 ± 0.03	0.77 ± 0.02	0.60 ± 0.01	0.75 ± 0.01
F1	0.60 ± 0.01	0.73 ± 0.01	0.60 ± 0.02	0.74 ± 0.01	0.58 ± 0.02	0.76 ± 0.01
Relative Recall	0.00 ± 0.00	0.17 ± 0.01	-0.01 ± 0.02	0.17 ± 0.02	0.01 ± 0.01	0.21 ± 0.02
Relative Precision	0.00 ± 0.00	0.11 ± 0.01	0.02 ± 0.03	0.12 ± 0.03	-0.03 ± 0.01	0.13 ± 0.01
Relative F1	0.00 ± 0.00	0.13 ± 0.01	-0.02 ± 0.02	0.12 ± 0.02	-0.01 ± 0.01	0.18 ± 0.01
Threat Recall	0.71 ± 0.02	0.85 ± 0.01	0.72 ± 0.03	0.85 ± 0.01	0.72 ± 0.02	0.88 ± 0.01
Threat Precision	0.89 ± 0.01	0.82 ± 0.00	0.92 ± 0.01	0.83 ± 0.01	0.90 ± 0.01	0.80 ± 0.00
Threat F1	0.78 ± 0.01	0.83 ± 0.01	0.79 ± 0.02	0.83 ± 0.01	0.78 ± 0.02	0.84 ± 0.00
Relative Threat Recall	0.00 ± 0.00	0.14 ± 0.02	-0.01 ± 0.03	0.11 ± 0.03	0.04 ± 0.01	0.20 ± 0.02
Relative Threat Precision	0.00 ± 0.00	-0.07 ± 0.01	-0.01 ± 0.01	-0.10 ± 0.01	0.05 ± 0.01	-0.05 ± 0.01
Relative Threat F1	0.00 ± 0.00	0.05 ± 0.01	-0.04 ± 0.02	0.01 ± 0.02	0.05 ± 0.01	0.10 ± 0.02
User Control Time	45.62 ± 1.13	22.42 ± 1.46	43.22 ± 1.61	23.81 ± 1.87	45.42 ± 1.68	19.61 ± 2.13
Relative User Control Time	0.00 ± 0.00	-23.97 ± 1.30	-0.24 ± 1.31	-20.27 ± 1.65	-1.70 ± 1.15	-28.96 ± 1.58
User Correction	n/a	11.12 ± 0.63	n/a	12.87 ± 0.77	n/a	8.72 ± 0.84

Table 7.2: Average scores for Performance and Reliance metrics yielded by participants in each group related to Situational Awareness level 2. Scores are presented for each session with or without an aiming agent and/or with or without a visual agent (noted as "VA" in the table). Higher scores indicate a better performance or lower reliance on the agent.

	No Visual Agent		Threat Prior	ritisation (SA Level 2)	Missile Path (SA Level 2)	
	User Only	$\operatorname{User} + \operatorname{AA}$	User + VA	User + AA + VA	User + VA	User + AA + VA
Recall	0.54 ± 0.02	0.78 ± 0.01	0.62 ± 0.02	0.78 ± 0.01	0.53 ± 0.02	0.73 ± 0.01
Precision	0.60 ± 0.01	0.76 ± 0.01	0.64 ± 0.02	0.78 ± 0.01	0.65 ± 0.02	0.75 ± 0.02
F1	0.56 ± 0.01	0.76 ± 0.01	0.62 ± 0.02	0.78 ± 0.01	0.57 ± 0.02	0.73 ± 0.01
Relative Recall	0.00 ± 0.00	0.23 ± 0.01	0.05 ± 0.01	0.21 ± 0.02	0.00 ± 0.01	0.21 ± 0.02
Relative Precision	0.00 ± 0.00	0.16 ± 0.01	0.05 ± 0.02	0.19 ± 0.02	0.05 ± 0.02	0.15 ± 0.02
Relative F1	0.00 ± 0.00	0.20 ± 0.01	0.05 ± 0.01	0.20 ± 0.02	0.02 ± 0.01	0.18 ± 0.02
Threat Recall	0.65 ± 0.02	0.87 ± 0.01	0.74 ± 0.02	0.88 ± 0.01	0.67 ± 0.03	0.84 ± 0.01
Threat Precision	0.85 ± 0.00	0.79 ± 0.00	0.84 ± 0.01	0.80 ± 0.01	0.90 ± 0.01	0.82 ± 0.01
Threat F1	0.72 ± 0.01	0.83 ± 0.00	0.77 ± 0.02	0.83 ± 0.00	0.75 ± 0.02	0.83 ± 0.01
Relative Threat Recall	0.00 ± 0.00	0.23 ± 0.01	0.07 ± 0.01	0.22 ± 0.02	0.03 ± 0.01	0.21 ± 0.02
Relative Threat Precision	0.00 ± 0.00	-0.05 ± 0.00	0.01 ± 0.01	-0.04 ± 0.01	0.05 ± 0.01	-0.04 ± 0.01
Relative Threat F1	0.00 ± 0.00	0.11 ± 0.01	0.04 ± 0.01	0.11 ± 0.01	0.04 ± 0.01	0.12 ± 0.02
User Control Time	48.57 ± 1.04	19.55 ± 1.22	47.48 ± 1.41	23.78 ± 2.05	42.67 ± 1.62	22.73 ± 1.76
Relative User Control Time	0.00 ± 0.00	-29.55 ± 1.14	-2.81 ± 1.18	-28.04 ± 1.86	-4.17 ± 1.07	-25.20 ± 1.87
User Correction	n/a	10.97 ± 0.58	n/a	11.57 ± 0.94	n/a	12.92 ± 0.86

Table 7.3: Average Scores for Performance and Reliance metrics yielded by participants in each group related to Situational Awareness level 3. Scores are presented for each session with or without an aiming agent and/or with or without a visual agent (noted as "VA" in the table). Higher scores indicate a better performance or lower reliance on the agent.

	No Visual Agent		Agent Path	(SA Level 3)	Performance Graph (SA Level 3)	
	User Only	User + AA	User + VA	User + AA + VA	User + VA	User + AA + VA
Recall	0.58 ± 0.02	0.77 ± 0.01	0.49 ± 0.03	0.74 ± 0.01	0.58 ± 0.02	0.77 ± 0.01
Precision	0.63 ± 0.01	0.70 ± 0.01	0.49 ± 0.02	0.68 ± 0.02	0.69 ± 0.02	0.76 ± 0.01
F1	0.59 ± 0.01	0.73 ± 0.01	0.49 ± 0.02	0.71 ± 0.01	0.62 ± 0.02	0.76 ± 0.01
Relative Recall	0.00 ± 0.00	0.19 ± 0.01	-0.08 ± 0.02	0.17 ± 0.02	-0.00 ± 0.01	0.19 ± 0.02
Relative Precision	0.00 ± 0.00	0.08 ± 0.01	-0.11 ± 0.02	0.08 ± 0.02	0.04 ± 0.02	0.11 ± 0.02
Relative F1	0.00 ± 0.00	0.14 ± 0.01	-0.09 ± 0.02	0.13 ± 0.02	0.01 ± 0.01	0.16 ± 0.02
Threat Recall	0.68 ± 0.02	0.86 ± 0.01	0.59 ± 0.03	0.84 ± 0.01	0.68 ± 0.03	0.87 ± 0.01
Threat Precision	0.83 ± 0.01	0.80 ± 0.00	0.84 ± 0.01	0.80 ± 0.00	0.84 ± 0.01	0.79 ± 0.00
Threat F1	0.73 ± 0.01	0.83 ± 0.00	0.66 ± 0.02	0.81 ± 0.01	0.73 ± 0.02	0.82 ± 0.00
Relative Threat Recall	0.00 ± 0.00	0.19 ± 0.01	-0.09 ± 0.02	0.16 ± 0.02	-0.00 ± 0.01	0.19 ± 0.02
Relative Threat Precision	0.00 ± 0.00	-0.03 ± 0.01	-0.01 ± 0.01	-0.04 ± 0.01	0.02 ± 0.01	-0.03 ± 0.01
Relative Threat F1	0.00 ± 0.00	0.10 ± 0.01	-0.08 ± 0.02	0.08 ± 0.01	0.01 ± 0.01	0.10 ± 0.02
User Control Time	49.83 ± 1.01	16.99 ± 1.31	46.12 ± 1.45	16.67 ± 2.01	47.02 ± 1.49	16.66 ± 1.69
Relative User Control Time	0.00 ± 0.00	-33.34 ± 1.32	-4.23 ± 1.29	-33.97 ± 1.73	-2.28 ± 1.04	-32.76 ± 1.74
User Switch	n/a	8.97 ± 0.55	n/a	8.75 ± 0.79	n/a	9.10 ± 0.82

7.4.3 Relative Reported Trust

Figure 7.7 and Tables 7.4, 7.5 and 7.6 present relative ratings regarding the statement "I can trust the agent" which was presented at the end of every session. The ratings reflect the reported trust in the aiming agent in every session. The relative trust scores presented here were computed by taking the trust ratings from the sessions with an aiming agent and a visual agent and subtracting them from the trust ratings from sessions with the aiming agent and no visual agent. Positive relative trust ratings indicate that the visual agent had a positive impact on the user-aiming agent interaction.

7.4.3.1 Visual Agent and Aiming Agent

If we consult the results, we can observe that participants' trust levels in visual agents changed the most in sessions where an aiming agent was present. In general, the "Priority Number" (SA1), "Threat Prioritisation" (SA2) and "Performance Graph" (SA3) groups did not report widely different levels of trust in the aiming agent when a visual agent was added. Nonetheless, the addition of a visual agent actually reduced Relative Trust levels for the "Threat Shape" (SA1) and "Missile Path" (SA2) groups, and slightly increased Relative Trust levels for the "Agent Plan" group (SA3). In addition to relative trust scores, table 7.7 shows correlations between reported trust and various behavioural and reported metrics such as task performance, reliance and cognitive load. Looking at the results, we can see that, overall, most independent variables have a low correlation with reported trust scores. Surprisingly, Overall 3D SART our measure of general situational awareness in this study - has a slightly higher correlation with trust (ρ of 0.262) than Recall - a measure of performance (ρ of 0.257).

While performing between-groups comparisons on sessions with an aiming agent and a visual agent, a Kruskal Wallis tests yielded significant result for Relative Trust scores (H = 12.31, p = 0.03). Further pairwise comparisons indicate that participants in the "Threat Shape" (SA1) group trusted the aiming agent significantly less than in the "Agent Plan" (SA3) group (U = 1263, p = 0.0039, CLES = 0.35).



Figure 7.7: Relative ratings reported by participants based on the following statement: "I can trust the agent". Scores are organised by Situational Awareness levels. Higher scores indicate a higher reported trust in the agent. Overall, reported trust levels were the most inconsistent for the "Threat Shape" visual agent.

7.4.4 Relative Cognitive Workload

Figure 7.8 and Tables 7.4, 7.5 and 7.6 present relative Raw TLX scores for each session. Overall TLX scores represent the perceived cognitive workload of a session as perceived by participants.

Higher scores indicate higher relative cognitive workload for a given session, when compared to the baseline session which took place without an aiming or visual agent.

7.4.4.1 Visual Agent without Aiming Agent

As evidenced by Figure 7.8, participants reported slightly lower cognitive workload while supported by a visual agent when compared to baseline sessions without a visual or aiming agent in most groups, but this difference is not statistically significant. However, participants in the "Performance Graph" group (SA3) reported a higher relative workload with a visual agent and no aiming agent, but this difference is not statistically significant.

To further investigate differences in Relative Cognitive load between each group, we performed between-groups comparisons for Relative Overall Raw TLX scores. While performing between-groups comparisons on sessions with a visual agent and without an aiming agent, an ANOVA test on Relative Overall Raw TLX scores did not yield any significant results $(F = 0.43, p = 0.82, np^2 = 0.01)$.

7.4.4.2 Visual Agent and Aiming Agent

If we consult Figure 7.8, we can see the addition of an aiming agent, no matter whether a visual agent was present or not, reduces reported relative cognitive load in all groups. When comparing relative Raw TLX scores in sessions where an aiming agent and a visual agent were both present, relative TLX scores decrease, particularly in the "Threat Prioritisation" (SA2) and "Performance Graph" (SA3) groups, but this difference is not statistically significant.

To further investigate differences in Relative Cognitive load between each group, we performed between-groups comparisons for Relative Overall Raw TLX scores. While performing between-groups comparisons on sessions with an aiming agent and a visual agent, an ANOVA test on Relative Overall Raw TLX scores did not yield any significant results (F = 1.66, p = 0.14, $np^2 = 0.04$).



Figure 7.8: Relative scores calculated from the participants' ratings of the the NASA TLX survey to evaluate cognitive load. Relative overall Raw TLX scores are reported here, organised by Situational Awareness (SA) levels. Higher scores indicate a more cognitively taxing experience. Overall, reported Raw TLX scores were lower when an aiming agent was present. The "Threat Shape" (SA1) and "Performance Graph" (SA3) conditions saw the most variance in reported cognitive workload scores.

7.4.5 Relative Situational Awareness

Figure 7.9 and Tables 7.4, 7.5 and 7.6 present Relative 3D SART scores for each session. Positive scores indicate a better reported situational awareness when compared to baseline sessions (without an aiming agent or a visual agent), while negative scores indicate a lower reported situational awareness.

7.4.5.1 Visual Agent without Aiming Agent

Looking at Figure 7.9, we can observe that reported situational awareness varied widely between sessions and SA groups. Without the presence of an aiming agent, the introduction of a visual agent did not affect participants' reported SA in any major way.

To further compare reported overall relative situational awareness scores in each group, we performed between-groups comparisons on Relative Overall 3D SART scores. While performing between-groups comparisons on sessions with a visual agent and without an aiming agent, a Kruskal Wallis test on Overall 3D SART scores did not yield any significant results (H = 6.51, p = 0.26).

7.4.5.2 Visual Agent and Aiming Agent

For most groups, the addition of an aiming agent improved reported situational awareness. This is true for all groups except "Performance Graph" which actually saw a reduction in reported SA following the addition of an aiming agent (without visual agent). For "Priority Number" (SA1), "Threat Prioritisation" (SA2) and "Performance Graph" (SA3) groups, the addition
of a visual agent *and* an aiming agent increased participants' reported SA when compared to sessions with only an aiming agent, but this difference is not statistically significant. For the "Threat Shape" (SA1), "Missile Path" (SA2) and "Agent Plan" (SA3) groups, however, visual agents actually reduced reported SA, most noticeably for the "Threat Shape" group, but this difference is not statistically significant.

To further compare reported overall situational awareness in each group, we performed between-groups comparisons on Relative Overall 3D SART scores. While performing between-groups comparisons on sessions with an aiming agent and a visual agent, a Kruskal Wallis test on Relative Overall 3D SART scores did not yield any significant results (H = 9.64, p = 0.08).



Figure 7.9: Relative ratings collected using the 3D SART survey to evaluate situational awareness. Relative ratings are organised by visualisations intended to support specific Situational Awareness (SA) levels. Higher scores indicate a better sense of situational awareness, as reported by participants. Overall, the addition of an aiming agent and/or a visual agent did not change reported SA scores in any significant way.

Table 7.4: Average ratings or scores reported by participants for every session in the Situational Awareness level 1 groups. Scores are presented for each session with or without an aiming agent and/or with or without a visual agent (noted as "VA" in the table). Depending on the dimensions, a lower or higher score indicates a better or worse experience.

	No Visual Agent		Threat Shape (SA Level 1)		Prioritisation Number (SA Level 1)	
Question / Statement	User Only	User + AA	User + VA	$\mathbf{User} + \mathbf{AA} + \mathbf{VA}$	User + VA	User + AA + VA
I can trust the agent	n/a	4.46 ± 0.14	3.25 ± 0.21	3.88 ± 0.22	3.72 ± 0.23	4.45 ± 0.22
I can trust the agent (relative)	n/a	0.00 ± 0.00	-1.15 ± 0.27	-0.52 ± 0.25	-0.80 ± 0.28	-0.07 ± 0.24
How mentally demanding was the task?	17.05 ± 0.45	14.60 ± 0.49	15.37 ± 0.95	14.23 ± 0.83	17.07 ± 0.57	14.20 ± 0.79
How physically demanding was the task?	11.02 ± 0.80	8.57 ± 0.67	11.40 ± 0.90	10.50 ± 0.89	9.87 ± 1.23	7.33 ± 0.92
How hurried or rushed was the task?	17.14 ± 0.50	15.03 ± 0.47	15.63 ± 0.78	14.87 ± 0.65	17.23 ± 0.64	15.90 ± 0.71
How successful were you in accomplishing your level of performance?	14.14 ± 0.60	12.85 ± 0.56	12.43 ± 0.92	13.13 ± 0.89	14.77 ± 0.82	13.33 ± 0.83
How hard did you have to work to accomplish your level of performance?	15.90 ± 0.49	13.63 ± 0.54	15.00 ± 0.68	13.90 ± 0.76	15.93 ± 0.68	14.70 ± 0.82
How insecure, discouraged, irritated, stressed and annoyed were you?	13.02 ± 0.83	11.87 ± 0.64	14.40 ± 0.88	12.77 ± 0.89	12.67 ± 1.20	10.57 ± 1.15
Overall Raw TLX	56.51 ± 1.55	49.17 ± 1.44	54.66 ± 2.36	51.72 ± 2.03	55.93 ± 2.78	49.07 ± 2.41
Overall Raw TLX (relative)	0.00 ± 0.00	-7.65 ± 1.72	-1.23 ± 2.77	-4.05 ± 2.73	-1.56 ± 1.60	-8.41 ± 1.86
3D SART - Demand	70.57 ± 1.95	66.15 ± 2.01	65.33 ± 2.93	68.62 ± 2.53	72.93 ± 2.37	67.05 ± 2.73
3D SART - Supply	73.73 ± 1.66	74.17 ± 1.78	70.22 ± 2.96	72.88 ± 2.21	79.05 ± 1.84	73.28 ± 2.64
3D SART - Understanding	74.17 ± 1.75	75.47 ± 1.59	72.90 ± 2.40	72.62 ± 2.44	74.57 ± 2.59	78.72 ± 2.00
OVERALL 3D SART	77.33 ± 3.48	83.48 ± 3.23	77.78 ± 5.18	76.88 ± 4.36	80.68 ± 4.19	84.95 ± 4.62
OVERALL 3D SART (relative)	0.00 ± 0.00	6.29 ± 3.39	-1.86 ± 4.50	-3.25 ± 4.27	5.48 ± 4.52	9.75 ± 4.76
I know what the agent is trying to do	n/a	4.83 ± 0.14	4.00 ± 0.22	4.40 ± 0.20	4.28 ± 0.23	5.25 ± 0.20
I know why the agent is doing what it does	n/a	4.62 ± 0.15	3.83 ± 0.22	4.15 ± 0.20	4.30 ± 0.22	4.75 ± 0.21
I know what the agent is going to do next	n/a	3.79 ± 0.15	3.00 ± 0.18	3.25 ± 0.19	3.47 ± 0.22	4.10 ± 0.23

Table 7.5: Average ratings or scores reported by participants for every session in the Situational Awareness level 2 groups. Scores are presented for each session with or without an aiming agent and/or with or without a visual agent (noted as "VA" in the table). Depending on the dimensions, a lower or higher score indicates a better or worse experience.

	No Visual Agent		Threat Prioritisation (SA Level 2)		Missile Path (SA Level 2)	
Question / Statement	User Only	User + AA	User + VA	User + AA + VA	User + VA	User + AA + VA
I can trust the agent	n/a	4.90 ± 0.13	4.27 ± 0.22	4.87 ± 0.20	4.75 ± 0.19	4.67 ± 0.18
I can trust the agent (relative)	n/a	0.00 ± 0.00	-0.67 ± 0.23	-0.07 ± 0.19	-0.12 ± 0.24	-0.20 ± 0.19
How mentally demanding was the task?	17.15 ± 0.42	14.53 ± 0.58	17.13 ± 0.56	14.57 ± 0.88	17.20 ± 0.55	15.83 ± 0.58
How physically demanding was the task?	11.97 ± 0.76	9.47 ± 0.76	11.13 ± 1.07	10.00 ± 1.00	11.63 ± 0.99	11.60 ± 1.03
How hurried or rushed was the task?	17.33 ± 0.46	15.55 ± 0.53	17.00 ± 0.52	14.53 ± 1.01	17.40 ± 0.70	17.07 ± 0.65
How successful were you in accomplishing your level of performance?	13.87 ± 0.66	12.25 ± 0.61	13.80 ± 0.93	11.20 ± 0.79	12.63 ± 0.94	12.63 ± 0.81
How hard did you have to work to accomplish your level of performance?	16.73 ± 0.38	14.33 ± 0.50	16.43 ± 0.61	14.83 ± 0.85	16.57 ± 0.51	14.07 ± 0.61
How insecure, discouraged, irritated, stressed and annoyed were you?	14.35 ± 0.57	12.18 ± 0.71	14.43 ± 0.89	12.23 ± 0.79	14.90 ± 0.96	13.37 ± 0.94
Overall Raw TLX	58.93 ± 1.36	50.62 ± 1.70	57.78 ± 1.86	49.84 ± 2.20	58.04 ± 2.22	54.55 ± 1.99
Overall Raw TLX (relative)	0.00 ± 0.00	-8.31 ± 1.63	-1.51 ± 1.63	-9.44 ± 1.95	-0.53 ± 1.32	-4.02 ± 1.65
3D SART - Demand	74.01 ± 1.76	69.33 ± 1.69	71.98 ± 2.22	65.42 ± 2.98	73.08 ± 2.52	76.25 ± 2.17
3D SART - Supply	76.80 ± 1.69	77.71 ± 1.38	74.67 ± 2.93	74.65 ± 2.62	74.37 ± 2.21	78.12 ± 1.95
3D SART - Understanding	74.48 ± 1.73	77.50 ± 1.42	73.02 ± 2.33	77.48 ± 1.97	74.20 ± 2.34	78.68 ± 2.05
OVERALL 3D SART	77.28 ± 3.28	85.88 ± 2.58	75.70 ± 4.37	86.72 ± 3.47	75.48 ± 4.21	80.55 ± 3.80
OVERALL 3D SART (relative)	0.00 ± 0.00	8.60 ± 3.17	-2.87 ± 4.65	8.15 ± 5.11	-0.50 ± 4.22	4.57 ± 4.17
I know what the agent is trying to do	n/a	5.42 ± 0.12	4.62 ± 0.22	5.48 ± 0.19	5.20 ± 0.20	5.03 ± 0.19
I know why the agent is doing what it does	n/a	5.16 ± 0.14	4.27 ± 0.25	5.17 ± 0.20	5.15 ± 0.19	4.70 ± 0.20
I know what the agent is going to do next	n/a	4.30 ± 0.15	3.75 ± 0.21	4.58 ± 0.21	4.23 ± 0.21	3.73 ± 0.18

Table 7.6: Average ratings reported by participants on questions related to Trust, Cognitive Load (NASA TLX) and Situational Awareness (SART) in Situational Awareness level 3 groups. Higher trust ratings indicate greater trust in the agent, higher Raw TLX scores a more cognitively taxing experience and higher SART scores a better overall situational awareness.

	No Visual Agent		Agent Path (SA Level 3)		Performance Graph (SA Level 3)	
Question / Statement	User Only	User + AA	User + VA	$\mathbf{User} + \mathbf{AA} + \mathbf{VA}$	User + VA	User + AA + VA
I can trust the agent	n/a	4.62 ± 0.15	4.05 ± 0.21	5.07 ± 0.18	4.30 ± 0.24	4.85 ± 0.22
I can trust the agent (relative)	n/a	0.00 ± 0.00	-0.63 ± 0.22	0.38 ± 0.20	-0.27 ± 0.31	0.29 ± 0.23
How mentally demanding was the task?	16.30 ± 0.53	14.17 ± 0.61	17.97 ± 0.49	14.07 ± 0.74	14.87 ± 0.80	12.60 ± 0.96
How physically demanding was the task?	10.38 ± 0.71	8.24 ± 0.77	10.83 ± 1.05	9.00 ± 0.96	8.83 ± 1.24	7.30 ± 0.89
How hurried or rushed was the task?	16.25 ± 0.56	14.32 ± 0.62	17.03 ± 0.65	14.67 ± 0.80	15.63 ± 0.77	13.37 ± 0.92
How successful were you in accomplishing your level of performance?	13.75 ± 0.59	13.00 ± 0.58	13.40 ± 0.90	11.70 ± 0.72	16.20 ± 0.57	12.83 ± 1.00
How hard did you have to work to accomplish your level of performance?	16.53 ± 0.36	13.03 ± 0.60	16.80 ± 0.43	14.27 ± 0.63	15.23 ± 0.80	11.37 ± 0.88
How insecure, discouraged, irritated, stressed and annoyed were you?	12.97 ± 0.77	10.54 ± 0.77	14.67 ± 0.88	13.07 ± 1.15	11.73 ± 1.19	9.33 ± 1.07
Overall Raw TLX	55.46 ± 1.37	46.93 ± 1.75	57.72 ± 1.40	49.76 ± 2.00	53.68 ± 2.02	43.02 ± 2.47
Overall Raw TLX (relative)	0.00 ± 0.00	-8.57 ± 1.67	-0.26 ± 1.38	-8.23 ± 2.12	0.74 ± 1.13	-9.92 ± 1.94
3D SART - Demand	73.35 ± 1.80	66.01 ± 2.00	74.23 ± 2.38	69.85 ± 2.53	64.25 ± 2.92	62.12 ± 3.26
3D SART - Supply	80.31 ± 1.64	75.54 ± 1.91	82.48 ± 1.97	80.67 ± 1.88	73.95 ± 2.92	75.65 ± 2.58
3D SART - Understanding	80.30 ± 1.70	78.63 ± 1.59	71.63 ± 3.04	78.67 ± 2.30	72.32 ± 2.90	78.07 ± 2.23
OVERALL 3D SART	87.26 ± 3.08	88.16 ± 2.89	79.88 ± 5.21	89.48 ± 4.37	82.02 ± 4.40	91.60 ± 4.89
OVERALL 3D SART (relative)	0.00 ± 0.00	0.50 ± 2.92	-10.15 ± 3.85	-0.55 ± 4.14	-2.47 ± 4.40	7.12 ± 3.76
I know what the agent is trying to do	n/a	5.24 ± 0.14	4.35 ± 0.24	5.22 ± 0.18	4.05 ± 0.25	5.40 ± 0.18
I know why the agent is doing what it does	n/a	4.75 ± 0.15	4.32 ± 0.23	4.85 ± 0.20	4.15 ± 0.25	5.23 ± 0.20
I know what the agent is going to do next	n/a	3.87 ± 0.15	3.63 ± 0.21	4.67 ± 0.20	3.67 ± 0.24	4.07 ± 0.24

Table 7.7: Spearman's correlation tests between behavioural or reported metrics and trust ratings. A higher ρ scores indicates a greater correlation.

Parameter 1	Parameter 2	ρ	p-value
Overall 3D SART	I can trust the agent	0.262	< 0.001
Recall	I can trust the agent	0.2569	< 0.001
Raw TLX	I can trust the agent	-0.2292	< 0.001
F1	I can trust the agent	0.2258	< 0.001
Threat Recall	I can trust the agent	0.2221	< 0.001
Threat Precision	I can trust the agent	-0.2204	< 0.001
Relative F1	I can trust the agent	0.2091	< 0.001
Relative Precision	I can trust the agent	0.1994	< 0.001
Relative Recall	I can trust the agent	0.1965	< 0.001
Relative Threat Recall	I can trust the agent	0.1842	< 0.001
Task Difficulty	I can trust the agent	-0.1744	< 0.001
User Control Time	I can trust the agent	-0.1633	< 0.001
Precision	I can trust the agent	0.1626	< 0.001
Relative Threat F1	I can trust the agent	0.1234	< 0.001
Threat F1	I can trust the agent	0.1089	< 0.001
Extended Gamemode	I can trust the agent	0.0897	0.0032
Relative Threat Precision	I can trust the agent	-0.0852	0.0052
Gender	I can trust the agent	0.0257	0.4141
Age	I can trust the agent	0.0216	0.493

7.4.6 Participants' Feedback

At the end of each session, the "Critical Incident" Technique [62] was used, where participants were asked to write about the positive and negative aspects of each type of visual agent they interacted with. The resulting dataset was given to 3 independent coders who had no prior involvement with the study. More details about coding analysis are available in Section 3.4.5.

Coders were presented with the definition of Chen et al.'s levels of Situational Awareness [26] where level 1 is related to understanding *what* the system is doing, level 2 to "why" the system is acting in a specific way and level 3 to *what is going to happen next*. These codes were summarised as "What", "Why" and "Projection" and were used to code the comments left by all participants. Qualitative coding results are presented in Table 7.8.

Looking by Table 7.8, we can observe that the code "What" (SA level 1) was the most widely used for every type of visualisation, with more than 200 references for each visual agent. This finding is interesting, as even visual agents in the SA level 2 and 3 groups were coded as being helpful to understand "what" was happening, which is supposed to be a dimension supported by SA level 1 visual agents. Despite this anomaly, all visual agents also helped participants understand important elements regarding the current state of the task. However, the associated Kappa scores of the "what" code (SA level 1) were by far the lowest among all codes, which denotes a lack of agreement between coders.

Code "Why" had the second highest number of references throughout the dataset, with more than 100 for each visual agent group. In addition, its associated Kappa scores were found to be consistent around 0.36, which denotes a lack of agreement between coders.

Code "Projection" was the least used throughout the dataset, with around 50 references per visual agent group. Its associated Kappa scores, however, were found to be the highest in the visual agent Group "SA1 Priority Number". Interestingly, the visualisations intended to support SA level 3 (future states) were coded less frequently with the "Projection" code than SA level 1 visual agent groups.

Table 7.8: Results from qualitative coding analysis on post-hoc survey data asking participants to report one positive and one negative thing about each visual agent they encountered. 3 independent researchers then assigned relevant comments to individual SA levels (codes) as described in the framework by Chen et al. ("What", "Why", and "Projection").

visual agent Group	Code	References	Agreement score	Kappa score
SA1 Priority Number	What	227	57.48	0.05
	Why	127	84.41	0.36
	Projection	50	87.4	0.67
	What	221	59.53	0.07
SA 1 Threat Shape	Why	133	83.85	0.36
	Projection	52	87.18	0.47
SA 2 Missile Path	What	226	57.80	0.05
	Why	130	84.09	0.36
	Projection	49	87.15	0.04
SA 2 Threat Prioritisation	What	224	59.91	0.05
	Why	132	84.05	0.36
	Projection	44	88.65	0.07
SA 3 Agent Plan	What	224	59.13	0.06
	Why	135	84.16	0.37
	Projection	50	87.42	0.08
	What	219	59.95	0.06
SA 3 Performance Graph	Why	132	84.40	0.37
	Projection	47	87.78	0.05

7.5 Discussion

In this study, we created different visual agents intended to provide better transparency regarding the environment of interaction or an aiming agent's actions during a collaborative Human-Agent task. Each visualisation was intended to support a specific situational awareness level (1,2 or 3, see the work of Chen et al. [26] for more details) and was informed by previous work in Human Factors and HAI studies [124,139,163]. We analysed and compared the influence of each visual agent on task performance, reliance, trust, cognitive workload and situational awareness. This study sought to answer our fourth Research Question: How do different types of visual help (designed to elicit different levels of situational awareness) influence the human-agent relationship? and more specifically the following sub-research questions:

- **RQ4.a:** How beneficial is the introduction of visual agents when users are by themselves (no aiming agent)?
- **RQ4.b:** How beneficial is the introduction of visual agents when users are supported by an aiming agent?
- **RQ4.c:** Which visual agent provides the best overall support?

Our results indicate that different types of visual agent can reduce the uncertainty related to the agent's actions or the task itself, but also overload users and lead to sub-optimal behaviours.

7.5.1 Task Performance and Visual Agent

In this study we introduced new metrics that we named "Threat Recall, Threat Precision and Threat F1" to take into account changes related to the presence of False Positives (hitting missiles that are not going to hit a city) and False Negatives (not hitting missiles that are going to hit cities). In addition, we focused our analysis on "Relative metrics", which represent the relative gain or loss of points for a score in one session when compared against a baseline one (usually, the session without an aiming agent or visual agent).

Overall, changes following the introduction of a visual agent were mostly observed in sessions without an aiming agent, as the introduction of an aiming agent resulted in a clear improvement in most performance metrics throughout all groups, with or without the presence of a visual agent, which goes against our first hypothesis, and informs RQ4.b. However, visual agents resulted in some changes in users' performance and behaviours throughout most groups. We posited that visual agents intended to provide more prescriptive visualisations (instructing the user "what to do") would have a more positive impact on behavioural metrics, such as task performance. We found that the opposite may be true true for some visual agents. Overall, we found the most interesting results when analysing Relative Threat Precision scores (see Figure 7.5), which was to be expected, as Precision scores measure how efficient the user is at a task, and our visual agents were all developed to help participants understand the agent's actions and, in particular, select relevant targets (threat VS non-threats). In terms of Relative Threat Precision scores, participants supported by a visual agent scored the highest in the "Priority Number" (SA1) group. These findings are interesting, as other types of visualisations processing more of the data for the participants (prescriptive visualisations - for instance, the "Threat Shape" (SA1) or "Threat Prioritisation" (SA2) visual agents) did not lead to a significant increase in Relative Threat Precision scores (which informs RQ4.a). It is important to note, however, than we did not check whether prescriptive or descriptive visualisations were perceived as such, which is important to bear in mind while interpreting our results. These findings are, surprisingly, at odds with our hypothesis as visual agents intended to help participants understand "what" was happening likely led participants to spend more time processing information and making decisions (for instance, "Priority Number" at SA1 and "Missile path" at SA2) which, in turn, led to increased performance when compared to other visualisations that presented information that was easier to act on (for instance, "Threat Prioritisation" at SA1 or "Agent Plan" at SA3). Nonetheless, descriptive visualisations led participants to gain a better understanding of targets which resulted in fewer false positive errors (higher Relative Threat Precision scores). Other visualisations that focused on processing more data for participants ("Threat Shape" at SA1 and "Threat Prioritisation" at SA2) gave more information regarding the agent's reasoning which induced better performance in terms of missiles hit (higher Relative Threat Recall scores) but made it harder for participants to distinguish between true and false positives (lower Threat Precision scores).

Overall, it seems that visualisations that encouraged participants to make their own decisions ("Missile Path", "Priority Number") as opposed to letting the agent process most of the data, led participants to obtain a more complete understanding of the task and thus perform better at it (which informs RQ4.c). The type of situational awareness level initially associated with each visual agent, however, did not seem to match the actual impact they had on participants, as the "Priority Number" (SA1) and "Missile Path" (SA2) visual agent groups mostly affected participants' understanding of what target to aim at, which appears to be linked to the SA level 1 from the work of Chen et al. [26].

7.5.2 Reliance, Trust and Visual Agent

We measured reliance as the duration for which participants corrected the agent, while trust was measured with rating scales indicating participants' self-reported propensity to trust an agent.

Overall, participants clearly relied on the aiming agent when its help was available, across all conditions, which could be the sign of a more complacent attitude toward the aiming agent where participants were often more likely to rely on its help than trying to improve on its decision-making capabilities. These results are congruent with past work describing complacent behaviours when interacting with automation [154]. In general, reliance did not change in any important way during sessions without an aiming agent, no matter whether visual agent were present or not. However, in sessions with agents, clearer differences could be noticed. In the "Missile Path" (SA2) group, participants relied less on the agent when they could see the trajectory of the targets, which indicates a better awareness of the situation as these changes led to significant increases in performance in terms of Relative Precision scores. The group "Priority Number" (SA1) witnessed the opposite change, with reliance increasing when priority numbers were displayed on each target, which also resulted in better overall Relative Precision scores. These clear changes in Reliance did not translate to changes in Trust when an aiming agent was added, as trust scores in the "Priority Number" and "Missile Path" groups remained constant across most conditions, except for a noticeable decrease in reported trust for the "Priority Number" group when only a visual agent was supporting participants. Overall, these changes show how different types of visualisations influence users' propensity to rely on the system, with a more descriptive visual agent (in our study, the one displaying the paths of missiles) leading to reduced reliance on the aiming agent. However, this decrease in reliance did not seem to alter participants' reported trust in the aiming agent. This difference in trust and reliance could be due to how trust ratings were collected in the study, with no differentiation between trust in the visual help or the agent when both were present, which could explain our inconclusive results.

7.5.3 Overloading Users

In this study, Cognitive Workload and Situational Awareness were measured via post-task survey instruments such as NASA TLX and the 3 item SART instruments. These metrics help us understand how complex the situation was perceived to be by participants, and how much they understood about it. Compared to metrics collected directly during the task (reliance and performance scores), no significant results were found when analysing reported ratings for any of the visual help groups. Nonetheless, these results still indicate different attitudes towards automation when supported by various types of visual agent.

Overall scores for Cognitive Workload did not undergo significant changes between groups, and noticeable differences were only seen between sessions with and sessions without an aiming agent, with or without visual agents. This was, however, expected as an aiming agent provided the most assistance with the task, as demonstrated in our previous studies (see Chapters 4, 5 and 6). Analysis of reported situational awareness on the other hand, while not yielding any significant results, highlighted more interesting changes. Overall, participants in the "Threat Shape" group reported better situational awareness (higher SART scores) when they were interacting with an aiming agent and without a visual agent, despite not being statistically significant. The same was true for the "Missile Path" group. These findings are surprising, as visualisations were intended to *support* different levels of situational awareness, and not harm them. In the case of the "Threat Shape" group, it is even more surprising as participants reported a lower overall cognitive load and a higher situational awareness in a session that did not include a visual agent despite, once again, not resulting in a statistically significant difference.

By performing qualitative coding, it became clear that most of the participants' descriptions of the visual agents did not match the situational awareness levels they were intended to support as represented in Chen et al. framework [26]. These results could indicate flaws in the conception of the visual agents, despite having informed their design from relevant past work assessing situational awareness. In addition, agreement scores were low for a lot of visual agents, particularly with the code concerning descriptions associated with "what" was happening on the screen. Our results could indicate than Chen's framework may not most suitable to assess SA in non-safety critical scenarios with a general, non-expert audience, as opposed to past SA work that mostly made use of military-oriented scenarios [26].

7.6 Conclusion

For the purpose of this study, we designed six visualisations based on previous work on situational awareness (SA) and tested their effect on human-agent relationships. Each visualisation was intended to be either *descriptive* (letting users make sense of the information) or *prescriptive* (telling users what to do) while supporting different levels of situational awareness (SA level 1,2 or 3) regarding the agent's actions or context of interaction.

When analysing findings, we found that descriptive visualisations (in our study, "Missile Path" and "Priority Number"), led participants to perform significantly better when no agent was present in terms of hitting true positive targets, compared to sessions where no visual or aiming agent was present. Another more prescriptive visualisation (in our study, "Threat Prioritisation") led participants to hit a significantly higher number of targets. Moreover, different types of visualisations led to significant differences in reliance, where a descriptive visual agent (in our study, "Priority Number") induced more reliance on the aiming agent and a prescriptive visual agent (in our study, "Missile Path") led to less reliance on the aiming agent.

In terms of reported trust, cognitive load and situational awareness, differences were observed between visualisation groups. Often, participants in sessions without aiming agents but supported by a visual agent reported having a lower situational awareness than in sessions without an aiming or visual agent. These findings are surprising, as the addition of a visual agent significantly improved performance in some groups ("Missile Path" and "Priority Number" groups for instance) but ultimately resulted in lower SA. However, no significant differences were found when analysing any reported metrics, indicating that participants' perception of the aiming agent or task itself did not dramatically change with the addition of different kinds of visualisation.

Overall, our findings indicate that users react differently to various visualisations intended to give them more information about a real-time human-agent collaborative task. We found that better performance can be achieved by presenting participants with visualisations that clearly describe a situation, without instructing users what to do. However, no clear differences were found when analysing participants' perception of the agents via survey instruments or qualitative coding, which suggest that the way we designed each visualisation did not result in improvements on areas that they were designed to support. Furthermore, these findings indicate that longer interactions might be required for participants to be aware of their changing behaviours towards an agent, and that there is a mismatch between participants' interactions within a task and their reported perception of it.

Part III

Final Discussion and Conclusion

Chapter 8

Discussion

In this chapter, we summarise and discuss the implications of our study findings and present a series of recommendations to support future HAI research and agent designers. As we have seen in Chapter 2, collaborative agents are being used more and more often in environments where users and automated agents have to make quick decisions under various levels of uncertainty [143, 149]. This increased use of agents calls for more research on the elements most likely to influence the development of human-agent collaboration. Some of these features have to do with either the agent itself (its level of reliability, behaviour) or the context of interaction (adversarial conditions, transparency of the agent's actions).

In this thesis, four user studies (presented in Chapters 4, 5, 6 and 7) were conducted using a collaborative human-agent aiming framework detailed in Chapter 3. In our framework, we varied agents' reliability, error patterns and visual elements to understand their effects on users' perception, behaviours and team performance. While the focus of this thesis was on the study of trust in agents, we also looked at behavioural metrics such as reliance and task performance, as well as other reported metrics such as cognitive workload and situational awareness.

8.1 Trust in agents

One of the key focuses of this thesis is the study of reported trust in agents and, in particular, how trust evolves when users interact with agents that display different levels of reliability or behaviours in environments with perfect or sub-optimal access to information. In all studies presented in Chapters 4, 5, 6 and 7, trust was measured via survey instruments composed of either multiple items (Chapters 4 and 5) or a single-item (Chapters 6 and 7). While using multiple-items surveys such as the "Checklist for Trust in Automation" by Jian et al. [90] is useful in understanding specific elements related to trust (such as perceived deceptiveness or reliability), we mostly relied on single-item instruments in order to reduce task interruption and provide a more seamless experience. In particular, we relied on a single-item rating scale derived from the work of Jian et al. [90] (the statement "I can trust the agent") which participants could rate from 1 (low trust) to 7 (high trust).

8.1.1 With regards to Agent Reliability and Error Types

In all studies in this thesis, we ensured that agent reliability (how "good" the agent is at the task) was controlled and maintained at fixed levels in terms of Recall (capacity to hit targets) or False Positive rate (capacity to hit relevant targets - which is only applicable for the study presented in Chapter 7). Unsurprisingly, we found that reported trust in an aiming agent was higher when the agent's reliability was high. This finding was apparent from our first user study (see Chapter 4) where high agent reliability with high agent predictability led participants to trust it significantly more than similarly performing but less predictable agents. These results imply that repetitive automation failures can give a sense of consistency to users, and allow them to better prepare for upcoming errors, which is similar to findings from the work of Fan et al. [58] who experimented with systematic errors in a multi-agent environment. In addition, reported trust scores were found to have a higher correlation with reported cognitive load than any of our performance metrics, indicating that the mental load of a task can be a better predictor of trust than even task performance, which tend to support previous work who found a high inverse correlation between reported trust and cognitive workload [2].

For our follow-up studies (see Chapter 5 onward), we chose to stop integrating "low reliability" agents. These, we reasoned, do not represent realistic use-cases for studying human-agent interaction as they are likely to be underused by users anyway (see results in Chapter 4). In our second user study (see Chapter 5), we tested the impact of different types of agent errors or "behaviours" and the way in which they changed users' willingness to trust an agent and rely on its help. We found that agents making aiming errors (defined as errors of "slips" - errors of commission) were perceived as more trustworthy than agents "forgetting" to aim at targets (defined as "lapses" - errors of omission) or agents that aimed at the wrong targets (defined as "mistakes" - errors of intention). Of course, all-error prone agents were perceived as being less trustworthy than the baseline 80% accuracy agent. These findings imply that users find an agent more trustworthy when the agent takes the initiative and clearly shows that it knows what to do (i.e. which target to aim at) even if it is ultimately unsuccessful at the task. Our findings seem to indicate that ascribing intent to an automated agent is something that users tend to do naturally when engaging with automation, even in a goal-oriented task where agents display no anthropomorphic indication of their motivations. Furthermore, our results add to a growing body of studies attempting to explain and/or leverage anthropomorphic features [95], often to favour a continuously calibrated level of trust in the agent, where users' expectations match the actual capabilities of the agent, as described in the work of Merritt et al. [125].

8.1.2 With regards to visual uncertainty and agent transparency

In the studies presented in Chapter 6 and Chapter 7, we tested different types of adversarial visual conditions on users. In both studies, aiming agents were set to have a 80% accuracy.

We designed adversarial conditions to induce uncertainty in the task by preventing users from clearly seeing targets. This was achieved by including either dynamic or static occlusions that were partial (part of the screen) or near-total (nearly the entire screen). We found that users trusted an agent the most when the occlusion was total, which was to be expected as the agent continued to function no matter the kind of visual uncertainty present. A more surprising finding, however, was that participants preferred to "blindly" trust the agent's recommendation under the highest levels of uncertainty, even though it led to poorer task performance. This finding was particularly interesting as users could, at any moment, correct the agent and get a better understanding of the situation before making a decision. Our results underline the need to promote visualisations that enhance agent transparency by informing about "mistaken uncertainties" (when an agent is misunderstood) and "unaware uncertainties" (when users are missing an important information), as presented in the work on uncertainty and trust in visual analytics by Sacha et al. [149].

In the study presented in Chapter 7, we added different types of "visual agent" which were intended to explain the aiming agent's actions through visualisations highlighting either the task (which elements are important) or the aiming agent's actions (why the agent is doing something). Each visualisations was inspired from past SA-related work and based on the framework by Chen et al. [26]. As this study used a between-groups design where every participant experienced one of six potential types of visual agent, we focused on "relative metrics" which describe the relative gain (or loss) that visual agents were responsible for during the human-agent interaction. Overall, we found that reported trust in the agent did not change when participants were supported by a visual agent. Nonetheless, we found that displaying the "path" of an agent (i.e. the agent's future plan of action) improved users' trust the most, while displaying which targets the agents thought were relevant or not actually reduced trust the most, however these results were not statistically significant. These findings are surprising, as more information and transparency about an agent's reasoning process should improve reported trust in it, or at least help users to better calibrate it, which contrasts with past work on automation transparency and uncertainty communication in HAI [101]. These results could be explained by the type of transparency provided to users, as participants tended to trust and rely on visualisations that presented information intended to be more transparent ("descriptive" visualisations) more than visualisations that were intended to simply tell users what to do ("prescriptive" visualisations). However, during analysis of qualitative feedback from participants, we found that most visualisations were not perceived as supporting their intended SA level as defined by the framework of Chen et al. [26], which could also indicate a flaw in the design of our visual agents, despite being based on previous HAI work.

8.2 Reliance on agents

Reliance, in addition to trust, is an important metric in understanding how users perceive automated agents. Where reported trust measures users' subjective perception of an agent, reliance, studied via a behavioural proxy, represents users' actual interaction with an agent [103]. In all our studies presented in this thesis, we studied reliance by recording the amount of time (in seconds) for which participants assumed responsibility for the controls when interacting with an aiming agent. This measure, that we call "User Control Time", is a direct behavioural proxy that gives an estimate of how much users actually relied on the aiming agents, where a higher user control time represents lower reliance on the agent, and vice-versa.

In our first study investigating the effect of agent reliability and predictability on users (see Chapter 4), we found that reliance was positively affected by more reliable *and* predictable agents. We observed that participants relied significantly more on agents with high levels of reliability and high levels of predictability than agents with high levels of reliability but low levels of predictability. These findings showed that greater reliance on an agent can indeed be correlated with higher trust, as seen in Section 8.1. It also means that when users can more easily predict the actions of an agent, they tend to correct it more efficiently as well as being more willing to rely on its future input, as they know how to anticipate its potential failures. These results are largely coherent with past work that linked a higher ability to predict an agent's actions with a better calibration of users' reliance [130].

In our follow-up study (see Chapter 5), we designed different types of agent behaviours and tested their impact on users' reliance. We found that errors types, much like levels of predictability, did have an important influence on reliance. Through our results, we learned that participants relied significantly less on agents that were committing errors of judgement (defined as "mistakes" in our study) and commission (defined as "slips") than agents committing errors of omission (defined as "lapses"). These results indicate that, as with trust, users ascribe intents to agents, and notice patterns in the way they make errors. In return, these differences influence the way participants rely on agents. For instance, we found that users were more likely to rely on agents that chose not to take any action ("lapses") than those that took a decision and made an error ("slips" or "mistakes"). This finding represents an interesting avenue to pursue if we are to understand reliance on faulty automation, since errors of omission seem to have a less destructive impact on users' perception of agents, and might be more easily salvageable than other kind of agent errors. Our findings seem to be in line with past work, such as a study by Sanchez et al. [151] which found that a high amount of "false alarms" (which is similar to the "slips" and "mistakes" error types in our experiment) is linked with decreases in reliance compared to other error types.

In the study presented in Chapter 6, we experimented with different types of "visual uncertainty" that occluded information from users to see how reliance on agents would be affected. We observed that users tended to rely less on agents when crucial parts of the environment of interaction were occluded (in our study, the top of the screen, where targets appear from). This finding indicates that the introduction of an agent does not necessarily makes users more complacent, and that some kind of adversarial visual condition could even lead to improvements, as users performed better while relying less on the agent (in the session where the top of screen was hidden, for example). Conversely, other conditions where most elements were occluded did lead to changes in reliance albeit to the detriment of task performance, which underlines that *calibrated reliance* is just as important as *calibrated trust* in an agent is important for effective human-agent collaboration. These results add to past work that found reliance to be better calibrated in situations with visual uncertainty compared to situations with more readily available visual information [101].

In the study presented in Chapter 7 (and as opposed to the study presented in Chapter 6) we evaluated different types of visualisation intended to provide more transparency regarding the agent's actions or detailing important elements in the environment of interaction. Interestingly, while trust and performance were affected by the type of visualisation used, reliance evolved differently depending on the kind of visual agent present. For instance, participants that interacted with descriptive visual agents (intended to show more information about the task) tended to rely more on the aiming agent while the use of prescriptive visual agents (intended to tell participants "what to do") resulted in lower reliance on the agent. These findings could further highlight the situational and contextual nature of reliance, as reliance is heavily influenced by both the amount of information available during the task (used to make informed decisions) and the knowledge gained about the agent and its usefulness in a given situation. While further work is required to discuss the merits of specific visualisation types regarding reliance on an agent, our study contrast findings from previous work that found visualisations to significantly change (either positively or negatively) the way users rely on automation [101].

8.3 Task Performance

All of the lab-based and online user studies presented in this thesis were conducted using the framework described in Chapter 3. Despite differences in research questions and study goals, all studies were comprised of the same goal-oriented scenarios where participants had to protect cities from missiles that appeared at the top of the screen. As the task was goal oriented, we designed a set of metrics to evaluate participants' success in terms of performance. More precisely, we used the amount of shots fired, missiles hit and shots missed to compute Recall, Precision and F1 scores (more details in Section 3.4.3.1).

In our first experiment (see Chapter 4), we studied how agent reliability and predictability affect performance. Overall performance (in terms of missiles hit - Recall and ratio of shots fired to missile hit - Precision) was significantly better with more predictable agents, operating at a high level of reliability. Our findings indicate that an agent perceived as being more trustworthy and reliable was also linked to higher task performance. While it is hard to say precisely what influenced participants' perception of agents the most, higher task performance may have been a result of agents being easier to correct thanks to more predictable error patterns. When comparing correlations with performance metrics, Recall scores yielded a higher positive correlation with trust ratings than Precision score. Overall, these findings indicate that participants were perhaps more sensitive to the number of missiles hit (expressed with Recall scores) than the ratio of missiles hit to shots fired (expressed with Precision scores). These results are in line with past work that noted a strong correlation between trust and performance [25]. Nonetheless, in our experiment, we found that reported cognitive workload was more strongly correlated with trust ratings than any performance metrics.

In our second experiment (see Chapter 5), we compared the impact of different agent error types on participants. Of all the error-prone agents, we found that errors of omission (defined as "lapses" - where the agent simply does not do anything) actually had a *positive* impact on participants' performance in terms of Precision and F1 scores when compared against all other sessions, with error-prone agents or not. These results are surprising, as we expected errors to be a hindrance to participants, no matter their type or the way we designed them. Instead, it seems that errors of omission can lead users to be more focused on a task, as a closer monitoring of the agent's inaction is required. As a result, these lapses might have been more readily spotted and participants managed to react in a more timely manner which led to increased performance, especially in terms of accuracy with higher precision and F1 scores. Our findings add to a growing body of work on automation errors (for instance, an unusual request or behaviour) has been shown to make users more careful with their future agent interactions [150].

In our third experiment, different types of visual uncertainties were designed and tested on human-agent collaboration (see Chapter 6). Unsurprisingly, we found that the higher the uncertainty (where most of the screen is occluded), the worse participants' performance. In general, hiding the top of the screen (which reduced participants' time to detect, aim at and hit targets) significantly *improved* performance. These results might indicate that, as we have seen in the context of users interacting with agents prone to errors of omission, better task performance can be fostered by reducing the amount of time users have to react and make decisions, which can in turn help maintain good attention levels and, as a result, lead to higher task performance. Our findings contrast with past work that has shown automation uncertainty or sensor-related uncertainty to always negatively affect task performance [78, 149].

In our fourth experiment (see Chapter 7) we designed and experimented with different visual agents displaying task-related information. We found that visualisations intended to be more

descriptive (highlighting important elements) led to significant improvements in terms of relative performance (when compared to baseline sessions, without visualisation). These findings indicate that participants are likely to perform better while supported by visualisations intended to focus on increasing task transparency than while supported by visualisations intended to "tell them what to do". As with visual uncertainty, this kind of information transparency could be leveraged to foster increased levels of attention from users, while allowing them to perform better thanks to an added understanding of which elements require monitoring. These results add to an ever increasing body of work on visual analytics, where different solutions are proposed to support HAC in high-workload scenario where maintaining a high level of task performance is paramount [101].

8.4 Cognitive Workload

In all studies conducted and presented in this thesis, we used the NASA TLX rating scales [77] to evaluate reported cognitive workload. This survey instrument is the most widely used [75] method to evaluate cognitive workload thanks to its non context-specific nature and ease of administration. The higher the score reported, the more cognitively taxing the task is perceived to be. In all our studies, we focused on reporting *Raw TLX* which consists of an aggregate score (from 0 to 100) denoting the overall cognitive cost of a task.

In our first and second user studies (see Chapter 4 and 5), we tested different types of agent behaviour in terms of reliability, predictability (Chapter 4) and error types (Chapter 5). As expected, we found that higher agent reliability led to a significantly lower cognitive workload, and that the same was true for agent predictability, due to the ease with which users could predict errors and make adjustments accordingly. In addition, we found that reported cognitive workload tended to be higher when an agent makes a wrong decision (defined as "slips") than when it made no decision at all (defined as "lapses"). These results indicate that, assuming a high level of agent reliability, being able to better predict the actions of an agent will likely lead to a lower reported cognitive workload. Interestingly, in one of our studies (presented in Chapter 4), cognitive workload had the highest correlation with reported trust ratings, even higher than the task performance metrics (Recall, Precision F1). These findings highlight how important task complexity and the perceived complexity of interacting with an agent are regarding the development of cognitive workload, which is in line with past work that found a lack of agent transparency to have a negative impact on the development of cognitive workload, regardless even of task complexity [26].

In the third and fourth user studies presented in, respectively, Chapter 6 and 7, we tested different types of visual uncertainties on users and evaluated the benefits of visualisation displaying information about either the agent's actions or environment of interaction. More precisely, in our third user study, presented in Chapter 6, we tested different modalities of visual occlusions (dynamic, static, partial or total). Overall, we did not find any significant differences between them in terms of reported cognitive workload. These findings could be explained by the inherent complexity of the task, which led to high levels of reported cognitive workload even in sessions without any type of visual uncertainty. Similarly, in our fourth user study (see Chapter 7), we found that our different types of visualisation did not influence participants' reported cognitive workload. These findings were consistent throughout the study despite changes in either the intended type of visual agent (descriptive or prescriptive) or nature of visualisation used (see summary of the visual agents used in Section 7.3.1). Overall, and as with our study focusing on visual uncertainty (see Chapter 6), one could argue that participants' cognitive workload was most affected by task difficulty, and that visualisations did not significantly affect reported cognitive workload, despite other improvements in terms of reliance on the agent, trust or task performance.

8.5 Situational Awareness

In our studies, Situational Awareness (SA) represents the amount of information that a user can assimilate about what a system is doing (SA level 1), why it is acting in a specific way (SA level 2), and what it will likely do next (SA level 3) [26]. SA is an important concept in HCI studies as it allows for the measurement of users' understanding about a situation or an agent, and can be used to help understand the development of other related constructs such as task performance, reliance or reported trust. Situational Awareness was only assessed in two studies (see Chapter 4 and Chapter 5) and via two survey instruments: SAGAT and SART [53], which limits comparability between studies.

In our third user study (see Chapter 6), we used SAGAT to evaluate SA level 1 ("what" the user understands about a situation) by asking participants to report the number of missiles present in either the top or bottom half of the screen, and compare this to the actual number of missiles present. We found that there was no major discrepancy between conditions in terms of over- or underestimations, and that obscuring the bottom and top half of the screen led, respectively, to more under- and over estimations. Overall, these findings were not particularly insightful as regards to their relationship with other metrics (no clear correlations between SA and other behavioural or reported metrics were found). However, our findings served as a reminder that task complexity coupled with reduced access to important information will likely reduce situational awareness, no matter the kind of obstacle faced. Our findings seem to support past work that found limited access to information and high task complexity to negatively affect SA the most [135].

In our fourth study (see Chapter 6), we designed and experimented with different visual agents whose intended goal was to display more information about either the task or the aiming agent's actions. No matter the visualisations tested, we expected improvement in SA no matter

what visualisations we tested, as task transparency is key to higher levels of reported SA [72]. We did not find any significant differences in terms of SART scores between sessions. Our results might indicate that visual agents were not perceived as being helpful by participants, even less so when an aiming agent was present, and that the addition of a visual agent could even be distracting and harmful to task completion. Additionally, our findings could reflect the downside of studying a concept often investigated in safety-critical environments (SART, SAGAT), which may not apply to a different domain of interaction (here, a game-based task). In our studies, we found that SART scores were ineffective at capturing (often small) differences in information gains between conditions. Our findings should serve as motivation to develop a greater range of non-context-specific methods to analyse the development of SA in more details.

8.6 Lab-based and Asynchronous Online Studies

Due to disruptions caused by the COVID19 pandemic throughout 2020 and 2021, we decided to switch from using a lab-based framework (as seen in studies presented in Chapters 4 and 5) to an asynchronous online framework (see Chapters 6 and 7) where participants took part in the experiment using their own computers. While the task remained the same, we would like to discuss the implications of switching from one framework to another for our results and future HCI work in general.

First, the quantity and quality of post-hoc qualitative feedback given by participants tended to be better during lab-based studies, which could have been the result of participants being able to ask questions directly to the lead researcher and referring more easily to particular moments during the experiments, compared to writing feedback on a post-hoc survey form.

Secondly, while asynchronous online studies make it easier to reach a higher number of people, we noticed that lab-based studies require less attention to details during the creation of the experimental flow and interface, as a researcher can be present to guide users throughout the experiment and assist them if needed. This added development time for online studies should be taken into account when weighting the pros and cons of remote and lab-based experiments, as more pilot testing is required to ensure that the flow of an online experiment remains straightforward enough for all participants to complete in time, as well as potentially designing attention checks.

Finally, we recommend for the network and performance of participants' computers to be assessed during and, equally importantly, *before* they take part in the study. To this end, we recommend creating a simple, scaled-down version of the experimental framework to test communication to servers (checking participants' ping to the experiment's servers could be useful) as well as a benchmarking tool (similarly to the one used in all of our online studies) to let prospective participants know if their machines are fit to complete the study or not. This is especially important when the framework is being ran directly from participants' computers. This benchmark tool can be as simple as a short script using the same framework as the experiment and running a high workload where certain performance thresholds have to be met in order to allow prospective participants to take part in the study.

8.7 Limitations

8.7.1 Experimental Constraints

It should be noted that our work is not without limitations. In the studies presented in Chapters 4 an 5, we explored agent's behaviours and their influence on trust, reliance, task performance, cognitive workload and situational awareness in a goal-oriented, collaborative aiming task. While the selection criteria for the recruitment of participants were similar for all studies, our samples mostly consisted of students between the age of 18 and 30 and might not generalise to other, older, user groups. In addition, in order to ensure that studies could be completed within an hour for lab-based studies and 45 minutes for online studies, a number of constraints were set. We had to design our experiments with either four or five conditions, depending on the focus of each study (for instance, four different agents or types of visual occlusion). The duration for which participants interacted with each condition also had to be constrained. In most of our studies, participants spent between three and six minutes on each condition, which could have limited the amount of time participants had in order to get used to and adapt to different experimental conditions. In particular, it is possible that more time spent interacting with the agents could have helped participants calibrate their trust over time, which could coincided with changes in reliance and task performance. On the other hand, interactions that were too lengthy could have lead to complacency or complete distrust.

Additionally, of the four user studies conducted in this thesis, two took place in a lab environment (see Chapters 4 and 5) while two others were conducted online and asynchronously (see Chapters 6 and 7) due to disruptions caused by the COVID19 pandemic throughout 2020 and 2021. While we used the same framework for all studies, participants in remote online experiments took part using their own computers and keyboards as opposed to a controller in a lab-based environment. While participants' task performance in all studies was found to be comparable, we would like to acknowledge this as a limitation for the inter-comparability of our lab-based and remote studies. Furthermore, while we recorded software-related details about participants' machines such as frame-rate or screen resolution, hardware-level information were beyond the scope of our logging system. Screen-size, among other elements such as screen refresh rate, could be a relevant information to record as it has the potential to affect participants' ability to notice on-screen items and can change the way they perform during the task.

Longitudinal studies could be an interesting avenue that we left to future work. Such studies could focus on different aspects of the human-agent collaboration. For instance, when studying the impact of agent behaviours on users, such studies could assess the amount of time needed for users to feel comfortable enough with an agent, and how long it takes before potentially complacent attitudes set in. Findings from such experiments could further our understanding of the nature of trust calibration by linking it to the evolution of performance levels, reliance and/or reported trust. Furthermore, longitudinal studies including adverse environmental conditions would also be beneficial to the field of HAI. Such experiments could test a wide variety of adversarial conditions affecting the accuracy of the agent's help and/or the knowledge available to users. These studies could be easily contextualised by using scenarios most commonly affected by changes in the weather, terrain or location. While remaining relatively rare, these studies would prove useful to understand the impact of external factors on HAC, which have, for now, mostly been studied in war-oriented scenarios [99].

8.7.2 Domain-specific Constraints

Our framework was designed to assess human-agent collaboration in a task characterised by the following elements:

- A fast-paced task where participants had to rely on their reflexes to perform well.
- Multiple targets to monitor and track at once.
- A task divided into short sessions, the lengths of which depended on the constraints of each study.

We acknowledge that our findings may not generalise to different contexts of interaction, especially those where decisions will have an immediate or long-lasting impact on either the user interacting with the agent or other parties (for instance, civilians involved in the decisions of a bomb-seeking task [143] or practitioners in the medical domain [72]). Nonetheless, our findings and research framework could prove useful to other scientific fields concerned with collaborative decision-making such as autonomous vehicles where challenges involving uncertainty, appropriate reliance and explainability have been long-standing issues, as presented in the meta review of trust-related studies involving autonomous vehicles by Lockey et al. [113]. Indeed, our studies provided some insights regarding how, for instance, specific visualisations could support HAC. Our results could serve as the basis for follow-up studies where the validity of our findings and experimental design are put to the test in more realistic environments.

In addition, participants interacted with agents in tasks for which they had received little training, and which were necessarily short in nature to reduce fatigue. A completely different environment (for instance an asset tracking task [143]) where users' attention has to be sustained during hours, or even days (e.g. a submarine tracking task [28]) could yield different results. Longer interactions could allow users to have a better mental model of the agent and its decision-making, but could also increase the chance of complacent behaviours developing. Participants' expertise is another area that was under-explored in this thesis. While participants' skills and

individual levels of performance were controlled by having them complete the task without the help of an agent first, our framework was new to all participants and previous expertise in related tasks was not assessed prior to undertaking the experiment. In more expert settings, the attitude users displayed towards the agent would likely be very different, as new tasks would require users to rethink their methods which could be, in turn, result in under-reliance on the agent, depending on the context of interaction.

We would like to note that the limitations mentioned above do not undermine the main findings of our studies, but we acknowledge that additional investigations are required to understand more precisely the relationship between the different variables linked to trust in agents, as well as how other types of tasks influence this relationship. We leave these avenues open for future work.

8.8 Recommendations

Based on the results gathered from studies conducted in this thesis along with past work in HAI and Human Factors, we present a series of four recommendations for the research and development of future collaborative agents. A summary of the areas we covered in this work as well as future research opportunities is provided in Figure 8.1.



Figure 8.1: Venn diagram representing interactions between a user, agent, environment of interaction and type of task. Blue areas indicate which combination this work studied while orange areas are avenues left for future work.

8.8.1 Agent transparency and consistency can be prioritised over reliability

Regarding agent transparency and consistency, our recommendations can be summarised as follows:

- Beyond a certain level of agent reliability, system makers should prioritise more predictable and easy to understand agent behaviours, rather than maximising reliability at all costs.
- Increasing transparency regarding the agent's actions and providing a better on-boarding process should result in better calibration of users' reliance and trust in the agent.

Evidently, agent reliability is one of the core factors in users' adoption of any system, and this applies to human-agent collaboration: the agent must be reliable "enough" to carry a task successfully and be perceived as helpful [57, 103]. Quantifying how reliable the agent should be in order to be perceived as "good" is, however, difficult, as it largely depends on users' own expertise and the context of interaction, including the likely consequences of any decisions taken. In previous HAI studies, the threshold for acceptable performance was usually set at around 80% accuracy [21, 57], as we found that users tend to ascribe a high reliability to any agent whose input are correct (true positives) between 90 and 70% of the time. While high reliability is essential, we argue that beyond a certain threshold, further increases in reliability can yield diminishing returns, and even harm the Human-Agent team. These side effects could occur if they imply sacrifices in the transparency of the agent's actions or the consistency of its behaviours. As evidenced by our findings in Chapters 5 and 7, assuming agents' levels of reliability are the same, users tend to prefer more predictable. This preference for higher predictability and consistency is evidenced by higher reported trust in agents that would choose not to do anything (false negative error) rather than acting and failing in a situation where the likelihood of agent errors was high. We posit that increased agent consistency makes it not only easier to anticipate an agent's actions but also allows for more flexibility in the case of sudden changes in mission goals or the environment of interaction. More transparent and predictable agent actions would likely further reduce users' cognitive workload by allowing for a more calibrated mental model of the agent's reasoning process. Increasing predictability could also help mitigate adoption and on-boarding issues, as we found that users ascribe fewer negative intents to agents that are more consistent in their behaviours.

While high agent reliability, transparency and consistency are important, there is no alternative to a well calibrated, informed understanding of an agent's capabilities. Making users aware of the weaknesses of an agent, through repeated training and/or by displaying real-time information, is paramount to achieving effective human-agent collaboration and will foster more informed reliance on the agent.

8.8.2 Agent behaviours should be clearly categorised and defined

Regarding agent behaviours, our recommendations can be summarised as follows:

- The perceived behaviour of an agent in terms of its reasoning process and perceived intent should be clearly elicited and defined in order to anticipate the impact on users' perception of the agent.
- A framework of agent behaviours, errors and likely consequences should be created. In particular, this framework should vary the context of interaction to incorporate a wide range of domain-specific findings.

The type of support and level of autonomy of an agent will influence the way users develop a mental model of the inner workings of the agent [119]. While anthropomorphised agents will be more likely to have intents ascribed to them [7], intent will also be ascribed to agents that don't display human characteristics (such as speech, or a form of physical embodiment). This tendency for users to anthropomorphise systems should be taken into account, if not leveraged, by system makers. Wrongfully ascribing a specific intent to an agent (for instance, about the perception of its "real" intentions) could have harmful consequences. As an example, users could believe that a system is actually working against them which could likely lead to underreliance issues and defeat the whole purpose of designing a collaborative agent. On the contrary, not being transparent about the limitations of an agent's decision-making process could lead to dangerous over-reliance in situations of high uncertainty.

In our studies, we mostly documented participants' impressions of agents' capabilities via trust-related ratings scales or open-ended post-hoc interviews. We found that not only does the type of agent behaviour have a clear impact on users' perception of the agents, it also influences the way users perform and rely on the agent. These differences are important to take into account, as our studies showed that certain types of error are perceived as being less harmful than others, while not resulting in improved task performance. The work of Marinaccio [118] and Reason [145] pioneered the elicitation and categorisation of agent errors and their consequences for interpersonal or human-agent interactions. We recommend that more work should focus on testing newly defined and existing automation errors in a variety of contexts, from general non-expert scenarios to safety-critical settings requiring expert knowledge. More information on potential differences and similarities of users' perceptions of errors across different domains could allow system makers to better anticipate and mitigate adoption issues, with a selection of repair mechanisms designed to account for an agent's shortcomings.

8.8.3 Taking into account the environment of interaction to better anticipate changes in human-agent collaboration

Regarding the environment in which HAC takes places, we summarise our recommendations as follows:

- In future HAC studies, the environment of interaction and its features should be clearly controlled and accounted for independently of the task.
- As with agent behaviours, a framework of different types of environment of interaction and their features (availability of required information, noise, uncertainty) should be created to compare their influence across a wide range of HAC tasks.

As we have seen with the behaviour of virtual agents, the environment of interaction is an often under-studied aspect of HAI experiments. In our studies, we found that the environment of interaction also impacts *users* in a significant way, regardless of the actual capabilities of the agent. For instance, in environments with high levels of uncertainty, users are likely to trust an agent more irrespective of its reliability, which subsequently induces more reliance on the agent's input. Additionally, we found that restricting the amount of information required to make informed decisions led participants to react more quickly and, as a result, perform better. We believe that human-agent training could benefit from such findings. In simulated environments, uncertainty and sub-optimal access to important information could be artificially induced in order to train participants to prioritise tasks more effectively and be more critical of the help provided by an agent.

As with past studies investigating and classifying different types of agent errors and their impact on users [118], there is, to the best of our knowledge, no work focused on assessing how different types of tasks, constraints, and environments of interaction in general affect humanagent collaboration. Such a framework could detail how the features of different environments of interaction (such as availability of information, noise or uncertainty level) would affect the outcome of a task as well as the general attitude of a user towards an agent. This work should also carefully account for the risks and consequences of each interaction scenario, as the outcome of a task (in safety critical environments for instance) are likely to influence users' decision-making and risk-taking potential.

8.8.4 Creating a framework to categorise transparent interface elements and their impact on situational awareness

Regarding the communication of an agent's intent and decision-making process, we recommend the following:

- More reproduction studies on past SAT-based work.
- The development of more general SA assessment tools for use in non safetycritical contexts.

Situational awareness (SA) and the related SAT framework are often used as a basis for qualifying and designing interfaces that support increased agent transparency. Most SA studies have been conducted in defence-oriented, military-focused environments where task require users to assess situations over extended periods of time (asset tracking, dispatch missions etc.). While insights from SA studies are a useful way to categorise the impact of different visual aids on SA in high-risk situations, they are less helpful when it comes to selecting specific types of visualisations for a given task. We recommend that Human Factors and HCI researchers reproduce past SA-based designs and develop new ones in order to test their effectiveness in more varied scenarios. In particular, we recommend that more studies are conducted in non safetycritical environments. This would allow for a better understanding of which characteristics inherent to visual aids impact SA the most, and which contexts of interaction make it most relevant to assess SA in the first place.

As a follow-up study, and using results from reproduction studies, we recommend that research efforts are centred around designing more context-free assessment tools for SA. Current SA assessment tools are either deployed in a context-specific manner (SAGAT [53]) or as a series of rating-scale instruments (SART [53]). Most tools have been designed and calibrated in safety-critical environments. However, and line with the results of our non-safety critical user studies, findings from previously designed SA assessment tools may not prove conclusive in tasks designed to be accessible to non-expert users. For instance, the question wording in the SART tool clearly refers to concept from aviation, where it was originally designed. These instruments may not resonate with a general audience. If SA is to be an important factor in assessing and monitoring a wide range of scenarios, a more general means of assessment is needed, one that (like NASA TLX [76] for cognitive workload, for instance) could be easily administered in a wide range of scenarios. These results could help create a framework where likely drawbacks and advantages of specific visualisations are presented, along with their relationship to different types of task.

We hope that these recommendations will speak to designers of human-agent systems and be tested in a wide range of interaction scenarios. For future work on trust in collaborative agents, we recommend that studies focus on the development of trust-aware agents, which includes the design of systems capable of *detecting loss of trust in real-time* and *repairing trust via a series of mechanisms* depending on the type of error committed and the context of interaction. The notion of "trust-aware agents" is most commonly found in multi-agents environments where trust is often operationalised as a score based on system's performance and used to plan future decisions, such as in the context of AI-supported transportation systems [30]. It would be insightful for future work to extend the concept of trust awareness to agents involved in more social interactions where the detection and assessment of trust is usually more complex. These studies could then deploy new or previously studied [145] trust-repair mechanisms and assess their relevance in a variety of scenarios and tasks.

Chapter 9 Conclusion

Collaboration between human and automated agents is now commonplace in many applied settings, from safety environments to more casual, everyday activities. Due to the complex interplay between users, agents and the environment of interaction, it is important to study specific factors that are likely to influence how users trust an agent, as well as the implications for task performance, reliance and other reported metrics. In this thesis, we have presented four user-studies designed to test the impact of different agent behaviours and visualisations on users. To perform all of our experiments, we designed a game-like human-agent framework (see Chapter 3) where users and agents collaborated to complete goal-oriented tasks with various levels of difficulty. Using this framework, we were able to record behavioural metrics to assess task performance and reliance, as well as reported constructs with the use of verified survey instruments. We would like our results to inform the design of collaborative agents and motivate future research into key components of the human-agent partnership.

In this Section, we summarise our answers to the overall research questions defined in Section 1.3. Our first research questions was defined as follows: How do changes in agent predictability (how easy it is to guess its next actions) and reliability (how good the agent is at the task) impact the human-agent relationship? With our lab-based user study presented in Chapter 4, we showed that, when agent reliability is high, added predictability increases task performance, reliance on the agent, trust and reduces cognitive load.

Our second research question was defined as follows: How do different types of agent errors defined from previous related work such as slips, mistakes and lapses affect the human-agent relationship? With our lab-based user study presented in Chapter 5, we demonstrated that, when agent reliability is high, the type of error committed by an agent affects the way users interact with it. Errors of omission (defined as "lapses") are the easiest to correct and can even improve task performance in terms of Precision by reducing the chance of users engaging in complacent behaviours. Errors of intention ("mistakes") and commission ("slips") were harder for users to correct, and resulted in the worst task performance as well as higher reported cognitive load.

Our third research question was defined as follows: How do different types of environmental conditions (static or moving), which impair vision and induce uncertainty, affect the human-agent relationship?. With our remote online user study presented in Chapter 6, we showed that conditions impairing vision and inducing visual uncertainty can affect the way users perform and rely on an agent. Near-total visual uncertainty led to higher reported trust and increased reliance on the agent while negatively affecting task performance. Visual occlusions that forced participants to react more quickly led to higher task performance, despite also resulting in a higher reported cognitive load.

Our fourth and final research question was defined as follows: How do different types of visual help (designed to elicit different levels of situational awareness) influence the human-agent relationship? With our remote online user study presented in Chapter 7, we showed that the type of visualisation used in a human-agent collaborative task has important implications for human-agent collaboration. Visualisations that are descriptive in nature (highlighting important elements) were perceived as more helpful and led to higher task performance than visualisations that were prescriptive in nature (telling the user "what to do"). Overall, visualisations designed to support SA level 1 and 2 had a more positive influence on users while an SA level 3 visualisation led to lower task performance and a higher cognitive load.

Bibliography

- ADLER, R. F., AND BENBUNAN-FICH, R. The effects of task difficulty and multitasking on performance. *Interacting with Computers* 27, 4 (2015), 430–439.
- [2] AHMAD, M. I., BERNOTAT, J., LOHAN, K., AND EYSSEL, F. Trust and cognitive load during human-robot interaction. arXiv (2019).
- [3] AJENAGHUGHRURE, I. B., SOUSA, S. C., KOSUNEN, I. J., AND LAMAS, D. Predictive model to assess user trust: a psycho-physiological approach. In *Proceedings of the 10th Indian Conference on Human-Computer Interaction* (2019), pp. 1–10.
- [4] ALAN, A. T., LIU, C., SALISBURY, E., PRIOR, S. D., RAMCHURN, S. D., WU, F., TATLOCK, K., AND REES, G. Human-uav teaming in dynamic and uncertain environments. In Proceedings of the 17th International Conference on Autonomous Agents and Multi-Agent Systems (AAMAS) (2018), pp. 1791–1793.
- [5] ALJUKHADAR, M., SENECAL, S., AND DAOUST, C.-E. Using recommendation agents to cope with information overload. *International Journal of Electronic Commerce* 17, 2 (2012), 41–70.
- [6] ALVARADO-VALENCIA, J. A., AND BARRERO, L. H. Reliance, trust and heuristics in judgmental forecasting. *Computers in Human Behavior 36* (July 2014), 102–113.
- [7] ARAUJO, T. Living up to the chatbot hype: The influence of anthropomorphic design cues and communicative agency framing on conversational agent and company perceptions. *Computers in Human Behavior 85* (2018), 183–189.
- [8] ARGOTE, L., AND GOODMAN, P. S. The organizational implications of robotics. Implementing advanced technology (1985).
- [9] ARMSTRONG, R. A. When to use the b onferroni correction. Ophthalmic and Physiological Optics 34, 5 (2014), 502–508.
- [10] ATARI, I. Missile Command. Game [Atari 2600], July 1980. Atari, Inc, United States.

- [11] BAKER, A. L., PHILLIPS, E. K., ULLMAN, D., AND KEEBLER, J. R. Toward an understanding of trust repair in human-robot interaction: Current research and future directions. ACM Transactions on Interactive Intelligent Systems (TiiS) 8, 4 (2018), 1–30.
- [12] BERNER, C., BROCKMAN, G., CHAN, B., CHEUNG, V., DEBIAK, P., DENNISON, C., FARHI, D., FISCHER, Q., HASHME, S., HESSE, C., ET AL. Dota 2 with large scale deep reinforcement learning. arXiv preprint arXiv:1912.06680 (2019).
- [13] BISBEY, R., AND KAY, M. The mind (management of information through natural discourse) translation system: a study in man-machine collaboration. Tech. rep., RAND CORP SANTA MONICA CALIF, 1972.
- [14] BOUBIN, J. G., RUSNOCK, C. F., AND BINDEWALD, J. M. Quantifying compliance and reliance trust behaviors to influence trust in human-automation teams. *Proceedings of the Human Factors and Ergonomics Society Annual Meeting 61*, 1 (Sept. 2017), 750–754.
- [15] BRADLEY, J. V. Complete counterbalancing of immediate sequential effects in a latin square design. Journal of the American Statistical Association 53, 282 (1958), 525–528.
- [16] BRADSHAW, J. M., FELTOVICH, P., JOHNSON, M., BREEDY, M., BUNCH, L., ES-KRIDGE, T., JUNG, H., LOTT, J., USZOK, A., AND VAN DIGGELEN, J. From tools to teammates: Joint activity in human-agent-robot teams. In *International conference on human centered design* (2009), Springer, pp. 935–944.
- [17] CAO, A., CHINTAMANI, K. K., PANDYA, A. K., AND ELLIS, R. D. Nasa tlx: Software for assessing subjective mental workload. *Behavior research methods* 41, 1 (2009), 113– 117.
- [18] CAO, A., CHINTAMANI, K. K., PANDYA, A. K., AND ELLIS, R. D. NASA TLX: Software for assessing subjective mental workload. *Behavior Research Methods* 41, 1 (Feb. 2009), 113–117.
- [19] CARTER, M., DOWNS, J., NANSEN, B., HARROP, M., AND GIBBS, M. Paradigms of games research in hci: a review of 10 years of research at chi. In *Proceedings of the* first ACM SIGCHI annual symposium on Computer-human interaction in play (2014), pp. 27–36.
- [20] CARTER, M., DOWNS, J., NANSEN, B., HARROP, M., AND GIBBS, M. Paradigms of games research in hci: A review of 10 years of research at chi. In *Proceedings of the First* ACM SIGCHI Annual Symposium on Computer-Human Interaction in Play (New York, NY, USA, 2014), CHI PLAY '14, Association for Computing Machinery, p. 27–36.

- [21] CHANCEY, E. T., BLISS, J. P., YAMANI, Y., AND HANDLEY, H. A. Trust and the Compliance-Reliance Paradigm: The Effects of Risk, Error Bias, and Reliability on Trust and Dependence. *Human Factors* 59, 3 (2017), 333–345.
- [22] CHANG, M., KUO, R., AND LIU, E. Z. F. Revised computer game attitude scale. Proceedings of the 22nd International Conference on Computers in Education, ICCE 2014 (2014), 598–607.
- [23] CHARISSIS, V., AND PAPANASTASIOU, S. Human-machine collaboration through vehicle head up display interface. *Cognition, Technology & Work 12*, 1 (2010), 41–50.
- [24] CHAVAILLAZ, A., SCHWANINGER, A., MICHEL, S., AND SAUER, J. Automation in visual inspection tasks: X-ray luggage screening supported by a system of direct, indirect or adaptable cueing with low and high system reliability. *Ergonomics 61*, 10 (Oct. 2018), 1395–1408.
- [25] CHAVAILLAZ, A., WASTELL, D., AND SAUER, J. System reliability, performance and trust in adaptable automation. *Applied Ergonomics* 52 (2016), 333–342.
- [26] CHEN, J. Y., PROCCI, K., BOYCE, M., WRIGHT, J., GARCIA, A., AND BARNES, M. Situation awareness-based agent transparency. Tech. rep., Army research lab aberdeen proving ground md human research and engineering ..., 2014.
- [27] CHEN, J. Y. C., LAKHMANI, S. G., STOWERS, K., SELKOWITZ, A. R., WRIGHT, J. L., AND BARNES, M. Situation awareness-based agent transparency and human-autonomy teaming effectiveness. *Theoretical Issues in Ergonomics Science 19*, 3 (Feb. 2018), 259– 282.
- [28] CHEN, S., LOFT, S., HUF, S., BRAITHWAITE, J., AND VISSER, T. Static and adaptable automation in simulated submarine track management. In *Proceedings of the Human Factors and Ergonomics Society Annual Meeting* (2014), vol. 58, SAGE Publications Sage CA: Los Angeles, CA, pp. 2280–2284.
- [29] CHEN, T., CAMPBELL, D., GONZALEZ, L. F., AND COPPIN, G. Increasing Autonomy Transparency through capability communication in multiple heterogeneous UAV management. In *IEEE International Conference on Intelligent Robots and Systems* (2015).
- [30] CHENG, M., ZHANG, J., NAZARIAN, S., DESHMUKH, J., AND BOGDAN, P. Trustaware control for intelligent transportation systems. In 2021 IEEE Intelligent Vehicles Symposium (IV) (2021), IEEE, pp. 377–384.

- [31] CHRISTEL, M. G., STEVENS, S. M., MAHER, B. S., BRICE, S., CHAMPER, M., JAYA-PALAN, L., CHEN, Q., JIN, J., HAUSMANN, D., BASTIDA, N., ET AL. Rumbleblocks: Teaching science concepts to young children through a unity game. In 2012 17th International Conference on Computer Games (CGAMES) (2012), IEEE, pp. 162–166.
- [32] CIPOLLONE, M., SCHIFTER, C. C., AND MOFFAT, R. A. Minecraft as a creative tool: A case study. International Journal of Game-Based Learning (IJGBL) 4, 2 (2014), 1–14.
- [33] COMPANY, P. Prolific, 2021.
- [34] CORREIA, F., GUERRA, C., MASCARENHAS, S., MELO, F. S., AND PAIVA, A. Exploring the impact of fault justification in human-robot trust. In *Proceedings of the 17th International Conference on Autonomous Agents and MultiAgent Systems* (Richland, SC, 2018), AAMAS '18, International Foundation for Autonomous Agents and Multiagent Systems, pp. 507–513.
- [35] COSTANZA, E., FISCHER, J. E., COLLEY, J. A., RODDEN, T., RAMCHURN, S. D., AND JENNINGS, N. R. Doing the laundry with agents: a field trial of a future smart energy system in the home. In *Proceedings of the SIGCHI Conference on Human Factors* in Computing Systems (2014), pp. 813–822.
- [36] DARONNAT, S. Human-agent trust relationships in a real-time collaborative game. In Extended Abstracts of the 2020 Annual Symposium on Computer-Human Interaction in Play (2020), pp. 18–20.
- [37] DARONNAT, S., AZZOPARDI, L., AND HALVEY, M. Impact of agents' errors on performance, reliance and trust in human-agent collaboration. In *Human Factors and Ergonomics Society Annual Meeting* (2020), pp. 1–5.
- [38] DARONNAT, S., AZZOPARDI, L., AND HALVEY, M. Investigating the impact of visual environmental uncertainty on human-agent teaming. In *Proceedings of the Human Factors* and Ergonomics Society Annual Meeting (2021), vol. 65, SAGE Publications Sage CA: Los Angeles, CA, pp. 1185–1189.
- [39] DARONNAT, S., AZZOPARDI, L., HALVEY, M., AND DUBIEL, M. Human-agent collaborations: trust in negotiating control. *CHI 2019* (2019).
- [40] DARONNAT, S., AZZOPARDI, L., HALVEY, M., AND DUBIEL, M. Impact of agent reliability and predictability on trust in real time human-agent collaboration. In *Proceedings* of the 8th International Conference on Human-Agent Interaction (2020), pp. 131–139.

- [41] DARONNAT, S., AZZOPARDI, L., HALVEY, M., AND DUBIEL, M. Inferring trust from users behaviours; agents' predictability positively affects trust, task performance and cognitive load in human-agent real-time collaboration. *Frontiers in Robotics and AI 8* (2021), 194.
- [42] DE VISSER, E. J., KRUEGER, F., MCKNIGHT, P., SCHEID, S., SMITH, M., CHALK, S., AND PARASURAMAN, R. The world is not enough: Trust in cognitive agents. *Proceedings* of the Human Factors and Ergonomics Society Annual Meeting 56, 1 (Sept. 2012), 263– 267.
- [43] DE VISSER, E. J., PEETERS, M. M. M., JUNG, M. F., KOHN, S., SHAW, T. H., PAK, R., AND NEERINCX, M. A. Towards a theory of longitudinal trust calibration in human-robot teams. *International Journal of Social Robotics* 12, 2 (Nov. 2019), 459–478.
- [44] DEMIR, M., MCNEESE, N. J., AND COOKE, N. J. Team situation awareness within the context of human-autonomy teaming. *Cognitive Systems Research* 46 (2017), 3–12.
- [45] DURSO, F. T., HACKWORTH, C. A., TRUITT, T. R., CRUTCHFIELD, J., NIKOLIC, D., AND MANNING, C. A. Situation awareness as a predictor of performance for en route air traffic controllers. Air Traffic Control Quarterly 6, 1 (Jan. 1998), 1–20.
- [46] DZINDOLET, M. T., PETERSON, S. A., POMRANKY, R. A., PIERCE, L. G., AND BECK,
 H. P. The role of trust in automation reliance. *International journal of human-computer studies 58*, 6 (2003), 697–718.
- [47] EISENHARDT, K. M. Building theories from case study research. Academy of Management Review 14, 4 (Oct. 1989), 532–550.
- [48] ENDSLEY, M. R. Design and evaluation for situation awareness enhancement. In Proceedings of the Human Factors Society annual meeting (1988), vol. 32, SAGE Publications Sage CA: Los Angeles, CA, pp. 97–101.
- [49] ENDSLEY, M. R. Situation awareness global assessment technique (sagat). In Proceedings of the IEEE 1988 national aerospace and electronics conference (1988), IEEE, pp. 789– 795.
- [50] ENDSLEY, M. R. Toward a theory of situation awareness in dynamic systems. Human factors 37, 1 (1995), 32–64.
- [51] ENDSLEY, M. R. Direct measurement of situation awareness: Validity and use of sagat. In Situational awareness. Routledge, 2017, pp. 129–156.

- [52] ENDSLEY, M. R. A systematic review and meta-analysis of direct objective measures of situation awareness: A comparison of SAGAT and SPAM. *Human Factors: The Journal* of the Human Factors and Ergonomics Society (Sept. 2019), 001872081987537.
- [53] ENDSLEY, M. R., SELCON, S. J., HARDIMAN, T. D., AND CROFT, D. G. Comparative analysis of SAGAT and SART for evaluations of situation awareness. *Proceedings of the Human Factors and Ergonomics Society* 1 (1998), 82–86.
- [54] EPIC GAMES. Unreal engine.
- [55] EREBAK, S., AND TURGUT, T. Caregivers' attitudes toward potential robot coworkers in elder care. Cognition, Technology & Work 21, 2 (2019), 327–336.
- [56] FAN, X., MCNEESE, M., SUN, B., HANRATTY, T., ALLENDER, L., AND YEN, J. Human-Agent Collaboration for Time-Stressed Multicontext Decision Making. *IEEE Transactions on Systems, Man, and Cybernetics - Part A: Systems and Humans 40*, 2 (Mar. 2010), 306–320.
- [57] FAN, X., OH, S., MCNEESE, M., YEN, J., CUEVAS, H., STRATER, L., AND ENDSLEY, M. R. The influence of agent reliability on trust in human-agent collaboration. In Proceedings of the 15th European conference on Cognitive ergonomics: the ergonomics of cool interaction (2008), ACM, p. 7.
- [58] FAN, X., OH, S., MCNEESE, M., YEN, J., CUEVAS, H., STRATER, L., AND ENDSLEY, M. R. The influence of agent reliability on trust in human-agent collaboration. Proceedings of the 15th European conference on Cognitive ergonomics the ergonomics of cool interaction - ECCE '08 (2008), 1.
- [59] FINK, J. Anthropomorphism and human likeness in the design of robots and human-robot interaction. In International Conference on Social Robotics (2012), Springer, pp. 199–208.
- [60] FISCHER, J. E., GREENHALGH, C., JIANG, W., RAMCHURN, S. D., WU, F., AND RODDEN, T. In-the-loop or on-the-loop? interactional arrangements to support team coordination with a planning agent. *Concurrency and Computation: Practice and Experience 33*, 8 (2021), e4082.
- [61] FISCHER, J. E., JIANG, W., AND MORAN, S. Atomicorchid: A mixed reality game to investigate coordination in disaster response. In *International Conference on Entertainment Computing* (2012), Springer, pp. 572–577.
- [62] FLANAGAN, J. C. The critical incident technique. Psychological Bulletin 51, 4 (1954), 327–358.

- [63] FLEISS, J. L., LEVIN, B., AND PAIK, M. C. Statistical methods for rates and proportions. john wiley & sons, 2013.
- [64] GAMES, P. A., AND HOWELL, J. F. Pairwise multiple comparison procedures with unequal n's and/or variances: a monte carlo study. *Journal of Educational Statistics* 1, 2 (1976), 113–125.
- [65] GARNICK, C. J., BINDEWALD, J. M., AND RUSNOCK, C. F. Designing an automated agent to encourage human reliance. *Proceedings of the Human Factors and Ergonomics Society 2017-October* (2017), 1730–1734.
- [66] GILLESPIE, N., AND DIETZ, G. Trust repair after an organization-level failure. Academy of management review 34, 1 (2009), 127–145.
- [67] GIRDEN, E. R. ANOVA: Repeated measures. No. 84 in Quantitative Applications in the Social Sciences. Sage, 1992.
- [68] GISH, K. W., AND STAPLIN, L. Human factors aspects of using head up displays in automobiles: A review of the literature. U.S. Department of Transportation, National Highway Traffic Safety Administration (1995).
- [69] GODOT ENGINE DEVELOPMENT TEAM. Godot engine.
- [70] GOMBOLAY, M., BAIR, A., HUANG, C., AND SHAH, J. Computational design of mixedinitiative human-robot teaming that considers human factors: situational awareness, workload, and workflow preferences. *International Journal of Robotics Research 36*, 5-7 (2017), 597–617.
- [71] GORDON, N. Colour blindness. Public health 112, 2 (1998), 81-84.
- [72] GRAAFLAND, M., BEMELMAN, W. A., AND SCHIJVEN, M. P. Game-based training improves the surgeon's situational awareness in the operation room: a randomized controlled trial. *Surgical endoscopy 31*, 10 (2017), 4093–4101.
- [73] GRABOWSKI, M., AND SANBORN, S. D. Human performance and embedded intelligent technology in safety-critical systems. *International journal of human-computer studies* 58, 6 (2003), 637–670.
- [74] HAAS, J. A history of the unity game engine. Diss. WORCESTER POLYTECHNIC INSTITUTE (2014).
- [75] HART, S. G. Nasa-task load index (nasa-tlx); 20 years later. In Proceedings of the human factors and ergonomics society annual meeting (2006), vol. 50, Sage publications Sage CA: Los Angeles, CA, pp. 904–908.
- [76] HART, S. G., AND STAVELAND, L. E. Development of nasa-tlx (task load index): Results of empirical and theoretical research. In *Advances in psychology*, vol. 52. Elsevier, 1988, pp. 139–183.
- [77] HART, S. G., AND STAVELAND, L. E. Development of nasa-tlx (task load index): Results of empirical and theoretical research. In *Advances in psychology*, vol. 52. Elsevier, 1988, pp. 139–183.
- [78] HAWKINS, K. P., BANSAL, S., VO, N. N., AND BOBICK, A. F. Anticipating human actions for collaboration in the presence of task and sensor uncertainty. In 2014 ieee international conference on Robotics and automation (ICRA) (2014), IEEE, pp. 2215– 2222.
- [79] HEIDLAUF, P., COLLINS, A., BOLENDER, M., AND BAK, S. Verification challenges in f-16 ground collision avoidance and other automated maneuvers. In ARCH@ ADHS (2018), pp. 208–217.
- [80] HEO, J. S., AND LEE, K. Y. A multi-agent system-based intelligent heuristic optimal control system for a large-scale power plant. 2006 IEEE Congress on Evolutionary Computation, CEC 2006 (2006), 1544–1551.
- [81] HERDENER, N., CLEGG, B. A., WICKENS, C. D., AND SMITH, C. A. Anchoring and Adjustment in Uncertain Spatial Trajectory Prediction. *Human Factors* 61, 2 (2019), 255–272.
- [82] HOC, J. M. From human machine interaction to human machine cooperation. Ergonomics 43, 7 (2000), 833–843.
- [83] HOC, J. M., YOUNG, M. S., AND BLOSSEVILLE, J. M. Cooperation between drivers and automation: Implications for safety. *Theoretical Issues in Ergonomics Science* 10, 2 (2009), 135–160.
- [84] HOC, J.-M., YOUNG, M. S., AND BLOSSEVILLE, J.-M. Cooperation between drivers and automation: implications for safety. *Theoretical Issues in Ergonomics Science* 10, 2 (2009), 135–160.
- [85] HOFF, K. A., AND BASHIR, M. Trust in automation: Integrating empirical evidence on factors that influence trust. *Human Factors* 57, 3 (2015), 407–434.
- [86] HOFFMAN, R. R., JOHNSON, M., BRADSHAW, J. M., AND UNDERBRINK, A. Trust in automation. *IEEE Intelligent Systems 28*, 1 (2013), 84–88.

- [87] HUSSEIN, A., ELSAWAH, S., AND ABBASS, H. A. Trust Mediating Reliability-Reliance Relationship in Supervisory Control of Human-Swarm Interactions. *Human Factors 62*, 8 (2020), 1237–1248.
- [88] IORDANESCU, L., GRABOWECKY, M., AND SUZUKI, S. Demand-based dynamic distribution of attention and monitoring of velocities during multiple-object tracking. *Journal* of vision 9, 4 (2009), 1–1.
- [89] JENSEN, T., KHAN, M. M. H., ALBAYRAM, Y., FAHIM, M. A. A., BUCK, R., AND COMAN, E. Anticipated emotions in initial trust evaluations of a drone system based on performance and process information. *International Journal of Human-Computer Interaction 36*, 4 (2020), 316–325.
- [90] JIAN, J.-Y., BISANTZ, A. M., AND DRURY, C. G. Foundations for an Empirically Determined Scale of Trust in Automated Systems. *International Journal of Cognitive Ergonomics* 4, 1 (Mar. 2000), 53–71.
- [91] JUNG, M. F., DIFRANZO, D., STOLL, B., SHEN, S., LAWRENCE, A., AND CLAURE,
 H. Robot Assisted Tower Construction A Resource Distribution Task to Study Human-Robot Collaboration and Interaction with Groups of People. arXiv (2018), 1–18.
- [92] KARIKAWA, D., AOYAMA, H., TAKAHASHI, M., FURUTA, K., WAKABAYASHI, T., AND KITAMURA, M. A visualization tool of en route air traffic control tasks for describing controller's proactive management of traffic situations. *Cognition, Technology & Work* 15, 2 (2013), 207–218.
- [93] KILGORE, R., AND VOSHELL, M. Increasing the transparency of unmanned systems: Applications of ecological interface design. Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics) 8526 LNCS, PART 2 (2014), 378–389.
- [94] KIM, P. H., COOPER, C. D., DIRKS, K. T., AND FERRIN, D. L. Repairing trust with individuals vs. groups. Organizational Behavior and Human Decision Processes 120, 1 (jan 2013), 1–14.
- [95] KIM, T., AND SONG, H. How should intelligent agents apologize to restore trust? interaction effects between anthropomorphism and apology attribution on trust repair. *Telematics and Informatics 61* (2021), 101595.
- [96] KJELDSKOV, J., AND GRAHAM, C. A review of mobile hci research methods. In International Conference on Mobile Human-Computer Interaction (2003), Springer, pp. 317–335.

- [97] KLIEN, G., WOODS, D. D., BRADSHAW, J. M., HOFFMAN, R. R., AND FELTOVICH, P. J. Ten challenges for making automation a" team player" in joint human-agent activity. *IEEE Intelligent Systems 19*, 6 (2004), 91–95.
- [98] KLIEN, G., WOODS, D. D., BRADSHAW, J. M., HOFFMAN, R. R., AND FELTOVICH, P. J. Ten challenges for making automation a" team player" in joint human-agent activity. *IEEE Intelligent Systems* 19, 6 (2004), 91–95.
- [99] KOTT, A. Challenges and characteristics of intelligent autonomy for internet of battle things in highly adversarial environments. arXiv preprint arXiv:1803.11256 (2018).
- [100] KRÄMER, N. C., VON DER PÜTTEN, A., AND EIMLER, S. Human-agent and humanrobot interaction theory: similarities to and differences from human-human interaction. In *Human-computer interaction: The agency perspective*. Springer, 2012, pp. 215–240.
- [101] KUNZE, A., SUMMERSKILL, S. J., MARSHALL, R., AND FILTNESS, A. J. Automation transparency: implications of uncertainty communication for human-automation interaction and interfaces. *Ergonomics* 62, 3 (Feb. 2019), 345–360.
- [102] LEBAS, M. J. Performance measurement and performance management. International Journal of Production Economics 41, 1-3 (1995), 23–35.
- [103] LEE, J. D., AND SEE, K. A. Trust in Automation: Designing for Appropriate Reliance. Human Factors: The Journal of the Human Factors and Ergonomics Society 46, 1 (2004), 50–80.
- [104] LEVENE, H. Robust tests for equality of variances. Contributions to probability and statistics. Essays in honor of Harold Hotelling (1961), 279–292.
- [105] LEWICKI, R. J., BUNKER, B. B., ET AL. Developing and maintaining trust in work relationships. Trust in organizations: Frontiers of theory and research 114 (1996), 139.
- [106] LEWICKI, R. J., WIETHOFF, C., AND TOMLINSON, E. C. What is the role of trust in organizational justice. *Handbook of organizational justice* (2005), 247–270.
- [107] LEWIS, J., BROWN, D., CRANTON, W., AND MASON, R. Simulating visual impairments using the unreal engine 3 game engine. In 2011 IEEE 1st International Conference on Serious Games and Applications for Health (SeGAH) (2011), IEEE, pp. 1–8.
- [108] LEWIS, J. R. The system usability scale: past, present, and future. International Journal of Human-Computer Interaction 34, 7 (2018), 577–590.
- [109] LEWIS, M. Designing for human-agent interaction. AI Magazine 19, 2 (1998), 67–67.

- [110] LEWIS, M., AND JACOBSON, J. Game engines. Communications of the ACM 45, 1 (2002), 27.
- [111] LICKLIDER, J. C. Man-computer symbiosis. IRE transactions on human factors in electronics, 1 (1960), 4–11.
- [112] LIU, H. Comparing Welch ANOVA, a Kruskal-Wallis test, and traditional ANOVA in case of heterogeneity of variance. Virginia Commonwealth University, 2015.
- [113] LOCKEY, S., GILLESPIE, N., HOLM, D., AND SOMEH, I. A. A review of trust in artificial intelligence: Challenges, vulnerabilities and future directions. *Hawaii International Conference on System Sciences* (2021).
- [114] LOHSE, G. L., BIOLSI, K., WALKER, N., AND RUETER, H. H. A classification of visual representations. *Communications of the ACM 37*, 12 (1994), 36–50.
- [115] MA, Y., AND GHASEMZADEH, H. Head-mounted sensors and wearable computing for automatic tunnel vision assessment. Proceedings of the 2017 Design, Automation and Test in Europe, DATE 2017 (2017), 634–637.
- [116] MACEACHREN, A. M. Visualizing Uncertain Information. Cartographic Perspective 13, 13 (1992), 10–19.
- [117] MACKWORTH, N. H. Visual noise causes tunnel vision. Psychonomic science 3, 1 (1965), 67–68.
- [118] MARINACCIO, K., KOHN, S., PARASURAMAN, R., AND DE VISSER, E. J. A Framework for Rebuilding Trust in Social Automation Across Health-Care Domains. Proceedings of the International Symposium on Human Factors and Ergonomics in Health Care 4, 1 (jun 2015), 201–205.
- [119] MATTHEWS, G., PANGANIBAN, A. R., LIN, J., LONG, M., AND SCHWING, M. Supermachines or sub-humans: mental models and trust in intelligent autonomous systems. In *Trust in Human-Robot Interaction*. Elsevier, 2021, pp. 59–82.
- [120] MAYER, R. C., DAVIS, J. H., AND SCHOORMAN, F. D. An integrative model of organizational trust. Academy of management review 20, 3 (1995), 709–734.
- [121] MCGRAW, K. O., AND WONG, S. P. A common language effect size statistic. Psychological bulletin 111, 2 (1992), 361.
- [122] MCKIGHT, P. E., AND NAJAB, J. Kruskal-wallis test. The corsini encyclopedia of psychology (2010), 1–1.

- [123] MCKNIGHT, P. E., AND NAJAB, J. Mann-whitney u test. The Corsini encyclopedia of psychology (2010), 1–1.
- [124] MERCADO, J. E., RUPP, M. A., CHEN, J. Y., BARNES, M. J., BARBER, D., AND PROCCI, K. Intelligent Agent Transparency in Human-Agent Teaming for Multi-UxV Management. *Human Factors* 58, 3 (2015), 401–415.
- [125] MERRITT, S. M., LEE, D., UNNERSTALL, J. L., AND HUBER, K. Are well-calibrated users effective users? Associations between calibration of trust and performance on an automation-aided task. *Human Factors* 57, 1 (2015), 34–47.
- [126] MORAY, N., AND INAGAKI, T. Attention and complacency. Theoretical Issues in Ergonomics Science 1, 4 (2000), 354–365.
- [127] MUIR, B. M. Trust between humans and machines, and the design of decision aids. International Journal of Man-Machine Studies 27, 5-6 (1987), 527–539.
- [128] NOVITZKY, M., ROBINETTE, P., BENJAMIN, M. R., GLEASON, D. K., FITZGERALD, C., AND SCHMIDT, H. Preliminary interactions of human-robot trust, cognitive load, and robot intelligence levels in a competitive game. In *Companion of the 2018 ACM/IEEE* international conference on human-robot interaction (2018), pp. 203–204.
- [129] OGRETEN, S., LACKEY, S., AND NICHOLSON, D. Recommended roles for uninhabited team members within mixed-initiative combat teams. In 2010 International Symposium on Collaborative Technologies and Systems (2010), IEEE.
- [130] OSOSKY, S., SANDERS, T., JENTSCH, F., HANCOCK, P., AND CHEN, J. Y. C. Determinants of system transparency and its influence on trust in and reliance on unmanned robotic systems. Unmanned Systems Technology XVI 9084 (2014), 90840E.
- [131] PAK, R., FINK, N., PRICE, M., BASS, B., AND STURRE, L. Decision support aids with anthropomorphic characteristics influence trust and performance in younger and older adults. *Ergonomics* 55, 9 (July 2012), 1059–1072.
- [132] PARASURAMAN, R., AND MANZEY, D. H. Complacency and bias in human use of automation: An attentional integration. *Human factors* 52, 3 (2010), 381–410.
- [133] PARASURAMAN, R., AND MILLER, C. A. Trust and etiquette in high-criticality automated systems. *Communications of the ACM* 47, 4 (2004), 51–55.
- [134] PARASURAMAN, R., AND RILEY, V. Humans and automation: Use, misuse, disuse, abuse. Human factors 39, 2 (1997), 230–253.

- [135] PETERSEN, L., ROBERT, L., JESSIE YANG, X., AND TILBURY, D. M. Situational awareness, driver's trust in automated driving systems and secondary task performance. arXiv (2019), 1–26.
- [136] POWERS, D. M. Evaluation: from precision, recall and f-measure to roc, informedness, markedness and correlation. arXiv preprint arXiv:2010.16061 (2020).
- [137] PRINZEL, L. The relationship of self-efficacy and complacency in pilot-automation interaction (technical memorandum no. tm-2002-211925). NASA Langley Research Center, Hampton, VA (2002).
- [138] PYLYSHYN, Z. W. Seeing and visualizing: It's not what you think. MIT press, 2003.
- [139] PYLYSHYN, Z. W., AND STORM, R. W. Tracking multiple independent targets: Evidence for a parallel tracking mechanism. *Spatial vision* 3, 3 (1988), 179–197.
- [140] RACHEV, S. T., STOYANOV, S. V., AND FABOZZI, F. J. Risk and uncertainty, vol. 211. John Wiley & Sons, 2011.
- [141] RAMBUSCH, J., JAKOBSSON, P., AND PARGMAN, D. Exploring e-sports: A case study of game play in counter-strike. In 3rd Digital Games Research Association International Conference:" Situated Play", DiGRA 2007, Tokyo, 24 September 2007 through 28 September 2007 (2007), vol. 4, Digital Games Research Association (DiGRA), pp. 157–164.
- [142] RAMCHURN, S. D., FISCHER, J. E., IKUNO, Y., WU, F., FLANN, J., AND WALDOCK, A. A study of human-agent collaboration for multi-uav task allocation in dynamic environments. In *Proceedings of the 24th International Joint Conference on Artificial Intelli*gence (IJCAI) (2015), pp. 1184–1192.
- [143] RAMCHURN, S. D., WU, F., JIANG, W., FISCHER, J. E., REECE, S., ROBERTS, S., RODDEN, T., GREENHALGH, C., AND JENNINGS, N. R. Human-agent collaboration for disaster response. Autonomous Agents and Multi-Agent Systems 30, 1 (2016), 82–111.
- [144] RAPP, A., HOPFGARTNER, F., HAMARI, J., LINEHAN, C., AND CENA, F. Strengthening gamification studies: Current trends and future opportunities of gamification research. *International Journal of Human Computer Studies 127*, November 2018 (2019), 1–6.
- [145] REASON, J. Human error. Cambridge university press, 1990.
- [146] ROBINETTE, P., HOWARD, A. M., AND WAGNER, A. R. Effect of Robot Performance on Human-Robot Trust in Time-Critical Situations. *IEEE Transactions on Human-Machine* Systems 47, 4 (2017), 425–436.
- [147] ROSS, J. M. Moderators of trust and reliance across multiple decision aids. university of central florida, 2008.

- [148] ROSS, J. M., SZALMA, J. L., HANCOCK, P. A., BARNETT, J. S., AND TAYLOR, G. The effect of automation reliability on user automation trust and reliance in a searchand-rescue scenario. In proceedings of the human factors and ergonomics society annual meeting (2008), vol. 52, Sage Publications Sage CA: Los Angeles, CA, pp. 1340–1344.
- [149] SACHA, D., SENARATNE, H., KWON, B. C., ELLIS, G., AND KEIM, D. A. The role of uncertainty, awareness, and trust in visual analytics. *IEEE Transactions on Visualization* and Computer Graphics 22, 1 (Jan. 2016), 240–249.
- [150] SALEM, M., LAKATOS, G., AMIRABDOLLAHIAN, F., AND DAUTENHAHN, K. Would You Trust a (Faulty) Robot?: Effects of Error, Task Type and Personality on Human-Robot Cooperation and Trust. ACM/IEEE International Conference on Human-Robot Interaction 2015-March (2015), 141–148.
- [151] SANCHEZ, J., ROGERS, W. A., FISK, A. D., AND ROVIRA, E. Understanding reliance on automation: Effects of error type, error distribution, age and experience. *Theoretical Issues in Ergonomics Science* 15, 2 (2014), 134–160.
- [152] SARTER, N., SCHROEDER, B., AND MCGUIRL, J. Supporting decision-making and action selection under time pressure and uncertainty: The case of inflight icing. 39th Aerospace Sciences Meeting and Exhibit 43, 4 (2001), 573–583.
- [153] SATTERFIELD, K., BALDWIN, C., DE VISSER, E., AND SHAW, T. The influence of risky conditions in trust in autonomous systems. In *Proceedings of the Human Factors and Ergonomics Society* (2017), vol. 2017-Octob.
- [154] SAUER, J., CHAVAILLAZ, A., AND WASTELL, D. Experience of automation failures in training: effects on trust, automation bias, complacency and performance. *Ergonomics* 59, 6 (June 2016), 767–780.
- [155] SCHAEFER, K. E., CHEN, J. Y. C., SZALMA, J. L., AND HANCOCK, P. A. A metaanalysis of factors influencing the development of trust in automation. *Human Factors: The Journal of the Human Factors and Ergonomics Society* 58, 3 (Mar. 2016), 377–400.
- [156] SCHAEKERMANN, M., RIBEIRO, G., WALLNER, G., KRIGLSTEIN, S., JOHNSON, D., DRACHEN, A., SIFA, R., AND NACKE, L. E. Curiously motivated: Profiling curiosity with self-reports and behaviour metrics in the game" destiny". In *Proceedings of the Annual Symposium on Computer-Human Interaction in Play* (2017), pp. 143–156.
- [157] SCHAFFER, J., O'DONOVAN, J., MARUSICH, L., YU, M., GONZALEZ, C., AND HÖLLERER, T. A study of dynamic information display and decision-making in abstract trust games. *International Journal of Human Computer Studies 113*, December 2017 (2018), 1–14.

- [158] SEABOLD, S., AND PERKTOLD, J. statsmodels: Econometric and statistical modeling with python. In 9th Python in Science Conference (2010).
- [159] SELCON, S., TAYLOR, R., AND KORITSAS, E. Workload or situational awareness?: TLX vs. SART for aerospace systems design evaluation. *Proceedings of the Human Factors Society Annual Meeting 35*, 2 (Sept. 1991), 62–66.
- [160] SENARATNE, H., GERHARZ, L., PEBESMA, E., AND SCHWERING, A. Usability of spatiotemporal uncertainty visualisation methods. In *Bridging the geographic information sciences.* Springer, 2012, pp. 3–23.
- [161] SGOBBA, T. B-737 max and the crash of the regulatory system. Journal of Space Safety Engineering 6, 4 (2019), 299–303.
- [162] SHAPIRO, S. S., AND WILK, M. B. An analysis of variance test for normality (complete samples). *Biometrika* 52, 3-4 (Dec. 1965), 591–611.
- [163] SHEKHAR, S., COYLE, M., SHARGAL, M., KOZAK, J., AND HANCOCK, P. Design and validation of headup displays for navigation in ivhs. In Vehicle Navigation and Information Systems Conference, 1991 (1991), vol. 2, IEEE, pp. 537–542.
- [164] SHELDON, M. R., FILLYAW, M. J., AND THOMPSON, W. D. The use and interpretation of the friedman test in the analysis of ordinal-scale data in repeated measures designs. *Physiotherapy Research International 1*, 4 (1996), 221–228.
- [165] SHERIDAN, T. Trustworthiness of command and control systems. In Analysis, Design and Evaluation of Man-Machine Systems 1988. Elsevier, 1989, pp. 427–431.
- [166] SHIRADO, H., AND CHRISTAKIS, N. A. Locally noisy autonomous agents improve global human coordination in network experiments. *Nature* 545, 7654 (2017), 370–374.
- [167] SINGH, I. L., MOLLOY, R., AND PARASURAMAN, R. Automation-induced" complacency": Development of the complacency-potential rating scale. *The International Journal of Aviation Psychology* 3, 2 (1993), 111–122.
- [168] SIX, F., AND SORGE, A. Creating a high-trust organization: An exploration into organizational policies that stimulate interpersonal trust building. *Journal of Management Studies* 45, 5 (2008), 857–884.
- [169] SOSA, A., STANTON, R., PEREZ, S., KEYES-GARCIA, C., GONZALEZ, S., AND TOUPS, Z. O. Imperfect robot control in a mixed reality game to teach hybrid human-robot team coordination. In *Proceedings of the 2015 Annual Symposium on Computer-Human Interaction in Play* (2015), pp. 697–702.

- [170] STOWERS, K., KASDAGLIS, N., RUPP, M., CHEN, J., BARBER, D., AND BARNES, M. Insights into human-agent teaming: Intelligent agent transparency and uncertainty. In Advances in Human Factors in Robots and Unmanned Systems. Springer, 2017, pp. 149– 160.
- [171] SWELLER, J. Cognitive load theory. In Psychology of learning and motivation, vol. 55. Elsevier, 2011, pp. 37–76.
- [172] TAHA, Z., AND JIZAT, J. A. M. A comparison of two approaches for collision avoidance of an automated guided vehicle using monocular vision. *Applied Mechanics and Materials* 145 (2012), 547–551.
- [173] TAYLOR, R. M. Situational awareness rating technique (sart): The development of a tool for aircrew systems design. In *Situational awareness*. Routledge, 2017, pp. 111–128.
- [174] TORRE, I., GOSLIN, J., WHITE, L., AND ZANATTO, D. Trust in artificial voices: A" congruency effect" of first impressions and behavioural experience. In *Proceedings of the Technology, Mind, and Society.* Association for Computing Machinery, Inc., 2018, pp. 1–6.
- [175] TUKEY, J. W. Comparing individual means in the analysis of variance. Biometrics (1949), 99–114.
- [176] TULSHAN, A. S., AND DHAGE, S. N. Survey on virtual assistant: Google assistant, siri, cortana, alexa. In *International symposium on signal processing and intelligent recognition* systems (2018), Springer, pp. 190–201.
- [177] TURKLE, S. The intimate machine. Science'84 5 (1984), 40–47.
- [178] VALLAT, R. Pingouin: statistics in python. The Journal of Open Source Software 3, 31 (Nov. 2018), 1026.
- [179] VAN HUYCK, J. B., BATTALIO, R. C., AND BEIL, R. O. Coordination Games, Strategic Uncertainty, and Coordination Failure. *American Economic Association 80*, 1 (1990), 234–248.
- [180] VAN LIER, R., VAN DER HELM, P., AND LEEUWENBERG, E. Integrating global and local aspects of visual occlusion. *Perception 23*, 8 (1994), 883–903.
- [181] VAN ROSSUM, G., AND DRAKE, F. L. Python 3 Reference Manual. CreateSpace, Scotts Valley, CA, 2009.
- [182] VAN WISSEN, A., GAL, Y., KAMPHORST, B., AND DIGNUM, M. Human-agent teamwork in dynamic environments. *Computers in Human Behavior* 28, 1 (2012), 23–33.

- [183] VIERA, A. J., GARRETT, J. M., ET AL. Understanding interobserver agreement: the kappa statistic. *Fam med 37*, 5 (2005), 360–363.
- [184] VIRTANEN, P., GOMMERS, R., OLIPHANT, T. E., HABERLAND, M., REDDY, T., COUR-NAPEAU, D., BUROVSKI, E., PETERSON, P., WECKESSER, W., BRIGHT, J., VAN DER WALT, S. J., BRETT, M., WILSON, J., MILLMAN, K. J., MAYOROV, N., NELSON, A. R. J., JONES, E., KERN, R., LARSON, E., CAREY, C. J., POLAT, İ., FENG, Y., MOORE, E. W., VANDERPLAS, J., LAXALDE, D., PERKTOLD, J., CIMRMAN, R., HEN-RIKSEN, I., QUINTERO, E. A., HARRIS, C. R., ARCHIBALD, A. M., RIBEIRO, A. H., PEDREGOSA, F., VAN MULBREGT, P., AND SCIPY 1.0 CONTRIBUTORS. SciPy 1.0: Fundamental Algorithms for Scientific Computing in Python. *Nature Methods 17* (2020), 261–272.
- [185] WALKER, F., WANG, J., MARTENS, M., AND VERWEY, W. Gaze behaviour and electrodermal activity: Objective measures of drivers' trust in automated vehicles. *Transportation research part F: traffic psychology and behaviour 64* (2019), 401–412.
- [186] WANG, N., PYNADATH, D. V., AND HILL, S. G. The Impact of POMDP-Generated Explanations on Trust and Performance in Human-Robot Teams. *Aamas*, Aamas (2015), 997–1005.
- [187] WANG, X., SHI, Z., ZHANG, F., AND WANG, Y. Dynamic real-time scheduling for human-agent collaboration systems based on mutual trust. *Cyber-Physical Systems* 1, 2-4 (2015), 76–90.
- [188] WHITE, M. P., AND EISER, J. R. Marginal trust in risk managers: Building and losing trust following decisions under uncertainty. *Risk Analysis 26*, 5 (Oct. 2006), 1187–1203.
- [189] WICZOREK, R., AND MANZEY, D. Supporting attention allocation in multitask environments: Effects of likelihood alarm systems on trust, behavior, and performance. *Human Factors 56*, 7 (2014), 1209–1221.
- [190] WIEBE, E. N., LAMB, A., HARDY, M., AND SHAREK, D. Measuring engagement in video game-based environments: Investigation of the User Engagement Scale. *Computers* in Human Behavior 32 (2014), 123–132.
- [191] WILCOX, L. M., ALLISON, R. S., HELLIKER, J., DUNK, B., AND ANTHONY, R. C. Evidence that viewers prefer higher frame-rate film. ACM Transactions on Applied Perception (TAP) 12, 4 (2015), 1–12.
- [192] WILCOXON, F. Individual comparisons by ranking methods. In Breakthroughs in statistics. Springer, 1992, pp. 196–202.

- [193] WILLIAMS, E. Experimental designs balanced for the estimation of residual effects of treatments. Australian Journal of Chemistry 2, 2 (1949), 149.
- [194] WILLIAMS, L. J. Tunnel vision induced by a foveal load manipulation. Human factors 27, 2 (1985), 221–227.
- [195] WILLIAMSON, D. F. The box plot: A simple visual method to interpret data. Annals of Internal Medicine 110, 11 (June 1989), 916.
- [196] WOOLGAR, S., AND SUCHMAN, L. A. Plans and Situated Actions: The Problem of Human Machine Communication. *Contemporary Sociology* 18, 3 (1989), 414.
- [197] WRIGHTSMAN, L. S. Interpersonal trust and attitudes toward human nature. In Measures of Personality and Social Psychological Attitudes. Elsevier, 1991, pp. 373–412.
- [198] WU, W., EKAETTE, E., AND FAR, B. H. Uncertainty management framework for multi-agent system. In *Proceedings of ATS* (2003), vol. 2003.
- [199] XU, M., FRALICK, D., ZHENG, J. Z., WANG, B., TU, X. M., AND FENG, C. The differences and similarities between two-sample t-test and paired t-test. *Shanghai archives* of psychiatry 29, 3 (2017), 184.
- [200] ZAPHIRIS, P., AND ANG, C. S. HCI issues in computer games. Interacting with Computers 19, 2 (03 2007), 135–139.
- [201] ZHOU, J., LI, Z., HU, H., YU, K., CHEN, F., LI, Z., AND WANG, Y. Effects of influence on user trust in predictive decision making. In *Extended Abstracts of the 2019 CHI Conference on Human Factors in Computing Systems* (May 2019), ACM.

Part IV Appendices

Appendix A

Agent Reliability and Predictability Study

A.A Pre-Hoc Survey

Start of Block: General Questions

Q0 Thank you for participating in this study. Please answer the questions below.

Q1 Please enter your participant ID:

Q2 What is your age?

Q3 What is your gender?

 \bigcirc Male (1)

 \bigcirc Female (2)

O Non-Binary (3)

 \bigcirc Self-Defined (4)

 \bigcirc Prefer not to say (5)

Q9 What is your current status?

O Student (1)

 \bigcirc Employed (2)

 \bigcirc Unemployed (3)

 \bigcirc Other (4)

 \bigcirc Prefer not to say (5)

Q4 Are you a native English Speaker?

○ Yes (1)

O No (2)

Q5 How would you rate your English comprehension skills?

	Very Poor (1)	Poor (2)	Average (3)	Good (4)	Very Good (5)	l'm a native speaker (6)
Make your selection: (1)	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc

Q6 How often do you play video-games?

	Never (1)	Once a year (2)	Once in Several Months (3)	Once a Month (4)	Several times a month (5)	Everyday (6)
Make your selection: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q7 What kind of games do you usually play (example: 3D shooters, 2D plateformers, short mobiles games etc...) if you selected "never" on the last question, you can skip this one.

End of Block: General Questions

Start of Block: Please read and rate the following statements

Q0 Please rate the following statements.

Q8 Manually sorting through card catalogues is more reliable than computer-aidedsearches for finding items in a library.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q10 If I need to have a tumor in my body removed, I would choose to undergo computeraided surgery using laser technology because computerized surgery is more reliable and safer than manual surgery.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	0	0	0	\bigcirc	0

Q11 People save time by using automatic teller machines (ATMs) rather than a bank teller in making transactions.

Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
-----------------------	--------------	-------------	-----------	-----------------------

Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q51 I do not trust automated devices such as ATMs and computerized airline reservations.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	0	0	\bigcirc	\bigcirc

Q12 People who work frequently with automated devices have lower job satisfaction because they feel less involved in their job than those who work manually.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q13 I feel safer depositing my money at an ATM than with a human teller.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	0	\bigcirc

Q14 I have to tape an important TV program for a class assignment. To ensure that the correct program is recorded, I would use the automatic programming facility on my recorder rather than manual taping.

Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
--------------------------	--------------	-------------	-----------	-----------------------

Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q15 People whose jobs require them to work with automated systems are lonelier than people who do not work with such devices.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	0	0	\bigcirc	\bigcirc

Q16 Automated systems used in modern aircraft, such as the automatic landing system, have made air journey safer.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc

Q17 ATMs provide safeguard against the inappropriate use of an individual's bank account by dishonest people.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)				
Select an answer: (1)	\bigcirc	0	\bigcirc	\bigcirc	\bigcirc				
Q18 Automated devices used in aviation and banking have made work easier for both employees and customers.									

Strongly Disagre	ee (2) Neutral (3)) Agree (4)	Strongly
------------------	--------------------	-------------	----------

Disagree (1)					Agree (5)
Select an answer: (1)	0	\bigcirc	0	\bigcirc	0

Q19 I often use automated devices.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q20 People who work with automated devices have greater job satisfaction because they feel more involved than those who work manually.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	0	0	0	0	0

Q21 Automated devices in medicine save time and money in the diagnosis and treatment of diseases.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q22 Even though the automatic cruise control in my car is set at a speed below the speed limit, I worry when I pass a police radar speed-trap in case the automatic control is not working properly.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	0	0	\bigcirc	\bigcirc

Q23 Bank transactions have become safer with the introduction of computer technology for the transfer of funds.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q24 I would rather purchase an item using a computer than have to deal with a sales representative on the phone because my order is more likely to be correct using the computer.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q25 Work has become more difficult with the increase of automation in aviation and banking.

Dis	sagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Agree (5)
Select an answer: (1)	0	0	0	0	\bigcirc

Q27 I do not like to use ATMs because I feel that they are sometimes unreliable.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0

Q28 I think that automated devices used in medicine, such as CAT-scans and ultrasound, provide very reliable medical diagnosis.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	0	0	\bigcirc	\bigcirc

End of Block: Please read and rate the following statements

Start of Block: Block 2

Q37 Thank your for answering this preliminary set of questions.

You can now begin to play the game.

When you're done, please click on the "next" button to answer a final short set of questions.

End of Block: Block 2

A.B Post-Hoc Survey

Start of Block: Block 3

Q41 All of the following statements are referring to the game you just played. Please rate them according to your own personal experience.

Q40 I thought it	was fun Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Q42 I was fully	occupied with th Strongly	e game Disagree (2)	Neutral (3)	Agree (4)	Strongly
Select an answer: (1)		0	0	0	Agree (5)
Q43 I thought a	bout other things Strongly Disagree (1)	s Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	0	0	0	0	\bigcirc
Q44 I found it ti	resome Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)

Select an answer: (1)	\bigcirc	0	0	0	\bigcirc

Q45 I felt comp	petent				
	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	0	0	0	0	0
Q46 I thought i	t was hard Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Q47 I felt frustr	ated				
	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc
Q48 I felt time	pressure Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	0	\bigcirc	\bigcirc	\bigcirc	0

Q49 I felt succe	essful Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)		
Select an answer: (1)	\bigcirc	0	0	\bigcirc	0		
End of Block: Block 3							

Appendix B

Agent Errors and Behaviours Study

B.A Pre-Hoc Survey

Start of Block: General Questions

Q0

Thank you for participating in this study. Please answer the questions below. Please note that all of your answers will be anonymized.

Q1

Please enter the ID given to you by the researcher in charge of study:

Q2 Please enter your age:

Q3 What is your gender?

 \bigcirc Male (1)

 \bigcirc Female (2)

O Non-Binary (3)

 \bigcirc Self-Defined (4)

 \bigcirc Prefer not to say (5)

Q4 Education: What is the highest degree or level of school you have completed? If currently enrolled, highest degree received.

- \bigcirc No schooling completed (1)
- \bigcirc Nursery school to 8th grade (2)
- \bigcirc Some high school, no diploma (3)
- O High school graduate, diploma or equivalent (4)
- \bigcirc Some college credit, no degree (5)
- Trade/technical/vocational training (6)

Associate degree (7)

- O Bachelor's degree (8)
- O Master's degree (9)
- O Professional degree (10)
- O Doctorate degree (12)
- O Prefer not to say (13)

Q5 In which field are you currently employed? (e.g "automotive industry", "medical" etc...) If you are studying or on the lookout for a job, which field is your main focus?

End of Block: General Questions

Start of Block: Block 5

Q6 How often do you play video-games?

Never (1)	Very Rarely (2)	Rarely (7)	Occasionally (8)	Frequently (9)	Very Frequently (10)
-----------	--------------------	------------	---------------------	-------------------	----------------------------

Make your selection: (1)	\bigcirc	0	0	0	0	\bigcirc

Q7 What kind of games do you usually play (example: 3D first person shooters, 2D plateformers, small mobiles games etc...) if you selected "never" on the last question, you can skip this one.

End of Block: Block 5

Start of Block: Block 3

Q8 Please rate the following statements

	Never (1)	Rarely (2)	Sometimes (6)	Very Often (7)	Constantly (8)
When I have a lot to do, it makes sense to delegate a task to automation. (1)	0	0	0	0	0
If life were busy, I would let an automated system handle some tasks for me. (2)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Automation should be used to ease people's workload. (3)	0	\bigcirc	\bigcirc	\bigcirc	\bigcirc
If automation is available to help me with	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

something, it makes sense for me to pay more attention to my other tasks. (4) Even if an automated aid can help me with a task, I should pay attention to its performance. (5) Distractions and interruptions are less of a problem for me when I have an automated system to cover some of the work. (6) Constantly monitoring an automated system's performance is a waste of time. (7) Even when I have a lot to do, I am likely to watch automation carefully for errors. (8) It's not usually necessary to

pay much

 \bigcirc \bigcirc \bigcirc () \bigcirc
attention to automation when it is running. (9)					
Carefully watching automation takes time away from more important or interesting things. (10)	0	0	\bigcirc	0	0

End of Block: Block 3

Start of Block: Block 2

Q9 Thank your for answering this preliminary set of questions.

You can now begin to play the game.

After you're done playing, please click on the button below to answer a final short set of questions.

B.B Post-Hoc Survey

Start of Block: Block 3

Q10 All of the following statements are referring to the game you just played. Please rate them according to your own personal experience.

Q11 I forgot about my immediate surroundings while playing this game. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Select an answer: (1) \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Q12 I was so involved in this game that I ignored everything around me. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Select an answer: (1) \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Q13 I lost myself in this game. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Select an answer: (1) \bigcirc \bigcirc \bigcirc \bigcirc Q14 I was so involved in this game that I lost track of time. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5)

Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q15 I blocked out things around me when I was playing this game.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	0	0	\bigcirc	\bigcirc

Q16 When I was playing this game, I lost track of the world around me.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	0	0	\bigcirc	\bigcirc

Q18 The time I spent playing this game just slipped away.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc

Q19 I was absorbed in this game.

	Strongly Disagree (1)	Disagree (2)	Neutral (3)	Agree (4)	Strongly Agree (5)
Select an answer: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q20 During this	game I let mys Strongly agree (1)	elf go. Disagree (6)	Neutral (7)	Agree (8)	Strongly Agree (9)
Click to write Statement 1 (1)	0	0	0	\bigcirc	0
End of Block: E	Block 3				

Appendix C

Visual Uncertainty Study

C.A Pre-Hoc Survey

Start of Block: General Questions

Q0

Thank you for participating in this study. Please answer the questions below. Please note that all of your answers will be anonymized.

Q1

Please enter the ID given to you by the researcher in charge of study:

Q2 Please enter your age:

Q3 What is your gender?

 \bigcirc Male (1)

 \bigcirc Female (2)

O Non-Binary (3)

 \bigcirc Self-Defined (4)

 \bigcirc Prefer not to say (5)

Q7 What is your gender?

 \bigcirc Male (1)

 \bigcirc Female (2)

O Non-Binary (3)

 \bigcirc Self-Defined (4)

O Prefer not to say (5)

Q8 What is your country of origin?

 \bigcirc Please enter your country of origin below: (1)

 \bigcirc Prefer not to say (2)
Q9 Education: What is the highest degree or level of school you have completed? If currently enrolled, highest degree received.

 \bigcirc No formal education (1)

 \bigcirc High school diploma (2)

College degree (3)

○ Vocational training (14)

O Bachelor's degree (15)

O Master's degree (16)

O Professional degree (17)

O Doctorate degree (18)

O Prefer not to say (13)

Q10 In which field are you currently employed? (e.g "automotive industry", "medical" etc...) If you are studying or on the lookout for a job, which field is your main focus?

End of Block: Demographic Survey

Start of Block: Gaming-related Questions

Q11 How often do you play video-games?

	Never (1)	Very Rarely (2)	Rarely (7)	Occasionally (8)	Frequently (9)	Very Frequently (10)
Make your selection: (1)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Q12 What kind of games do you usually play (example: 3D first person shooters, 2D platformers, casual mobiles games etc...). If you selected "never" on the last question, you can skip answering this one.

Q13 How old is the computer that you are using for this study?

Q14

Please indicate below the name of the CPU and/or GPU that your computer is currently using. If you do not know it, simply write the brand/name of your computer (ex: "HP Laptop", "Apple Macbook Air" etc...)

End of Block: Gaming-related Questions

C.B Post-Hoc Survey

Start of Block: Block 3

Q10 All of the following statements are referring to the game you just played. Please rate them according to your own personal experience.

Q11 I forgot about my immediate surroundings while playing this game. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Select an answer: (1) \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Q12 I was so involved in this game that I ignored everything around me. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Select an answer: (1) \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Q13 I lost myself in this game. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Select an answer: (1) \bigcirc \bigcirc \bigcirc \bigcirc Q14 I was so involved in this game that I lost track of time. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Q8 During one of the session, **a "disruption" (static-like effect) hid the TOP of the screen**. How did that affected your ability to play and your reliance on the agent?

Q10 During one of the session, **a "disruption" (static-like effect) hid** the BOTTOM of the screen. How did that affected your ability to play and reliance on the agent?

Q13 Please rate the difficulty of each conditions from 0 (very easy) to 100 (very hard): Easy Medium Hard 0 10 20 30 40 50 60 70 80 90 100 Darkness & Spotlight () Clouds () Disruption (top of the screen) () Disruption (bottom of the screen) ()

Q15 Please give us some feedback or recommendations about the study below:

End of Block: Default Question Block

Appendix D

Visual Explanation and Agent Transparency Study

D.A Pre-Hoc Survey

Start of Block: General Questions

Q0

Thank you for participating in this study. Please answer the questions below. Please note that all of your answers will be anonymized.

Q1

Please enter the ID given to you by the researcher in charge of study:

Q2 Please enter your age:

Q3 What is your gender?

 \bigcirc Male (1)

 \bigcirc Female (2)

O Non-Binary (3)

 \bigcirc Self-Defined (4)

 \bigcirc Prefer not to say (5)

Q7 Gender: How do you identify?

O Female (1)

 \bigcirc Male (2)

O Non-Binary (3)

O Self-Defined: (4) _____

 \bigcirc Prefer not to say (5)

X→

Q19 Origins: Where are you from?

▼ Afghanistan (1) ... Zimbabwe (1357)

Q9 Education: What is the highest degree or level of school you have completed? If currently enrolled, highest degree received.

 \bigcirc No formal education (1)

 \bigcirc High school diploma (2)

 \bigcirc College degree (3)

O Vocational training (14)

 \bigcirc Bachelor's degree (15)

O Master's degree (16)

O Professional degree (17)

 \bigcirc Doctorate degree (18)

O Prefer not to say (13)

End of Block: Demographic Survey

Start of Block: Gaming-related Questions

Q11 Please read and rate the following statements.

	1 Strongly Disagree (12)	2 Somewhat Disagree (13)	3 Neither disagree or agree (14)	4 Somewhat Agree (15)	5 Strongly Agree (16)
l am good at playing computer games. (2)	0	0	0	0	0
Playing computer games is easy for me. (3)	\bigcirc	0	0	0	\bigcirc
l understand and play computer games well.	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc



End of Block: Gaming-related Questions

Start of Block: Block 4

Q18 Please read and rate the following statements.

	1 Strongly disagree (18)	2 Somewhat disagree (19)	3 Neither agree nor disagree (20)	4 Somewhat agree (21)	5 Strongly agree (22)
When I have a lot to do, it makes sense to delegate a task to automation. (1)	0	0	0	0	0
If life were busy, I would let an automated system handle some tasks for me. (2)	0	0	0	0	0
Automation should be used to ease people's workload. (3)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0
If automation is available to help me with something, it makes sense for me to pay more attention to my other tasks. (4)	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

Page 4 of 6

Even if an automated aid can help me with a task, I should pay attention to its performance. (5) Distractions and interruptions are less of a problem for me when I have an automated system to cover some of the work. (6) Constantly monitoring ar automated system's performance is a waste of time. (7) Even when I have a lot to do, I am likely to watch automation carefully for errors. (8) It's not usually necessary to pay much attention to automation when it is running. (9) Carefully watching

automation takes time

ל ר י	\bigcirc	0	0	\bigcirc	0
i I	0	\bigcirc	0	\bigcirc	\bigcirc
n e f	\bigcirc	0	0	\bigcirc	\bigcirc
y Y	0	0	0	0	0
)	0	\bigcirc	\bigcirc	\bigcirc	0
	\bigcirc	0	0	\bigcirc	0

away from more important or interesting things. (10)

End of Block: Block 4

D.B Post-Hoc Survey

Start of Block: Block 3

Q10 All of the following statements are referring to the game you just played. Please rate them according to your own personal experience.

Q11 I forgot about my immediate surroundings while playing this game. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Select an answer: (1) \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Q12 I was so involved in this game that I ignored everything around me. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Select an answer: (1) \bigcirc \bigcirc \bigcirc \bigcirc \bigcirc Q13 I lost myself in this game. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Select an answer: (1) \bigcirc \bigcirc \bigcirc \bigcirc Q14 I was so involved in this game that I lost track of time. Strongly Strongly Disagree (2) Neutral (3) Agree (4) Disagree (1) Agree (5) Q12 You can use the textbox below to give us some additional feedback about the study:

End of Block: Last questions

Appendix E

Survey Instruments

E.A Complacency Potential Questionnaire

	001	nplacency-Poter	itial Factors an	d Associated	Items	
Factor			Factor Item			Item Loading
Confidence						
	1. I think that automated devices used in medicine, such as CT					
		scans and ultraso	und, provide very	reliable medica	1	
	2	diagnosis.				.68
	2.	Automated device	es in medicine save	e time and mon	ey in the	
	3	If I need to have	a tumor in my bo	dy removed I	would	.55
	5.	choose to undergo	a computer-aided	surgery using le	would	
		technology becaus	se it is more reliab	le and safer the	an	
		manual surgery.				.54
	4.	Automated system	ns used in modern	aircraft, such	as the	
		automatic landing	system, have mad	de air journeys	safer.	.51
Reliance						
	1.	ATMs provide a s	afeguard against	the inappropria	te use of	
		an individual's ba	nk account by dis	honest people.		.78
	2.	Automated device	s used in aviation	and banking h	ave made	
	2	work easier for bo	oth employees and	customers.		.51
	5.	a speed below the	speed limit I way	ontrol in my ca	r is set at	
		radar sneed tran i	n case the automa	tic control is n	a ponce	
		working properly.	ii case the automa			35
Trust		6 F - F - F - I , I				.55
	1.	Manually sorting	through card catal	logues is more	reliable	
		than computer-aid	led searches for fi	nding items in	a library.	.69
	2.	I would rather pu	rchase an item usi	ng a computer	than have	
	1	to deal with a sale	es representative of	n the phone be	cause my	
		order is more likely to be correct using the computer.				
	3.	Bank transactions	have become safe	er with the intro	oduction	
Safety		of computer techn	lology for the tran	ister of funds.		.53
Salety	1	I feel safer denosi	ting my money at	an ATM than	with a	
		human teller.	ting my money at	an Arwi than	with a	60
	2.	I have to tape an	important TV pro	gram for a clas	s	.02
		assignment. To en	sure that the corre	ect program is	recorded,	
		I would use the au	itomatic program	ning facility on	my VCR	
	1	rather than manua	al taping.			.66
			TABLE 2			
	Fa	actor Structure	and Internal Co	onsistency of	the	
		Complacen	cy-Potential Ra	ting Scale		
				Factor		
Criteria		General	Confidence	Reliance	Trust	Safety
Eigenvalue		3.05	1.76	1.44	1.21	1.05
Cumulative p	ercent					
of variance		19.1	30.1	39.1	46.7	53.2
nternal cons	istency					
(alpha)		.97	.82	.85	.89	.95

Complacency-Potential Factors and Associated Items

Figure E.1: Complacency Potential Questionnaire by Singh et al. [167]

Checklist for Trust between People and Automation

Below is a list of statement for evaluating trust between people and automation. There are several scales for you to rate intensity of your feeling of trust, or your impression of the system while operating a machine. Please mark an "x" on each line at the point which best describes your feeling or your impression.

(Note: not at all=1; extremely=7)

1		The system is deceptive				
			-		L	
	1	2 3 4	5	6	7	
2		The system behaves in an unde	rhanded	manner		
	1		5	6	7	
0		Lam supplicious of the sustant's	intent of	-*'		
3		am suspicious of the system s	intent, ad	caon, or o	utputs	
	1	2 3 4	5	6	7	
4		I am wary of the system				
	1	2 3 4	5	6	7	
5		The system's actions will have a	harmful	or injurio	us outcome	
	1		5	6		
	'	2 3 4	5	0	'	
6		I am confident in the system				
	1	2 3 4	5	6	7	
7		The system provides security				
	1	2 3 4	5	6	7	
8		The system has integrity				
	1		5	6	l	
	'	2 5 4	5	0	,	
9		The system is dependable				
	1	2 3 4	5	6	7	
10		The system is reliable				
	1	2 3 4	5	6	7	
11		I can trust the system				
	4		5	6	7	
	'	2 3 4	5	0	/	
12		I am familiar with the system				
	1	2 3 4	5	6	7	

 $\label{eq:FIGURE8} \textbf{FIGURE8} \quad \mbox{Proposed questionnaire to measure trust between people and automated systems}.$

Figure E.2: Checklist for Trust Between People and Automation by Jian et al. [90].

NASA Task Load Index

Hart and Staveland's NASA Task Load Index (TLX) method assesses work load on five 7-point scales. Increments of high, medium and low estimates for each point result in 21 gradations on the scales.

Name	Task		Date
Mental Demand	Ho	w mentally der	nanding was the task?
Very Low			Very High
Physical Demand	How physic	ally demanding) was the task?
Temporal Demand	How hurried	d or rushed was	s the pace of the task?
Very Low			Very High
Performance	How succes you were as	ssful were you i iked to do?	in accomplishing what
Perfect			Failure
Effort	How hard d your level o	id you have to f performance?	work to accomplish
Very Low			Very High
Frustration	How insecu and annoye	re, discourage d wereyou?	d, irritated, stressed,
Very Low			Very High

Figure E.3: NASA TLX Cognitive Workload survey designed by Hart et al. [76]



Figure E.4: Situational Awareness Rating Technique designed by Taylor et al. [173]